ESSENTIAL NUMBER THEORY

Counting locally supercuspidal newforms

Andrew Knightly

2025 vol. 4 no. 2



https://doi.org/10.2140/ent.2025.4.349



Counting locally supercuspidal newforms

Andrew Knightly

The trace formula is a versatile tool for computing sums of spectral data across families of automorphic forms. Using specialized test functions, one can treat small families with refined spectral properties. This has proven fruitful in analytic applications. We detail such methodology here, with the aim of counting newforms in certain small families. The result is a general formula for the number of holomorphic newforms of weight k and level N whose local representation type at each $p \mid N$ is a fixed supercuspidal representation σ_p of $\mathrm{GL}_2(\mathbb{Q}_p)$. This is given in terms of local elliptic orbital integrals attached to matrix coefficients of the σ_p . We evaluate the formula explicitly in the case where each σ_p has conductor $\leq p^3$. The technical heart of the paper is the explicit calculation of elliptic orbital integrals attached to such σ_p . We also compute the traces of Hecke operators on the span of these newforms. Some applications are given to biases among root numbers of newforms.

1. In	troduction	350
1.1.	Overview	350
1.2.	Description of main results	351
1.3.	Dimension formulas and root number bias	357
2. N	otation and Haar measure	361
3. T	he simple trace formula	362
3.1.	Background on supercuspidal representations of GL(2)	362
3.2.	Simple trace formula	363
3.3.	Factorization of orbital integrals	364
4. C	ounting locally supercuspidal newforms	367
4.1.	Isolating $S_k(\hat{\sigma})$ spectrally	368
4.2.	First main result: the trace of a Hecke operator	371
4.3.	Known results about the elliptic terms	372
4.4.	Local orbital integrals at primes $\ell \nmid N$: hyperbolic case	373
4.5.	Local orbital integrals at primes $\ell \nmid N$: elliptic case	374
4.6.	The set of relevant γ	381
4.7.	The measure of $\overline{G_{\gamma}(F)} \setminus \overline{G_{\gamma}(\mathbb{A}_F)}$	382

MSC2020: 11F41, 11F70, 11F72.

Keywords: modular forms, trace formula, supercuspidal representations, automorphic forms.

5. T	he case $N = S^2 T^3$: proof of Theorem 1.1	388
5.1.	Depth zero supercuspidal representations	388
5.2.	Simple supercuspidal representations	392
5.3.	Global setup	394
6. L	ocal orbital integrals at primes $p \mid N$ for $N = S^2 T^3$	397
6.1.	Preliminaries when $p \mid T$	398
6.2.	The case where $p \mid T$ and γ is ramified	400
6.3.	The case where $p \mid T$ and γ is unramified	403
6.4.	The case where $p \mid S$	407
7. G	eneral dimension formula, and examples with $N = S^2T^3$	411
7.1.	Dimension formula and root number bias when $N = S^2$	411
7.2.	Simplification when $n = 1$ and $T > 1$	419
7.3.	Global orbital integrals for $n = 1$, $N = T^3$	421
7.4.	Dimension formulas when $N = T^3$	426
7.5.	Some examples with $n > 1$	430
Acknowledgements		436
References		436

1. Introduction

1.1. Overview. Modular forms are holomorphic functions on the complex upper half-plane \mathbb{H} that obey a type of symmetry under the action of $SL_2(\mathbb{Z})$ (or a congruence subgroup) on \mathbb{H} by linear fractional transformations. They belong to the realm of analysis, but this symmetry embodies a deep link with number theory and algebra. Indeed, Langlands' famous functoriality conjecture predicts that there is a precise connection between the algebraic structure of the field \mathbb{Q} of rational numbers (as captured by representations of its absolute Galois group) and spectral properties of automorphic forms (the latter being simultaneous eigenfunctions of the Laplace operator and its p-adic analogs, the Hecke operators) [13]. This connection is expressed as an equality of L-functions.

Automorphic forms can be elusive, and for most purposes it is not feasible to study them and their L-functions one at a time. The trace formula is a technique that provides access to averages of spectral data across families of forms, where the family is determined by a choice of test function. For instance, by choosing a test function with certain invariance properties, one obtains a sum of Hecke eigenvalues $\lambda_n(h)$ for all eigenforms h of a given level and weight, i.e., the trace of the Hecke operator T_n on $S_k(N)$ (see, for example, [23]).

The trace formula and its relative cousins have seen widespread use in analytic number theory, with applications to such problems as estimating moments of L-functions with consequent subconvexity bounds for a single L-function, determining

the asymptotic distribution of the Hecke eigenvalues of a growing family of cusp forms (vertical Sato–Tate laws), and finding densities of low-lying zeros of families *L*-functions (Katz–Sarnak philosophy). See [3] for a recent survey of these and other applications.

Our aim in the present article is to train the trace formula microscope more narrowly through the use of specialized test functions, thereby providing access to thinner families in the automorphic spectrum. This is achieved using the "simple trace formula", variants of which have been in use since the 1970s [15, (7.21)]. Our motivation (described in the next section) is to count cusp forms in these thin families. But the explicit and flexible local-to-global techniques detailed here for GL(2) can be used in many other applications.

Counterintuitively, by considering smaller families, in some situations one obtains simpler trace formulas and stronger analytic results. We mention here some examples that illustrate this. First, Hu [18] and Hu, Petrow and Young [19] have recently developed Fourier relative trace formulas for newforms with certain prescribed local representation types. This is used to estimate thin averages of Rankin–Selberg *L*-functions, leading to improved hybrid subconvexity bounds.

In a different direction, in 2007 Booker and Strömbergsson [4] used the Selberg trace formula to provide evidence for Selberg's conjecture that the first Laplace eigenvalue in the cuspidal spectrum of $\Gamma \setminus \mathbb{H}$ for a congruence subgroup $\Gamma \subseteq \operatorname{SL}_2(\mathbb{Z})$ is $\geq \frac{1}{4}$. In verifying the conjecture for $\Gamma = \Gamma_1(N)$ for square-free N < 857, they observed that the trace formula simplifies upon sieving out the contribution of oldforms in this case. They were also able to restrict to the even (or odd) part of the spectrum. With Lee in [5], they subsequently extended this work to remove the square-free hypothesis on N. However, in this case removing the oldform contribution introduces further complication. To proceed, they developed a novel method to sieve the spectrum down further to twist-minimal newforms, arriving at a simpler formula. In both papers, working with a thinner family extended the reach of their numerical computations.

A general discussion about the value of isolating small families of automorphic forms is given in [45, Section 1.5]. In the breakthrough papers [44; 45], Petrow and Young established Weyl-type subconvexity bounds for Dirichlet *L*-functions using a family of Maass forms that is locally principal series at all finite places.

1.2. Description of main results. Given an integer

$$N = \prod_{p \mid N} p^{N_p} > 1,$$

let $H_k(N)$ be the set of cuspidal Hecke newforms of level N and weight k. Each $h \in H_k(N)$ corresponds to a cuspidal automorphic representation π_h of $G(\mathbb{A}) = G(\mathbb{R}) \prod_p' G(\mathbb{Q}_p)$ where $G = \operatorname{PGL}_2$. The representation π_h factors as a restricted

tensor product

$$\pi_h \cong \bigotimes_{p \le \infty}' \pi_{h,p}$$

of infinite-dimensional irreducible admissible representations of the local groups. We know that $\pi_{h,\infty}=\pi_k$ is the weight k discrete series representation, that for each prime $p\nmid N$, $\pi_{h,p}$ is an unramified principal series representation with Satake parameters determined by the p-th Hecke eigenvalue of h, and that for each $p\mid N$, $\pi_{h,p}$ is ramified of conductor p^{N_p} (see, for example, [12]).

There is an algorithm, due to Loeffler and Weinstein [35], to determine the isomorphism class of each ramified $\pi_{h,p}$ given h. Here we consider the opposite problem, namely to understand the cusp forms h with prescribed local ramification behavior. To this end, we define the following spaces of newforms. For each $p \mid N$, fix an irreducible admissible representation σ_p of $\operatorname{PGL}_2(\mathbb{Q}_p)$ of conductor p^{N_p} , and let $\hat{\sigma} = (\sigma_p)_{p \mid N}$. We then let $H_k(\hat{\sigma})$ be the set of weight k newforms of level N having the local representation type σ_p at each $p \mid N$:

$$H_k(\hat{\sigma}) = \{ h \in H_k(N) \mid \pi_{h,p} \cong \sigma_p \text{ for all } p \mid N \}.$$

Defining

$$S_k(\hat{\sigma}) = \operatorname{Span} H_k(\hat{\sigma}), \quad S_k^{\text{new}}(N) = \operatorname{Span} H_k(N),$$

we have

$$S_k^{\text{new}}(N) = \bigoplus_{\hat{\sigma}} S_k(\hat{\sigma}),$$
 (1-1)

where $\hat{\sigma}$ ranges over all tuples as above.

The dimensions of the spaces $S_k^{\text{new}}(N)$ have been computed by Greg Martin in [37], by sieving the well-known dimension formulas for the full spaces $S_k(N)$. It is an open problem to refine these dimension formulas by computing $\dim S_k(\hat{\sigma}) = |H_k(\hat{\sigma})|$ for each tuple $\hat{\sigma}$. More generally one can ask for the traces of Hecke operators on $S_k(\hat{\sigma})$. A complete solution to this problem seems well out of reach, but even special cases are of great interest. For example, such information would enable investigations into the effect of the underlying representation type on various statistical properties of cusp forms.

In some special cases, asymptotic results about $|H_k(\hat{\sigma})|$ are known. When p is a finite prime, the representation σ_p of $G(\mathbb{Q}_p)$ is either principal series, special, or supercuspidal [7, Section 9.11]. Only the latter two types are square-integrable (assuming unitary central character), and these are amenable to study via the trace formula. Kim, Shin and Templier [21] gave asymptotics for automorphic representations with prescribed supercuspidal local behavior in a quite general setting. In the case of $PGL_2(\mathbb{Q})$, their work shows that if each σ_p is supercuspidal,

$$|H_k(\hat{\sigma})| \sim \frac{1}{12}(k-1) \prod_{p|N} d_{\sigma_p}$$
 (1-2)

as $k, N \to \infty$, where d_{σ_p} is the formal degree of σ_p , suitably normalized. They use the trace formula, and the main technical input is a bound for the elliptic orbital integrals attached to supercuspidal matrix coefficients. In a related earlier work, Weinstein [56] gave asymptotics for cusp forms with prescribed local inertial types, concluding that the set of types lacking global realization is finite. Fixing inertial type is weaker than fixing the local representation, but this result includes types which are not square-integrable. This is discussed further in a recent paper of Dieulefait, Pacetti and Tsaknias [10].

We remark that in Corollary 7.2 we will show that the asymptotic (1-2) is in fact an equality when $k \ge 3$ is odd (so in particular the nebentypus is nontrivial) and N has a prime factor p > 3 with $\operatorname{ord}_p(N)$ odd.

When N is square-free, each σ_p is necessarily special. Going beyond asymptotics, Kimball Martin [38] computed $|H_k(\hat{\sigma})|$ explicitly in this case, by applying Yamauchi's trace formula for Atkin–Lehner operators. As an interesting consequence, he discovered that there is a bias among newforms of square-free level, favoring root number ± 1 : letting $S_k^{\pm}(N)$ denote the span of the newforms of root number ± 1 , we have

$$\dim S_k^+(N) - \dim S_k^-(N) \ge 0,$$

when N is square-free, with the inequality being strict with finitely many explicit exceptions. For example, if N > 3 and k > 2,

$$\dim S_k^+(N) - \dim S_k^-(N) = c_N h(-N), \tag{1-3}$$

where $c_N \in \left\{ \frac{1}{2}, 1, 2 \right\}$ is a constant depending on the equivalence class of N modulo 8, and h(-N) is the class number of $\mathbb{Q}(\sqrt{-N})$.

In the present paper, we further investigate the case where each σ_p is supercuspidal. Our first main result is Theorem 4.2 giving, for such tuples $\hat{\sigma}$, a general formula for the trace of a Hecke operator T_n on $S_k(\hat{\sigma})$ as a main term plus a finite sum of elliptic orbital integrals $\Phi(\gamma, f)$. This theorem is obtained from the adelic GL_2 trace formula using a test function f built using supercuspidal matrix coefficients at the ramified places. In Section 3.3 we show how each global elliptic orbital integral can be factorized into a product of local ones, multiplied by a global measure term that is computed in Theorem 4.16. This global measure is the source of the class numbers of quadratic number fields that appear in classical trace formulas. The local orbital integrals at primes not dividing the level are evaluated explicitly over an arbitrary local field of characteristic 0 in Sections 4.4 and 4.5. We have kept these calculations as general as possible in order that they may find use in other applications of the trace formula.

Theorem 4.2 thereby reduces explicit evaluation of $tr(T_n | S_k(\hat{\sigma}))$ to the calculation of certain local elliptic orbital integrals at the places dividing the level. We

demonstrate proof of concept in Sections 5 and 6 by carrying out the latter in the special case where each σ_p has conductor $\leq p^3$. As recalled in Section 3.1, the supercuspidals come in two series: the unramified supercuspidals, of conductor p^{2r} , and the ramified supercuspidals, of conductor p^{2r+1} . We thus treat the first (r=1) family in each series. The result is the following explicit formula for $\operatorname{tr}(T_n | S_k(\hat{\sigma}))$ under this restriction. We allow nontrivial nebentypus, which requires the tuple $\hat{\sigma}$ to satisfy a global central character constraint described in Section 5.3. Of course, when $\dim S_k(\hat{\sigma}) = 1$ as sometimes happens when k and N are small, it provides a direct way to compute the Fourier coefficients of the associated newform.

Theorem 1.1. Let $N = S^2T^3 > 1$ for S, T relatively prime and square-free, and let ω' be a Dirichlet character of level N and conductor dividing ST. Let $\hat{\sigma} = (\sigma_p)_{p|N}$ be a tuple of supercuspidal representations, with σ_p of conductor p^2 (resp. p^3) if p|S (resp. p|T), chosen compatibly with ω' as in Section 5.3. For k > 2 satisfying $\omega'(-1) = (-1)^k$, let $S_k(\hat{\sigma}) \subseteq S_k^{\text{new}}(N, \omega')$ be the associated space of newforms. Then for (n, N) = 1 and T_n the usual Hecke operator defined in Section 4.1,

$$\begin{aligned} & \operatorname{tr}(T_{\mathbf{n}} \mid S_{k}(\hat{\sigma})) \\ &= \mathbf{n}^{(k/2)-1} \Bigg[\overline{\omega'(\sqrt{\mathbf{n}})} \frac{1}{12} (k-1) \prod_{p \mid S} (p-1) \prod_{p \mid T} \frac{1}{2} (p^{2}-1) \\ &\qquad \qquad + \frac{1}{2} \sum_{M \mid T} \Phi \Bigg(\begin{pmatrix} -\mathbf{n}M \\ 1 \end{pmatrix} \Bigg) + \sum_{M \mid T} \sum_{n \in M} \Phi \Bigg(\begin{pmatrix} 0 & -\mathbf{n}M \\ 1 & rM \end{pmatrix} \Bigg) \Bigg], \end{aligned}$$

where $\omega'(\sqrt{n})$ is taken to be 0 if n is not a perfect square. Each orbital integral $\Phi(\gamma)$ as above may be evaluated explicitly using

$$\Phi(\gamma) = \frac{2h(E)}{w_E 2^{\omega(d_E)}} \Phi_{\infty}(\gamma) \prod_{p \mid N} \Phi_p(\gamma) \prod_{\ell \mid \Delta_{\gamma}, \ell \nmid N} \Phi_{\ell}(\gamma). \tag{1-4}$$

Here, ℓ and p denote prime numbers, Δ_{γ} is the discriminant of the characteristic polynomial of γ , $E = \mathbb{Q}[\gamma]$ is an imaginary quadratic field with class number h(E), discriminant d_E (with $\omega(d_E)$ distinct prime factors) and w_E roots of unity. Given

$$\gamma = \begin{pmatrix} 0 & -nM \\ 1 & rM \end{pmatrix}$$

for $0 \le r < \sqrt{4n/M}$, the factors in (1-4) are given explicitly as follows. Taking $\theta_{\gamma} = \arctan(\sqrt{|\Delta_{\gamma}|}/rM)$ (interpreted as $\frac{\pi}{2}$ if r = 0),

$$\Phi_{\infty}(\gamma) = -\frac{\sin((k-1)\theta_{\gamma})}{\sin(\theta_{\gamma})}$$

(as in Proposition 4.3).

Suppose $\ell \mid \Delta_{\gamma}$ and $\ell \nmid N$. Then if γ is hyperbolic in $G(\mathbb{Q}_{\ell})$,

$$\Phi_{\ell}(\gamma) = |\Delta_{\gamma}|_{\ell}^{-1/2}$$

(as in Proposition 4.4). If γ is elliptic in $G(\mathbb{Q}_{\ell})$, then (as in Proposition 4.8 and (4-20), (4-21))

$$\Phi_{\ell}(\gamma) = e_{\gamma}(\ell) \sum_{i=0}^{\operatorname{ord}_{\ell}(b)} \ell^{j} \left(1 + \frac{2 - e_{\gamma}(\ell)}{\ell} \delta_{j>0} \right),$$

where $\delta_{j>0}$ is an indicator function, $e_{\gamma}(\ell) \in \{1, 2\}$ is 2 if and only if ℓ ramifies in E, and b is defined by $\Delta_{\gamma} = b^2 d_E$ for d_E the discriminant of E.

Suppose $p \mid N$. If γ is hyperbolic in $G(\mathbb{Q}_p)$, then $\Phi_p(\gamma) = 0$. So we will assume that γ is elliptic in $G(\mathbb{Q}_p)$. We consider the three cases $p \mid M$, $p \mid (T/M)$, and $p \mid S$ separately. If $p \mid M$, then $\Phi_p(\gamma) = 0$ unless there exists y such that $y^2 \equiv -pt_p/nM$ mod p, where t_p is the parameter of the fixed supercuspidal representation $\sigma_p = \sigma_{t_p}^{\zeta_p}$ of conductor p^3 (see Section 5.2). In this case,

$$\Phi_p\!\left(\!\!\begin{pmatrix} 0 & -\mathrm{n}M \\ 1 & rM \end{pmatrix}\!\!\right) = \overline{\zeta_p}\!\left[e\!\left(-\frac{yrM}{p^2}\right)\omega_p(y) + \delta(p \neq 2)\,e\!\left(\frac{yrM}{p^2}\right)\omega_p(-y)\right]$$

(as in Proposition 6.4), where ζ_p and ω_p are the root number and central character of σ_p respectively, $e(x) = e^{2\pi i x}$ and δ is an indicator function.

If p|(T/M), then $\Phi_p(\gamma) = 0$ unless the characteristic polynomial P_{γ} of γ has a nonzero double root modulo p, say

$$P_{\nu}(X) \equiv (X - z)^2 \mod p \quad \text{for some } z \in (\mathbb{Z}/p\mathbb{Z})^*.$$
 (1-5)

Under this condition, we have (as in Proposition 6.5 and its remarks)

$$\Phi_p(\gamma) = \frac{\overline{\omega_p(z)}}{p} \sum_{n=1}^{\operatorname{ord}_p(\Delta_\gamma) - 1} \sum_{c \bmod p} \mathcal{N}_\gamma(c, n) \sum_{y=1}^{p-1} e\left(\frac{yc}{zp}\right) e\left(-\frac{t_p}{yzp}\right)^{\delta(n=1)},$$

where $t_p \in (\mathbb{Z}/p\mathbb{Z})^*$ is the parameter of $\sigma_p = \sigma_{t_p}^{\zeta_p}$, ω_p is its central character, $e(x) = e^{2\pi i x}$, and

$$\mathcal{N}_{\gamma}(c, n) = \#\{b \mod p^{n+1} \mid P_{\gamma}(b) \equiv cp^n \mod p^{n+1}\}.$$

Finally, suppose $p \mid S$. If (1-5) is satisfied, then (as in Proposition 6.8),

$$\Phi_p(\gamma) = -\overline{\omega_p(z)} + \frac{\overline{\omega_p(z)}}{p} \sum_{n=1}^{\operatorname{ord}_p(\Delta_\gamma) - 1} \left[(p-1) \mathcal{N}_\gamma(0, n) - \sum_{c=1}^{p-1} \mathcal{N}_\gamma(c, n) \right]$$

for $\mathcal{N}_{\gamma}(c, n)$ as above. On the other hand, if P_{γ} is irreducible modulo p, then

$$\Phi_p(\gamma) = -\overline{\nu(\gamma)} - \overline{\nu^p(\gamma)},$$

where v is the primitive character of $\mathbb{F}_{p^2}^*$ attached to the fixed supercuspidal σ_p of conductor p^2 (see Section 5.1), $\omega_p = v|_{\mathbb{F}_p^*}$, and we interpret the above to mean $-\overline{v(x)} - \overline{v^p(x)}$ if $x \in \mathbb{F}_{p^2}^*$ has the same minimum polynomial over \mathbb{F}_p as the reduction of γ mod p.

Remarks. (1) What we call $S_k(N, \omega')$ is usually called $S_k(N, {\omega'}^{-1})$. See the beginning of Section 4 for explanation. The reason we assume that the conductor of ω' divides ST is that this is necessary for the existence of tuples $\hat{\sigma}$ given the conductor hypotheses, by [55, Proposition 3.4].

- (2) The theorem contains various simple conditions under which an orbital integral as in (1-4) vanishes. These are summarized and established in Proposition 5.6.
- (3) Analytic applications often require uniform bounds for the orbital integrals appearing on the geometric side. Such bounds were established in a much more general context by Kim, Shin and Templier [21, (1.5), (1.6), (1.8)]. Using these, they proved a vertical (fixed p) equidistribution result for p-th Hecke eigenvalues in $S_k(\hat{\sigma})$ as $N \to \infty$, refining the result of Serre [51]. Their paper includes several helpful examples to explain their results in the setting of PGL(2). The explicit formulas for local orbital integrals developed in the present paper illustrate their bounds; see the remarks after Proposition 6.5, for example.
- (4) Although we describe some interesting consequences of Theorem 1.1 below in Section 1.3, perhaps the main utility of this article is the methodology leading to the theorem, rather than this particular trace formula. Indeed, there are any number of variants that one could pursue simply by doing some additional local computations and updating the set of relevant global γ 's on the geometric side:
- One could capture newforms with prescribed representation type at some places, and, less restrictively, prescribed local conductor at some other places. For the latter places, the local elliptic orbital integral calculation is carried out in [28].
- We have excluded the case where $\operatorname{ord}_p(N)=1$ at a prime p for the same reason that we impose k>2: the matrix coefficients of the local representations in such cases are square-integrable but not integrable [23, Proposition 14.3; 53]. So these functions cannot be used directly in the trace formula. One could incorporate these representation types either by using pseudocoefficients [21, Example 6.6; 30; 43], or, via the Jacquet–Langlands correspondence, by computing the corresponding local orbital integrals on a quaternion algebra [22].
- One could capture Maass newforms with prescribed local behavior by taking the archimedean component f_{∞} of the test function to be bi-SO(2)-invariant, as described, for example, in [26, Chapters 3 and 4]. In this case, the inclusion of a supercuspidal matrix coefficient at some place p will annihilate the continuous and residual spectra, but at least two such places would be needed in order to annihilate

the hyperbolic and unipotent terms on the geometric side of the trace formula, as explained in Theorem 3.3 below. Further, in this case γ need no longer be elliptic in $G(\mathbb{R})$ in order to contribute nontrivially, so there are more relevant γ 's that would have to be considered.

• The nonarchimedean local calculations in the present paper are all carried out over arbitrary *p*-adic fields, so with some additional global considerations one could work over a number field.

The technical heart of the paper is Section 6, in which we calculate local elliptic orbital integrals attached to the supercuspidal representations of conductor $\leq p^3$. Character values of supercuspidal representations on various groups appear in many places, but the orbital integral calculations in Section 6 are new. Some related calculations were made by Palm in his doctoral thesis [43, Section 9.11]. Although there are some errors in that work, the methods have been adapted for our computations.

In Section 7 we illustrate Theorem 1.1 by computing dimension formulas and some examples of $tr(T_n | S_k(\hat{\sigma}))$ for n > 1.

1.3. Dimension formulas and root number bias. Upon taking n = 1 in Theorem 4.2, we obtain a general formula for dim $S_k(\hat{\sigma})$, given in Theorem 7.1. As shown there, the list of relevant γ can be narrowed considerably when n = 1; only M = T, $\frac{T}{2}$ contribute to the formula when T > 3. We will state some special cases below, but first we provide some additional motivation.

Simple supercuspidals are the representations of $GL_2(\mathbb{Q}_p)$ with conductor p^3 . Assuming trivial central character, they can be parametrized by the pairs (t, ζ) where $t \in (\mathbb{Z}/p\mathbb{Z})^*$ and $\zeta \in \{\pm 1\}$. There are thus 2(p-1) such representations, denoted σ_t^{ζ} , and each is constructed in the same way via compact induction from a character $\chi_{t,\zeta}$ of a certain open compact-mod-center subgroup H'_t of $GL_2(\mathbb{Q}_p)$.

An interesting question is whether each member of such a local family has the same global multiplicity, in the following sense. For T>1 square-free, consider $N=T^3$ in (1-1), with $\hat{\sigma}$ running over all tuples $(\sigma_{t_p}^{\zeta_p})_{p\mid T}$. (We assume trivial central character for the moment, though the general case is considered in the main body of this paper.) In this case we have the dimension formula

$$\dim S_k^{\text{new}}(T^3) = \frac{1}{12}(k-1) \prod_{p|T} (p^2 - 1)(p-1)$$
 (1-6)

as in [37]. Since there is no immediately apparent reason for nature favoring one simple supercuspidal over another, one might surmise that the subspaces $S_k(\hat{\sigma})$ all have the same dimension, i.e., that the asymptotic (1-2), which in the present situation becomes

$$\dim S_k(\hat{\sigma}) \sim \frac{1}{12}(k-1) \prod_{p \mid T} \frac{1}{2}(p^2 - 1),$$
 (1-7)

is an equality. (Note that the right-hand side of (1-7) results from dividing (1-6) by the number 2(p-1) of simple supercuspidals at each place $p \mid T$.) This would be consistent with a 2011 calculation of Gross [17, p. 1255], who fixed the tuple of parameters $(t_p)_{p\mid N}$ and allowed the ζ_p parameters to vary. Using the trace formula he showed

$$\sum_{(\zeta_p)_{p|T}} \dim S_k((\sigma_{t_p}^{\zeta_p})_{p|T}) = \frac{1}{12}(k-1) \prod_{p|T} (p^2 - 1), \tag{1-8}$$

which is what one would expect, upon dividing (1-6) by the number of tuples $(t_p)_{p|T}$. However, equation (1-7) is *not* in fact an equality in general, for the simple reason that, as we spell out at (5-22), the right-hand side of (1-7) fails to be an integer for infinitely many values of T. This is manifested in recent work of Pi and Qi [46], who considered a sum different from that treated by Gross, namely, varying the t_p and ζ_p parameters subject to the constraint $(-1)^{k/2} \prod_{p|T} \zeta_p = \epsilon$ for fixed $\epsilon \in \{\pm 1\}$. This amounts to counting the newforms with root number ϵ . They found, for $k \ge 4$ even and square-free T > 3, that

$$\dim S_k^{\text{new}}(T^3)^{\pm} = \frac{1}{24}(k-1) \prod_{p|T} (p^2 - 1)(p-1) \pm \frac{1}{2}c_T \,\varphi(T)h(-T), \qquad (1-9)$$

where c_T and h are as in (1-3) and φ is Euler's φ -function. This shows that, just as in the case of square-free level, there is a bias in favor of positive root number. Instead of the Arthur–Selberg trace formula, they used a Petersson formula obtained using the simple supercuspidal new vector matrix coefficient from [27].

By evaluating the S = n = 1 case of Theorem 1.1, in Section 7.4 we obtain an explicit formula for dim $S_k(\hat{\sigma})$ that refines each of the above results. For example, we have the following.

Theorem 1.2. Let $N = T^3$ for T > 3 odd and square-free, let k > 2 be even, and let $\hat{\sigma} = (\sigma_{t_p}^{\zeta_p})_{p|N}$ be a tuple of simple supercuspidal representations with trivial central characters. Then

$$\dim S_k(\hat{\sigma}) = \frac{1}{12}(k-1) \prod_{p|N} \frac{1}{2}(p^2-1) + \Delta(\hat{t}) \,\epsilon(k,\hat{\zeta}) \,b_T \,h(-T), \tag{1-10}$$

where $\Delta(\hat{t}) \in \{0, 1\}$ is nonzero if and only if $-pt_p/T$ is a square modulo p for each $p \mid T$, $\epsilon(k, \hat{\zeta}) = (-1)^{k/2} \prod_{p \mid N} \zeta_p$ is the common global root number of the newforms comprising $H_k(\hat{\sigma})$, b_T is a certain power of 2 depending on $T \mod 8$, and h(-T) is the class number of $\mathbb{Q}[\sqrt{-T}]$.

This is a special case of Theorem 7.17, which also allows for T even. The presence of $\Delta(\hat{t})$ demonstrates that the dimension is not simply a function of the weight, level and root number (even when the right-hand side of (1-7) is an integer). Indeed, as described in [6] for example, each σ_p has attached a ramified quadratic extension of \mathbb{Q}_p , namely $E_{\sigma_p} = \mathbb{Q}_p(\sqrt{t_p p})$, which depends only on the Legendre

symbol (t_p/p) . So dim $S_k(\hat{\sigma})$ depends only on T, k, the fields E_{σ_p} , and the global root number. (If T is even, the dimension also depends on the local root number ζ_2 .)

The second term in (1-10) comes from an elliptic orbital integral. These do not appear in (1-8), but combine to form the error term in (1-9). Indeed, the local root number already appears as a coefficient in our local test function at the places dividing T, so the global root number naturally appears in the elliptic orbital integral that yields the error term in (1-10). This helps explain the positive bias of the root number in this situation.

At the end of Section 7.4, we indicate how our results recover the dimension formula (1-6) and the root number bias (1-9). In Theorem 7.16 we find that the root numbers of newforms of level 27 have a strict bias toward -1 (among the possibilities $\pm 1, \pm i$) when $k \equiv 5 \mod 6$ and the nebentypus has conductor 3.

In a more recent paper, K. Martin [39] addressed the question of root number bias in $S_k^{\text{new}}(N)$ for arbitrary levels. He showed that there is a bias towards root number +1 with one exception: when $N=S^2$ for a square-free number S and $(-1)^{k/2}=-\prod_{p\mid S}(-1)$, then the root number has a strict negative bias when k is sufficiently large. In discussing why the exceptions arise, he noted that the picture is obscured by the existence of newforms of level S^2 which are twists of forms of lower level. (No such forms exist in the $N=T^3$ case discussed above.)

Theorem 1.1 allows us to investigate this further, since the subspace $S_k^{\min}(S^2) \subseteq S_k^{\text{new}}(S^2)$ spanned by the newforms which are not twists of newforms of lower level is the direct sum

$$S_k^{\min}(S^2) = \bigoplus_{\hat{\sigma}} S_k(\hat{\sigma}),$$

ranging over all $\hat{\sigma} = (\sigma_p)_{p|S}$ with each σ_p a supercuspidal representation of conductor p^2 (a "depth zero" supercuspidal) and trivial central character.

In fact, even without using a specialized trace formula, we can infer the existence of negative bias for the root numbers in $S_k^{\min}(S^2)$ for many pairs (S,k) by the following heuristic coming from finite fields (see Section 5.1 for more detail and a summary of the construction of depth zero supercuspidals). Given an odd prime p, there are p-1 primitive characters of $\mathbb{F}_{p^2}^*$ with trivial restriction to \mathbb{F}_p^* . It follows that the number of σ_p as above is $\frac{1}{2}(p-1)$. If $p\equiv 3 \mod 4$, this number is odd, so the set of such σ_p contains a preponderance either of local root number $\epsilon_p=+1$ or $\epsilon_p=-1$. So if S is a product of such primes, then for some integer $c\geq 1$ there are c more tuples $\hat{\sigma}$ with one nonarchimedean sign $\epsilon_{\text{fin}}=\prod_{p\mid S}\epsilon_p$ than the other. By (1-2), the spaces $S_k(\hat{\sigma})$ all have roughly the same dimension $\frac{1}{12}(k-1)\prod_{p\mid S}(p-1)$, up to variations of lower magnitude when k+S is sufficiently large. Then with k/2 of the appropriate parity, there is a bias towards root number $\epsilon=(-1)^{k/2}\epsilon_{\text{fin}}=-1$, with roughly $c\frac{1}{12}(k-1)\prod_{p\mid S}(p-1)$ more forms of global root number -1 than +1. (We will show that in fact c=1; see Proposition 7.6.)

To make a precise statement, we first apply Theorem 1.1 with n = T = 1 to obtain the following.

Theorem 1.3. Let $N = S^2$ for S > 1 square-free, let k > 2 be even, and let $\hat{\sigma} = (\sigma_{\nu_p})_{p|N}$ be a tuple of depth zero supercuspidal representations with trivial central characters, with ν_p the primitive character of $\mathbb{F}_{p^2}^*$ associated to σ_p . Then

$$\dim S_k(\hat{\sigma}) = \frac{1}{12}(k-1) \prod_{p \mid S} (p-1) + D_4(S) \frac{1}{4} \epsilon(k, \hat{\sigma})$$

$$\prod_{\substack{odd \ p \mid S}} 2 + D_3(S) b(k) \frac{1}{3} (-1)^{\delta_{3 \mid S}} \prod_{\substack{p \mid S, p \neq 3}} B(\nu_p), \quad (1-11)$$

where $\epsilon(k, \hat{\sigma})$ is the common global root number of the newforms in $S_k(\hat{\sigma})$, $D_4(S) \in \{0, 1\}$ is 0 if and only if S is divisible by a prime $p \equiv 1 \mod 4$, $D_3(S) \in \{0, 1\}$ is 0 if and only if S is divisible by a prime $p \equiv 1 \mod 3$, δ is the indicator function defined in (2-1),

$$b(k) = \begin{cases} 1 & \text{if } 6 \mid k, \\ -1 & \text{if } k \equiv 2 \mod 6, \\ 0 & \text{otherwise}, \end{cases}$$
 (1-12)

and, for $p \equiv 2 \mod 3$,

$$B(v_p) = \begin{cases} -2 & \text{if the order of } v_p \text{ (in the character group of } \mathbb{F}_{p^2}^* \text{) divides } \frac{1}{3}(p+1), \\ 1 & \text{otherwise.} \end{cases}$$

The above is a special case of Theorem 7.3, which allows for nontrivial nebentypus and k odd. We will use Theorem 1.3 to derive an explicit formula for the bias

$$\Delta(S^2, k)^{\min} = \dim S_k^{\min}(S^2)^+ - \dim S_k^{\min}(S^2)^-$$

for k > 2 even and S > 1 square-free. This is given in Proposition 7.6. For the time being, we just state the following consequence, which is somewhat different from the behavior observed for the larger spaces of newforms of level S^2 appearing in [39, Theorem 1.1(3) and Proposition 1.3].

Proposition 1.4. Assume $k \ge 4$ is even. With notation as above, $\Delta(S^2, k)^{\min} = 0$ in each of the following situations: (i) $D_4(S) = D_3(S) = 0$, (ii) S is divisible by some prime $p \equiv 5 \mod 12$, (iii) $D_4(S) = 0$ and $k \equiv 4 \mod 6$.

If $D_4(S) = 0$, $k \equiv 0$, $2 \mod 6$, $D_3(S) \neq 0$, and case (ii) does not apply, then $\Delta(S^2, k)^{\min} \neq 0$ and

$$\operatorname{sgn} \Delta(S^2, k)^{\min} = (-1)^{\delta(k \equiv 6, 8 \mod 12)} \mu(S)$$

for the indicator δ as in (2-1) and the Möbius function μ .

If $D_4(S) = 1$ and $k \ge 6$, then apart from the two exceptions $S_8^{\min}(2^2) = S_6^{\min}(3^2) = 0$, $\Delta(S^2, k)^{\min} \ne 0$, and

$$\operatorname{sgn} \Delta(S^2, k)^{\min} = (-1)^{\delta(2|S) + k/2}.$$

If $D_4(S) = 1$ and k = 4, then $\Delta(S^2, 4)^{\min} \ge 0$ for all square-free S > 1:

$$\Delta(S^2, 4)^{\min} = \begin{cases} \frac{1}{2} \prod_{p \mid S} (p-1) & \text{if } 2 \nmid S, \\ 0 & \text{if } 2 \mid S. \end{cases}$$

Remark. A noteworthy difference between the above and the bias for the full space of newforms is that here for any fixed even $k \ge 6$ there are infinitely many levels S^2 for which $\Delta(S^2, k)^{\min} < 0$, whereas by [39, Theorem 1.1(3)], for any fixed even k there are only finitely many levels N for which $\Delta(N, k)^{\text{new}} < 0$.

2. Notation and Haar measure

If P is a statement, then we will frequently use the indicator function

$$\delta(P) = \delta_P = \begin{cases} 1 & \text{if } P \text{ is true,} \\ 0 & \text{if } P \text{ is false.} \end{cases}$$
 (2-1)

We also use the shorthand

$$e(x) = e^{2\pi i x}$$

For rings R, we let R^* denote the group of units in R.

Let G be the group GL(2), and set $\overline{G} = G/Z$, where Z is the center of G. If H is a subgroup of G, then \overline{H} will denote the group $HZ/Z \cong H/(H \cap Z)$. For ℓ prime, we set $Z_{\ell} = Z(\mathbb{Q}_{\ell})$ the center and $K_{\ell} = G(\mathbb{Z}_{\ell})$ the maximal compact subgroup of $G(\mathbb{Q}_{\ell})$. Groups $K_0(\mathfrak{p})$, $K_1(\mathfrak{p}^j)$, K' will be defined in Sections 3.1 and 5.2.

Let $\mathbb A$ be the adele ring of the rational numbers $\mathbb Q$. We give $\overline{G}(\mathbb A)$ the standard Haar measure for which

$$\operatorname{meas}(\overline{G}(\mathbb{Q})\backslash \overline{G}(\mathbb{A})) = \frac{\pi}{3},$$

with the discrete group $\overline{G}(\mathbb{Q})$ receiving the counting measure. We normalize Haar measure on $\overline{G}(\mathbb{Q}_{\ell})$ so that $\overline{K_{\ell}}$ has measure 1. With this choice, there is a unique Haar measure on $\overline{G}(\mathbb{R})$ for which the above measure on $\overline{G}(\mathbb{A})$ is the restricted product of the measures on $\overline{G}(\mathbb{Q}_{\ell})$ for $\ell \leq \infty$. It has the form $dm \, dn \, dk$, where dm is the measure $(dx/|x|)^2$ on the diagonal subgroup $M \cong \mathbb{R}^* \times \mathbb{R}^*$, dn is the measure dx on the unipotent subgroup $N \cong \mathbb{R}$, and dk is the measure on $K_{\infty} = \mathrm{SO}(2)$ of total measure 1 [23, Corollary 7.45].

For a unitary Hecke character ω , let $L^2(\omega) = L^2(G(\mathbb{Q}) \setminus G(\mathbb{A}), \omega)$ be the space of (classes of) measurable \mathbb{C} -valued functions ϕ on $G(\mathbb{A})$ transforming under the center by ω and square integrable modulo $Z(\mathbb{A})G(\mathbb{Q})$. Let $L^1(\bar{\omega}) = L^1(G(\mathbb{A}), \bar{\omega})$ be defined in the analogous way; its elements are integrable modulo $Z(\mathbb{A})$.

3. The simple trace formula

3.1. Background on supercuspidal representations of GL(2). Let F be a nonarchimedean local field of characteristic 0 with integer ring \mathcal{O} and prime ideal \mathfrak{p} . In this section only, let $G = GL_2(F)$, B = B(F) the upper-triangular Borel subgroup, N = N(F) the unipotent subgroup of B, M the diagonal subgroup, Z the center, and $K = G(\mathcal{O})$ the standard maximal compact subgroup.

Given a smooth irreducible representation (π, V) of G, it is supercuspidal if it satisfies any of the following equivalent conditions (see, e.g., [7, Sections 9 and 10]):

- V is the span of the vectors of the form $\pi(n)v v$ for $v \in V$ and $n \in N$.
- The matrix coefficients of π are compactly supported modulo the center.
- π is not principal series or special, i.e., not a subquotient of a representation induced from a character of B.

The following property found by Harish-Chandra is crucial in what follows. We sketch a proof here for the reader's convenience, following [52, Section 2.2].

Proposition 3.1. Suppose that π is a supercuspidal representation of G, and that $f(g) = \langle \pi(g) v, v' \rangle$ is a matrix coefficient. Then for all $g, h \in G$,

$$\int_{N} f(gnh) dn = 0. \tag{3-1}$$

Proof. We assume for simplicity that π is unitary, which is always the case if the central character is unitary. Then

$$f(gnh) = \langle \pi(g) \, \pi(n) \, \pi(h) \, v, \, v' \rangle = \langle \pi(n) \, \pi(h) \, v, \, \pi(g^{-1}) \, v' \rangle,$$

so we can assume without loss of generality that g = h = 1. By linearity and the first bullet point above, we may also assume that $v = \pi(n_0) w - w$ for some $w \in V$ and $n_0 \in N$.

Let N(v) be an open compact subgroup of N containing n_0 . Then

$$\int_{N(v)} \pi(n) v \, dn = \int_{N(v)} \pi(n) (\pi(n_0) w - w) \, dn = \int_{N(v)} \pi(n) w \, dn - \int_{N(v)} \pi(n) w \, dn = 0.$$

(By smoothness, there exists an open compact subgroup N' of N(v) that fixes w, so the above integrals are really just finite sums.) Since f has compact support, the support of $f|_N$ is contained in some open compact subgroup N(v) as above. Therefore

$$\int_{N} f(n) dn = \int_{N(v)} \langle \pi(n) v, v' \rangle dn = \left\langle \int_{N(v)} \pi(n) v dn, v' \right\rangle = 0.$$

Corollary 3.2. If f is a matrix coefficient of a supercuspidal representation, then for any $g, h \in G$ and $m \in M$ (M being the diagonal subgroup),

$$\int_{N} f(gn^{-1}mnh) \, dn = 0.$$

Proof. This follows from $n^{-1}mn = mn'$, making a change of variables to integrate over n', and applying the above proposition.

For any supercuspidal representation σ of G, there exists an open and closed subgroup $H \subseteq G$ containing Z, with H/Z compact, and an irreducible representation ρ of H, such that σ is compactly induced from ρ : $\sigma = \operatorname{c-Ind}_H^G(\rho)$. Let $K_0(\mathfrak{p}) = \binom{\mathcal{O}^*}{\mathfrak{p}} \binom{\mathcal{O}}{\mathfrak{O}^*}$ be the Iwahori subgroup of G, and fix a prime element ϖ of \mathcal{O} . Up to conjugacy, there are two maximal compact-mod-center subgroups of G, namely

$$J = \begin{cases} ZK & \text{(the unramified case),} \\ ZK_0(\mathfrak{p}) \cup Z\binom{1}{\varpi} K_0(\mathfrak{p}) & \text{(the ramified case).} \end{cases}$$
(3-2)

The latter is the normalizer of $K_0(\mathfrak{p})$. Without loss of generality, one of these contains H, and we call σ unramified or ramified accordingly. There is a unique ideal \mathfrak{p}^j , called the *conductor* of σ , such that the space of vectors in σ fixed by the group

$$K_1(\mathfrak{p}^j) = \begin{pmatrix} \mathcal{O}^* & \mathcal{O} \\ \mathfrak{p}^j & 1 + \mathfrak{p}^j \end{pmatrix}$$

is one-dimensional. By [55], $j \ge 2$, and as explained in [6], j is even in the unramified case, and odd in the ramified case.

3.2. Simple trace formula. Given a unitary Hecke character ω and a function $f \in L^1(\bar{\omega})$, we define the operator R(f) on $L^2(\omega)$ via

$$R(f)\phi(x) = \int_{\overline{G}(\mathbb{A})} f(g)\phi(xg) dg. \tag{3-3}$$

For k > 2, let \mathcal{C}_k denote the space of all continuous factorizable functions $f = f_{\infty} \prod_{\ell < \infty} f_{\ell}$ on $G(\mathbb{A})$ which transform under the center by $\bar{\omega}$, such that f_{ℓ} is smooth and compactly supported modulo the center Z_{ℓ} for all ℓ , there is a finite set S of places of \mathbb{Q} such that for all $\ell \notin S$, f_{ℓ} is supported on $Z_{\ell}K_{\ell}$ and has the value 1 on K_{ℓ} , and lastly,

$$f_{\infty}\left(\begin{pmatrix} a & b \\ c & d \end{pmatrix}\right) \ll_k \frac{(ad - bc)^{k/2}}{(a^2 + b^2 + c^2 + d^2 + 2|ad - bc|)^{k/2}}.$$

Then $C_k \subseteq L^1(\bar{\omega})$, and we can consider the operators R(f) for such f.

Recall that $\gamma \in G(\mathbb{Q})$ is *elliptic* if its characteristic polynomial is irreducible. This concept is well defined on conjugacy classes and cosets of the center. We will use the following simple trace formula.

¹It should be borne in mind that in standard terminology, all supercuspidals are *ramified* in the sense that they have no K-fixed vector. We are using the word in a different sense here, reflecting the nature of the quadratic extension E/F determined by σ [6].

Theorem 3.3. For $f \in C_k$, suppose that for some finite place v of \mathbb{Q} , f_v is a matrix coefficient of a supercuspidal representation of $G_v = G(\mathbb{Q}_v)$, and therefore by Corollary 3.2 its local hyperbolic orbital integrals vanish identically:

$$\int_{M_v \backslash G_v} f_v \left(g^{-1} \begin{pmatrix} a \\ 1 \end{pmatrix} g \right) dg = 0 \tag{3-4}$$

for all $a \in \mathbb{Q}_v^*$, where M_v is the diagonal subgroup of G_v . Suppose further that (3-4) is also satisfied at a **second** place $w \neq v$ (which may be archimedean). Then

$$\operatorname{tr} R(f) = \operatorname{meas}(\overline{G}(\mathbb{Q}) \setminus \overline{G}(\mathbb{A})) f(1) + \sum_{\mathfrak{o} \text{ elliptic in } \overline{G}(\mathbb{Q})} \Phi(\mathfrak{o}, f),$$

where, for an elliptic conjugacy class $\mathfrak{o} \subseteq \overline{G}(\mathbb{Q})$, the orbital integral is defined by

$$\Phi(\mathfrak{o}, f) = \int_{\overline{G}(\mathbb{Q})\backslash \overline{G}(\mathbb{A})} \sum_{\gamma \in \mathfrak{o}} f(g^{-1}\gamma g) \, dg. \tag{3-5}$$

Proof. See [14, Proposition V.2.1 and Theorem V.3.1]. The idea is that the validity of (3-4) at two distinct places kills off the hyperbolic and (nonidentity) unipotent terms on the geometric side of the Arthur–Selberg trace formula, while the stronger condition (3-1) on f_v also forces the operator R(f) to have purely cuspidal image, so the continuous and residual spectral terms vanish as well. In Gelbart's exposition it is assumed that f is compactly supported, but for $f \in \mathcal{C}_k$ everything still converges absolutely as shown in [23], so the same proof is valid.

3.3. *Factorization of orbital integrals.* Here we explain how to compute elliptic orbital integrals locally. The statements and proofs in this section are applicable over an arbitrary number field, though we express everything in terms of \mathbb{Q} .

For $\gamma \in G(\mathbb{Q})$, let G_{γ} be its centralizer. There are two related groups that will be needed. First, since $Z(\mathbb{Q}) \subseteq G_{\gamma}(\mathbb{Q})$, we may form the quotient, denoted $\overline{G_{\gamma}(\mathbb{Q})}$. Second, the centralizer of γ (or, more accurately, of the coset $\gamma Z(\mathbb{Q})$) in $\overline{G}(\mathbb{Q})$ is denoted $\overline{G}_{\gamma}(\mathbb{Q})$. In general these are distinct subgroups of $\overline{G}(\mathbb{Q})$. This will be clarified in the proof of Lemma 3.4 below.

Giving the discrete group $\overline{G_{\nu}(\mathbb{Q})}$ the counting measure, define

$$\Phi(\gamma, f) = \int_{\overline{G_{\gamma}(\mathbb{Q})} \setminus \overline{G}(\mathbb{A})} f(g^{-1} \gamma g) \, dg.$$

For fixed measures on $\overline{G_{\gamma}(\mathbb{R})}$ and $\overline{G_{\gamma}(\mathbb{Q}_{\ell})}$, we also define the local orbital integrals

$$\Phi(\gamma, f_{\infty}) = \int_{\overline{G_{\gamma}(\mathbb{R})} \setminus \overline{G}(\mathbb{R})} f_{\infty}(g^{-1}\gamma g) \, dg$$

and

$$\Phi(\gamma, f_{\ell}) = \int_{\overline{G_{\gamma}(\mathbb{Q}_{\ell})} \setminus \overline{G}(\mathbb{Q}_{\ell})} f_{\ell}(g^{-1}\gamma g) \, dg.$$

For compatibility, some care must be taken regarding the normalization of measures. See the statement of Proposition 3.5 below.

Lemma 3.4. For an elliptic element $\gamma \in G(\mathbb{Q})$, let \mathfrak{o} be its conjugacy class in $\overline{G}(\mathbb{Q})$. Then with notation as above and in (3-5),

$$\Phi(\mathfrak{o}, f) = \begin{cases} \Phi(\gamma, f) & \text{if } \operatorname{tr} \gamma \neq 0, \\ \frac{1}{2}\Phi(\gamma, f) & \text{if } \operatorname{tr} \gamma = 0. \end{cases}$$

Proof. By definition,

$$\Phi(\mathfrak{o}, f) = \int_{\overline{G}(\mathbb{Q})\backslash \overline{G}(\mathbb{A})} \sum_{\delta \in \overline{G}_{\gamma}(\mathbb{Q})\backslash \overline{G}(\mathbb{Q})} f(g^{-1}\delta^{-1}\gamma \delta g) \, dg = \int_{\overline{G}_{\gamma}(\mathbb{Q})\backslash \overline{G}(\mathbb{A})} f(g^{-1}\gamma g) \, dg.$$

Notice that in the definition of $\Phi(\gamma, f)$, the quotient object is $\overline{G_{\gamma}(\mathbb{Q})}$ rather than $\overline{G}_{\gamma}(\mathbb{Q})$. The former is a subgroup of the latter, and we claim that

$$[\overline{G}_{\gamma}(\mathbb{Q}):\overline{G_{\gamma}(\mathbb{Q})}] = \begin{cases} 1 & \text{if } \mathrm{tr}\,\gamma \neq 0, \\ 2 & \text{if } \mathrm{tr}\,\gamma = 0. \end{cases}$$

The lemma follows immediately from this claim. To prove the claim, note that

$$\overline{G_{\gamma}(\mathbb{Q})} = \{\delta \in G(\mathbb{Q}) \mid \delta^{-1}\gamma \delta = \gamma\}/Z(\mathbb{Q})$$

and

$$\overline{G}_{\gamma}(\mathbb{Q}) = \{ \delta \in G(\mathbb{Q}) \mid \delta^{-1} \gamma \, \delta = z \gamma \text{ for some } z \in \mathbb{Q}^* \} / Z(\mathbb{Q}).$$

For any such z, taking determinants we see that $z^2=1$, so $z=\pm 1$. We also see that $\operatorname{tr} \gamma=z\operatorname{tr} \gamma$, so z=1 if $\operatorname{tr} \gamma\neq 0$, and in this case the two groups are equal, as claimed. On the other hand, if $\operatorname{tr} \gamma=0$, then γ is conjugate in $G(\mathbb{Q})$ to its rational canonical form $\begin{pmatrix} 0 & -\det \gamma \\ 1 & 0 \end{pmatrix}$, and

$$\begin{pmatrix} 1 \\ -1 \end{pmatrix} \begin{pmatrix} 0 & -\det \gamma \\ 1 & 0 \end{pmatrix} \begin{pmatrix} 1 \\ -1 \end{pmatrix} = \begin{pmatrix} 0 & \det \gamma \\ -1 & 0 \end{pmatrix},$$

from which it follows that $\delta^{-1}\gamma \delta = -\gamma$ has a solution δ . Given any such δ , we find easily that

$$\overline{G}_{\gamma}(\mathbb{Q}) = \overline{G_{\gamma}(\mathbb{Q})} \cup \delta \overline{G_{\gamma}(\mathbb{Q})}.$$

Proposition 3.5. Let $f \in C_k$ as defined in Section 3.2, and let $\gamma \in G(\mathbb{Q})$ be an elliptic element. Then for any fixed choice of Haar measures on $\overline{G}(\mathbb{A})$ and $\overline{G_{\gamma}(\mathbb{A})}$,

$$\Phi(\gamma, f) = \operatorname{meas}(\overline{G_{\gamma}(\mathbb{Q})} \setminus \overline{G_{\gamma}(\mathbb{A})}) \prod_{\ell < \infty} \Phi(\gamma, f_{\ell}), \tag{3-6}$$

where the measures on the groups $\overline{G}(\mathbb{Q}_{\ell})$ are chosen (noncanonically) so that the measure on $\overline{G}(\mathbb{A})$ is the restricted product of these local measures relative to the maximal compact subgroups almost everywhere, and likewise the measures on the groups $\overline{G_{\gamma}(\mathbb{Q}_{\ell})}$ are chosen compatibly with the fixed measure on $\overline{G_{\gamma}(\mathbb{A})} = \prod_{\ell < \infty}' \overline{G_{\gamma}(\mathbb{Q}_{\ell})}$.

Remarks. (1) This is well known, but as we have not found a proof in the literature, we include one below. Tate's thesis shows that if the product is absolutely convergent, then the left-hand integral converges absolutely and the equality holds. But here we need a kind of converse: we know a priori that $\Phi(\gamma, f)$ is absolutely convergent.

(2) The specific choice of measures to be used in this paper is summarized in Section 4.7.3, where it is shown that the quotient space $\overline{G_{\gamma}(\mathbb{Q})} \backslash \overline{G_{\gamma}(\mathbb{A})}$ is compact, and its measure is computed explicitly in the more general setting with \mathbb{Q} replaced by an arbitrary number field.

Proof. Observe that

$$\begin{split} \Phi(\gamma,f) &= \int_{\overline{G_{\gamma}(\mathbb{Q})} \backslash \overline{G}(\mathbb{A})} f(g^{-1}\gamma g) \, dg \\ &= \operatorname{meas}(\overline{G_{\gamma}(\mathbb{Q})} \backslash \overline{G_{\gamma}(\mathbb{A})}) \int_{\overline{G_{\gamma}(\mathbb{A})} \backslash \overline{G}(\mathbb{A})} f(g^{-1}\gamma g) \, dg \end{split}$$

for any choice of Haar measure on $\overline{G_{\gamma}}(\mathbb{A})$. Absolute convergence is proven for $f \in \mathcal{C}_k$ in [23, Corollary 19.3].

For notational convenience, write $\phi(g) = f(g^{-1}\gamma g)$ (a function on $\overline{G}(\mathbb{A})$), and $\phi_{\ell}(g_{\ell}) = f_{\ell}(g_{\ell}^{-1}\gamma g_{\ell})$ for $\ell \leq \infty$, so $\phi(g) = \prod_{\ell < \infty} \phi_{\ell}(g_{\ell})$. Also, define

$$X = \overline{G_{\gamma}(\mathbb{A})} \setminus \overline{G}(\mathbb{A}).$$

Then *X* is the restricted product of the spaces

$$X_{\ell} = \overline{G_{\nu}(\mathbb{Q}_{\ell})} \setminus \overline{G}(\mathbb{Q}_{\ell}),$$

relative to the open compact subsets $H_{\ell} = \overline{G_{\gamma}(\mathbb{Q}_{\ell})} \setminus \overline{G_{\gamma}(\mathbb{Q}_{\ell})} K_{\ell} \subseteq X_{\ell}$. Indeed, the natural map from $\overline{G}(\mathbb{A})$ to $\prod' X_{\ell}$ is clearly surjective, with kernel $\overline{G_{\gamma}(\mathbb{A})}$.

Fix Haar measures on each of the local groups $\overline{G}(\mathbb{Q}_{\ell})$ and $\overline{G_{\gamma}}(\mathbb{Q}_{\ell})$ compatibly with the fixed Haar measures on $\overline{G}(\mathbb{A})$ and $\overline{G_{\gamma}}(\mathbb{A})$. This determines a right- $\overline{G}(\mathbb{Q}_{\ell})$ -invariant measure on X_{ℓ} with the property that H_{ℓ} has measure 1 for almost all ℓ . Let S be the finite set of places of \mathbb{Q} outside of which f_{ℓ} is supported on $Z(\mathbb{Q}_{\ell})K_{\ell}$ with $f_{\ell}(zk) = \overline{\omega}(z)$. Let S' be a finite set of places outside of which $(1) \gamma \in K_{\ell}$, and $(2) H_{\ell}$ has measure 1. Then setting $S_0 = S \cup S'$, for $\ell \notin S_0$ we have

$$\int_{H_{\ell}} \phi_{\ell}(h) \, dh = \int_{H_{\ell}} f_{\ell}(k^{-1}\gamma k) \, dk = \text{meas}(H_{\ell}) = 1.$$

Let

$$S_0 \subseteq S_1 \subseteq S_2 \subseteq \dots$$

be a sequence of finite sets of primes (including ∞) whose union is the full set of primes. Let χ_n be the characteristic function of $X_{S_n} = \prod_{\ell \in S_n} X_\ell \times \prod_{\ell \notin S_n} H_\ell$, and

let $\phi_n = \phi \cdot \chi_n$. Note that $\phi_n \to \phi$ pointwise. Since $\phi \in L^1(X)$ as mentioned above, so is ϕ_n . By the dominated convergence theorem,

$$\int_X \phi(x) dx = \lim_{n \to \infty} \int_X \phi_n(x) dx = \lim_{n \to \infty} \prod_{\ell \in S_n} \int_{X_\ell} \phi_\ell(x_\ell) dx_\ell,$$

as needed. \Box

4. Counting locally supercuspidal newforms

Here we explain how to use the simple trace formula to count cusp forms with prescribed supercuspidal ramification. To set notation, let $N = \prod_{p \mid N} p^{N_p} > 1$ be a positive integer with the property that $N_p \geq 2$ for each prime $p \mid N$. Fix a Dirichlet character ω' modulo N of conductor dividing $\prod_{p \mid N} p^{\lfloor N_p/2 \rfloor}$. This requirement comes from the fact that the central character of a supercuspidal representation of conductor p^{N_p} divides $p^{\lfloor N_p/2 \rfloor}$ [55, Proposition 3.4]. Let $\omega : \mathbb{A}^* \to \mathbb{C}^*$ be the finite order Hecke character associated to ω' via

$$\mathbb{A}^* = \mathbb{Q}^*(\mathbb{R}^+ \times \hat{\mathbb{Z}}^*) \to \hat{\mathbb{Z}}^*/(1 + N\hat{\mathbb{Z}}) \cong (\mathbb{Z}/N\mathbb{Z})^* \to \mathbb{C}^*, \tag{4-1}$$

where the last arrow is ω' . Letting ω_p be the restriction of ω to \mathbb{Q}_p^* , for any prime $p \mid N$ we have

$$\omega_p(p) = \omega(1, \dots, 1, p, 1, \dots)$$

= $\omega(p^{-1}, \dots, p^{-1}, 1, p^{-1}, \dots) = \prod_{\ell \mid N, \ell \neq p} \omega_\ell(p^{-1}).$ (4-2)

Fix an integer $k \ge 2$ satisfying

$$\omega'(-1) = (-1)^k$$
,

and let $S_k(N, \omega')$ be the space of cusp forms h satisfying

$$h\left(\frac{az+b}{cz+d}\right) = \omega'(d)^{-1}(cz+d)^k h(z)$$

for $\binom{a\ b}{c\ d} \in \Gamma_0(N)$. The inverse on $\omega'(d)$ is somewhat nonstandard. It ensures that the adelic cusp form attached to h has central character ω rather than ω^{-1} ; see, e.g., [23, Sections 12.2–12.4]. Because we mostly work in the adelic setting, it eases the notation to include the inverse in the classical setting.

For each $p \mid N$, fix a supercuspidal representation σ_p of $GL_2(\mathbb{Q}_p)$ of conductor p^{N_p} and central character ω_p , and let $\hat{\sigma} = \{\sigma_p\}_{p \mid N}$. We define $H_k(\hat{\sigma})$ to be the set of newforms $h \in S_k(N, \omega')$ whose associated cuspidal representation π_h has the local representation type σ_p at each $p \mid N$. We set $S_k(\hat{\sigma}) = \operatorname{Span} H_k(\hat{\sigma})$. The Dirichlet character ω' is uniquely determined by the tuple $\hat{\sigma}$ via

$$\omega'(d) = \prod_{p|N} \omega_p(d) \quad ((d,N) = 1), \tag{4-3}$$

and this justifies our suppression of the central character ω' from the notation $S_k(\hat{\sigma})$. At a certain point we will use the fact that

$$\prod_{p \mid N} \omega_p(N) = \omega(N^{-1}, \dots, N^{-1}, 1, \dots, 1, N^{-1}, N^{-1}, \dots) = \omega'(1) = 1. \quad (4-4)$$

4.1. Isolating $S_k(\hat{\sigma})$ spectrally. For each prime $p \mid N$, we can write $\sigma_p = \text{c-Ind}_{H_p}^{G_p} \rho$, where H_p is contained either in $Z_p K_p$ or the normalizer of an Iwahori subgroup, as in (3-2). By [29, Proposition 2.1], there exists a unit vector w_p in the space of σ_p such that the matrix coefficient $\langle \sigma_p(g) w_p, w_p \rangle$ is supported in H_p . Fix once and for all such a vector w_p for each $p \mid N$. Based on this choice, we define a subspace $A_k(\hat{\sigma}) \subseteq L^2(\omega)$ by

$$A_k(\hat{\sigma}) = \bigoplus_{\pi} \mathbb{C}w_{\pi},$$

where π ranges over the cuspidal automorphic representations with central character ω for which $\pi_{\infty} = \pi_k$, $\pi_p = \sigma_p$ for each $p \mid N$, and π_ℓ is unramified for all finite primes $\ell \nmid N$, and $w_{\pi} = \otimes w_{\pi_\ell}$ is defined by

$$w_{\pi_{\ell}} = \begin{cases} \text{unit lowest weight vector} & \text{if } \ell = \infty, \\ \text{unit spherical vector} & \text{if } \ell \nmid N\infty, \\ w_p \text{ (fixed above)} & \text{if } \ell = p \mid N. \end{cases}$$
 (4-5)

Here, for almost all ℓ , the spherical vector is the one predetermined by the restricted tensor product $\pi \cong \bigotimes_{\ell \leq \infty}' \pi_{\ell}$. The space $A_k(\hat{\sigma})$ does not consist of adelic newforms in general because at places $p \mid N$, w_p is not necessarily a new vector in the space of the local representation σ_p . Nevertheless, $A_k(\hat{\sigma})$ has the same dimension as the space of newforms $S_k(\hat{\sigma}) = \operatorname{Span} H_k(\hat{\sigma})$.

Using matrix coefficients, we can define a test function $f \in L^1(\bar{\omega})$ for which R(f) is the orthogonal projection of $L^2(\omega)$ onto $A_k(\hat{\sigma})$. Without much extra work, we can incorporate a Hecke operator into the test function.

Fix an integer n > 1 with gcd(n, N) = 1, and let T_n be the classical Hecke operator defined by

$$T_{\mathbf{n}}h(z) = \mathbf{n}^{k-1} \sum_{\substack{ad=\mathbf{n} \\ a>0}} \sum_{r \bmod d} \omega'(a)^{-1} d^{-k} h\left(\frac{az+r}{d}\right) \quad (h \in S_k(N, \omega'), \ z \in \mathbb{H}).$$

When n = 1, T_n is simply the identity operator.

The operator T_n can be realized adelically. Let

$$M(\mathbf{n})_{\ell} = \{ g \in M_2(\mathbb{Z}_{\ell}) \mid \det g \in \mathbf{n}\mathbb{Z}_{\ell}^* \}$$

for each prime $\ell \nmid N$. (If working over a larger number field F, one would take n to be an ideal of the integer ring and set $M(n)_v = \{g \in M_2(\mathcal{O}_v) \mid (\det g) \mathcal{O}_v = n\}$ for a

place $v < \infty$.) Define a function $f_{\ell}^{\mathbf{n}} : G(\mathbb{Q}_{\ell}) \to \mathbb{C}$ by

$$f_{\ell}^{\mathbf{n}}(g) = \begin{cases} \overline{\omega_{\ell}(z)} & \text{if } g = zm \text{ for } z \in Z_{\ell}, \ m \in M(\mathbf{n})_{\ell}, \\ 0 & \text{if } g \notin Z_{\ell}M(\mathbf{n})_{\ell}, \end{cases}$$
(4-6)

where ω_{ℓ} is the local component of the Hecke character ω . Note that f_{ℓ}^{n} is bi- K_{ℓ} -invariant, and indeed when $n \in \mathbb{Z}_{\ell}^{*}$, this function is given by

$$f_{\ell}(g) = \begin{cases} \overline{\omega_{\ell}(z)} & \text{if } g = zk \in Z_{\ell}K_{\ell}, \\ 0 & \text{if } g \notin Z_{\ell}K_{\ell}. \end{cases}$$
 (4-7)

Next, let π_k be the discrete series representation of $\overline{G}(\mathbb{R})$ of weight k, and let v be a lowest weight unit vector in the space of π_k . We define $f_{\infty} = d_k \langle \overline{\pi_k(g) v}, v \rangle$, where $d_k = \frac{k-1}{4\pi}$ is the formal degree of π_k . Explicitly, with Haar measure on $\overline{G}(\mathbb{R})$ normalized as in Section 2,

$$f_{\infty}\begin{pmatrix} \begin{pmatrix} a & b \\ c & d \end{pmatrix} \end{pmatrix} = \begin{cases} \frac{k-1}{4\pi} \frac{(ad-bc)^{k/2}(2i)^k}{(-b+c+(a+d)i)^k} & \text{if } ad-bc > 0, \\ 0 & \text{otherwise} \end{cases}$$
(4-8)

[23, Theorem 14.5]. This function is integrable over $\overline{G}(\mathbb{R})$ exactly when k > 2, so the latter will be assumed throughout. It would be possible to treat the k = 2 case by using a pseudocoefficient of π_k , but we have not attempted to carry this out (see [43]).

At places $p \mid N$, define

$$f_p(g) = d_{\sigma_p} \langle \overline{\sigma_p(g) w_p, w_p} \rangle,$$
 (4-9)

where d_{σ_p} is the formal degree and w_p is the unit vector fixed above. The formal degree depends on a choice of Haar measure on $\overline{G}(\mathbb{Q}_p)$, which we normalize as in Section 2. By our choice of w_p , the support of f_p is contained in one of the two groups (3-2), according to whether or not σ_p is ramified.

Finally, we define the global test function

$$f^{\mathbf{n}} = f_{\infty} \prod_{p \mid N} f_p \prod_{\ell \nmid N} f_{\ell}^{\mathbf{n}}$$
(4-10)

for f_{∞} of weight k as in (4-8), f_{p} as in (4-9), and f_{ℓ}^{n} as in (4-6).

Proposition 4.1. With the above definition of f^n , the operator $R(f^n)$ (defined in (3-3) taking Haar measure on $\overline{G}(\mathbb{A})$ as normalized in Section 2) factors through the orthogonal projection onto the finite dimensional subspace $A_k(\hat{\sigma})$. On this space, $R(f^n)$ acts diagonally, with the vectors w_{π} being eigenvectors. In more detail, given a newform $h \in H_k(\hat{\sigma})$ with $T_n h = a_n(h)h$, let $w \in A_k(\hat{\sigma})$ be the vector associated to π_h as in (4-5). Then

$$R(f^{n}) w = n^{1-(k/2)} a_{n}(h) w.$$

Consequently,

$$\operatorname{tr}(T_{\mathbf{n}} | S_k(\hat{\sigma})) = \mathbf{n}^{(k/2)-1} \operatorname{tr} R(f^{\mathbf{n}}).$$

- **Remarks.** (1) The vector w is defined only up to unitary scaling, but of course the eigenvalue is independent of the choice.
- (2) One can also take f_p to be the complex conjugate of the trace of the representation ρ inducing σ_p , if normalized correctly. See Proposition 5.5 and its remark.

Proof. The first statement is proven in [25, Proposition 2.3], but we need to reproduce some of the argument here for the second part. Let $h \in H_k(\hat{\sigma})$, let π be the associated cuspidal representation, and let $w = w_{\pi} \in A_k(\hat{\sigma})$. For each place $v \mid \infty N$, the test function f_v was chosen so that

$$\pi_v(f_v) w_v = w_v$$

[23, Corollary 10.26]. Write

$$w = w_{\infty} \otimes \bigotimes_{p \mid N} w_p \otimes w' \otimes \bigotimes_{\ell \mid \mathbf{n}} w_{\ell},$$

where $w' = \bigotimes_{\ell \nmid N_n} w_\ell$. We may likewise decompose π as

$$\pi = \pi_{\infty} \otimes \bigotimes_{p \mid N} \pi_p \otimes \pi' \otimes \bigotimes_{\ell \mid \mathbf{n}} \pi_{\ell},$$

where π' is a representation of $G' = \prod_{p \nmid Nn}' G(\mathbb{Q}_p)$. Then letting $f' = \prod_{p \nmid Nn} f_p$, it is elementary to show that $\pi'(f')w' = w'$. Therefore (by [23, Proposition 13.17])

$$\begin{split} R(f^{\mathbf{n}}) \, w &= \pi_{\infty}(f_{\infty}) \, w_{\infty} \otimes \bigotimes_{p \mid N} \pi_{p}(f_{p}) \, w_{p} \otimes \pi'(f') \, w' \otimes \bigotimes_{\ell \mid \mathbf{n}} \pi_{\ell}(f_{\ell}^{\mathbf{n}}) \, w_{\ell} \\ &= w_{\infty} \otimes \bigotimes_{p \mid N} w_{p} \otimes w' \otimes \bigotimes_{\ell \mid \mathbf{n}} \pi_{\ell}(f_{\ell}^{\mathbf{n}}) \, w_{\ell}. \end{split}$$

Since w_{ℓ} is an unramified unit vector in the principal series representation $\pi_{\ell} = \pi(\chi_1, \chi_2)$ (say), we have $\pi_{\ell}(f_{\ell}^n) w_{\ell} = \lambda_{\ell} w_{\ell}$ for

$$\lambda_{\ell} = \ell^{a/2} \sum_{j=0}^{a} \chi_{1}(\ell)^{j} \chi_{2}(\ell)^{a-j}, \quad a = \text{ord}_{\ell}(\mathbf{n})$$

(see, e.g., [24, Proposition 4.4]). Thus $R(f^n)w = \lambda w$, where $\lambda = \prod_{\ell \mid n} \lambda_\ell$. The result now follows by the well-known fact that $\prod_{\ell \mid n} \lambda_\ell = n^{1-k/2} a_n(h)$. The latter may be proven as follows. If we let v (denoted φ_h in [23]) be the adelic new vector attached to h, then v is a pure tensor, differing from w only at the places $p \mid N$. A test function \tilde{f}^n , say, is used in [23] that differs from f^n only at the places $p \mid N$. By the same argument as above,

$$R(\tilde{f}^{\mathbf{n}})v = v_{\infty} \otimes \bigotimes_{p \mid N} v_p \otimes v' \otimes \bigotimes_{\ell \mid \mathbf{n}} R(f_{\ell}^{\mathbf{n}})v_{\ell}.$$

Since $v_{\ell} = w_{\ell}$ at places $\ell \mid n$, the eigenvalues are the same, i.e., $R(\tilde{f}^{n})v = \lambda v$. By [23, Theorem 13.14] (which uses a global argument), $\lambda = n^{1-k/2}a_{n}(h)$.

4.2. First main result: the trace of a Hecke operator. We now state our first main theorem, which is a general formula for the trace of T_n on $S_k(\hat{\sigma})$. Its proof will occupy the remainder of Section 4.

Theorem 4.2. Let k > 2, let the level N, nebentypus ω' , and tuple $\hat{\sigma} = (\sigma_p)_{p|N}$ of supercuspidals be fixed as at the beginning of Section 4 (ensuring compatibility of central characters with ω'), and let $f = f^n$ as in (4-10). Let T be the product of all primes p|N with $\operatorname{ord}_p(N)$ odd. Then

$$\operatorname{tr}(T_{\mathbf{n}} | S_{k}(\hat{\sigma})) = \mathbf{n}^{(k/2)-1} \left[\overline{\omega'(\mathbf{n}^{1/2})} \frac{1}{12} (k-1) \prod_{p \mid N} d_{\sigma_{p}} + \frac{1}{2} \sum_{M \mid T} \Phi\left(\begin{pmatrix} -\mathbf{n}M \\ 1 \end{pmatrix}, f \right) + \sum_{M \mid T} \sum_{1 \leq r < \sqrt{4\mathbf{n}/M}} \Phi\left(\begin{pmatrix} 0 & -\mathbf{n}M \\ 1 & rM \end{pmatrix}, f \right) \right],$$

where $\omega'(n^{1/2})$ is taken to be 0 if n is not a perfect square, d_{σ_p} is the formal degree of σ_p relative to Haar measure fixed in Section 2, and the orbital integrals $\Phi(\gamma, f)$ are defined in Section 3.3.

An orbital integral $\Phi(\gamma, f)$ as above vanishes unless γ is elliptic in $G(\mathbb{Q}_p)$ for each $p \mid N$. Assuming this condition is satisfied, let $E = \mathbb{Q}[\gamma]$ be the imaginary quadratic extension of \mathbb{Q} generated by γ , and let h(E), w(E), and d_E be the class number, number of units, and discriminant of E respectively. Then

$$\Phi(\gamma, f) = -\frac{2h(E)}{w(E)2^{\omega(d_E)}} \frac{\sin((k-1)\theta_{\gamma})}{\sin(\theta_{\gamma})} \prod_{p|\Delta_{\gamma}N} \Phi(\gamma, f_p), \tag{4-11}$$

where Δ_{γ} is the discriminant of γ , $\theta_{\gamma} = \arctan(\sqrt{|\Delta_{\gamma}|}/\operatorname{tr}\gamma)$ (interpreted as $\frac{\pi}{2}$ if $\operatorname{tr}\gamma = 0$) is the argument of one of the complex eigenvalues of γ , $\omega(d_E)$ is the number of prime factors of d_E , and our choice of measure for $\Phi(\gamma, f_p)$ is summarized in Section 4.7.3 below.

Remarks. (1) For primes $p \nmid N$, the local orbital integrals $\Phi(\gamma, f_p)$ are computed explicitly in Sections 4.4 and 4.5 below. Thus, for the explicit calculation of $\operatorname{tr}(T_n | S_k(\hat{\sigma}))$ it only remains to calculate the local orbital integrals $\Phi(\gamma, f_p)$ for $p \mid N$.

(2) When n = 1, the set of relevant γ is considerably smaller than what appears above if T > 1, due to local considerations at $p \mid T$. See Theorem 7.1.

The proof of Theorem 4.2 involves results from the rest of Section 4, outlined as follows. First, the test function f satisfies the hypotheses of Theorem 3.3. Indeed, the hyperbolic orbital integrals of f_{∞} vanish as shown in [23, Proposition 24.2], and the fact that $f \in \mathcal{C}_k$ is a consequence of the formula for f_{∞} (see [23, Lemma 14.2]).

Since we are normalizing measure so that meas $(\overline{G}(\mathbb{Q})\backslash \overline{G}(\mathbb{A})) = \frac{\pi}{3}$, the identity term in Theorem 3.3 is

$$\frac{\pi}{3}f(1) = \frac{1}{12}(k-1) \prod_{p|N} d_{\sigma_p} \prod_{\ell|n} f_{\ell}^{n}(1).$$

From the definition (4-6) of f_{ℓ}^{n} , we see that $f_{\ell}^{n}(1) \neq 0$ only if $1 \in Z_{\ell}M(n)_{\ell}$, which holds if and only if n is a perfect square. Assuming this is the case,

$$f_{\ell}^{\mathbf{n}}(1) = f_{\ell}^{\mathbf{n}}\left(\begin{pmatrix} \sqrt{\mathbf{n}} & \\ & \sqrt{\mathbf{n}} \end{pmatrix}^{-1} \begin{pmatrix} \sqrt{\mathbf{n}} & \\ & \sqrt{\mathbf{n}} \end{pmatrix}\right) = \omega_{\ell}(\sqrt{\mathbf{n}}).$$

Note that by (4-3),

$$\prod_{\ell \mid \mathbf{n}} \omega_\ell(\sqrt{\mathbf{n}}) = \prod_{\ell \nmid N} \omega_\ell(\sqrt{\mathbf{n}}) = \prod_{\ell \mid N} \overline{\omega_\ell(\sqrt{\mathbf{n}})} = \overline{\omega'(\sqrt{\mathbf{n}})}.$$

Therefore the identity term is

$$\frac{\pi}{3}f(1) = \overline{\omega'(\sqrt{\mathbf{n}})} \frac{1}{12}(k-1) \prod_{p \mid N} d_{\sigma_p},$$

where it is to be understood that $\omega'(\sqrt{n}) = 0$ if n is not a perfect square.

The structure of the first part of Theorem 4.2 is then immediate from Theorem 3.3, Lemma 3.4, and Proposition 4.1. The set of relevant γ is determined in Section 4.6 below, simply by considering the supports of the local test functions. The vanishing of $\Phi(\gamma, f)$ if γ is hyperbolic in $G(\mathbb{R})$ or $G(\mathbb{Q}_p)$ for some $p \mid N$ is explained in Proposition 4.3 below.

As for (4-11), the first factor is equal to meas($\overline{G_{\gamma}(\mathbb{Q})} \setminus \overline{G_{\gamma}(\mathbb{A})}$) under our normalization of Haar measures on $G(\mathbb{A})$ and $G_{\gamma}(\mathbb{A})$. This is shown in Theorem 4.16 below. The second factor of (4-11) (along with the negative sign) is $\Phi(\gamma, f_{\infty})$ as in (4-12) below. In Sections 4.4 and 4.5 we explicitly compute the local orbital integrals away from the level, and see in particular that the value is 1 at places not dividing $\Delta_{\gamma}N$.

The local orbital integrals at the places dividing N of course depend on the choice of supercuspidal representations. The method we use to treat the special cases of simple supercuspidals and depth zero supercuspidals in the second part of this paper is presumably applicable to other cases as well.

4.3. *Known results about the elliptic terms.* We record here some basic properties of the elliptic orbital integrals that arise in Theorem 4.2.

Proposition 4.3. Let γ be elliptic in $G(\mathbb{Q})$. Then for the test function $f = f^n$ of (4-10):

- (1) $\Phi(\gamma, f)$ is absolutely convergent.
- (2) $\Phi(\gamma, f)$ depends only on the conjugacy class of γ in $G(\mathbb{A})$ (rather than in $G(\mathbb{Q})$), and likewise for any prime ℓ , $\Phi(\gamma, f_{\ell})$ depends only on the $G(\mathbb{Q}_{\ell})$ -conjugacy class of γ .

- (3) $\Phi(\gamma, f) = 0$ unless: $\det \gamma > 0$ and γ is elliptic both in $G(\mathbb{R})$ and in $G(\mathbb{Q}_p)$ for each $p \mid M$.
- (4) If γ is elliptic in $G(\mathbb{R})$ with a complex eigenvalue $\rho = re^{i\theta}$, then

$$\Phi(\gamma, f_{\infty}) = -r^{2-k} \frac{\rho^{k-1} - \bar{\rho}^{k-1}}{\rho - \bar{\rho}}
= -\frac{e^{i(k-1)\theta} - e^{-i(k-1)\theta}}{e^{i\theta} - e^{-i\theta}} = -\frac{\sin((k-1)\theta)}{\sin(\theta)}.$$
(4-12)

Remarks. If γ has discriminant $\Delta_{\gamma} < 0$ and nonzero trace, then we may take $\theta = \arctan(\sqrt{|\Delta_{\gamma}|}/\operatorname{tr}\gamma)$ in (4-12). If γ has the form $\binom{u}{1}$, then we may take $\theta = \frac{\pi}{2}$, giving

$$\Phi\left(\begin{pmatrix} u \\ 1 \end{pmatrix}, f_{\infty}\right) = -\left[\frac{i^{k-1} - (-i)^{k-1}}{2i}\right] = \begin{cases} (-1)^{k/2} & \text{if } k \text{ is even,} \\ 0 & \text{if } k \text{ is odd.} \end{cases}$$
(4-13)

Proof. Nearly everything is proven in [23, pp. 295–302]. The only remaining point is that $\Phi(\gamma, f_p) = 0$ if γ is hyperbolic in $G(\mathbb{Q}_p)$ for some $p \mid M$. For such γ , after conjugating we can take γ diagonal, so $G_{\gamma}(\mathbb{Q}_p) = M(\mathbb{Q}_p)$. The orbital integral is then taken over $M_p \setminus G_p$ and involves integrating over $N(\mathbb{Q}_p)$ (see (4-15) below). We can use (3-1) to show that it vanishes (as in (3-4)).

4.4. Local orbital integrals at primes $\ell \nmid N$: hyperbolic case. If $\gamma \in G(\mathbb{Q})$ is elliptic, then for each prime ℓ , γ is either hyperbolic or elliptic in $G(\mathbb{Q}_{\ell})$. In this section and the next we evaluate the local elliptic orbital integrals at primes $\ell \nmid N$. The methods are standard and the results are presumably not new. For the dimension formulas we require the test function f_{ℓ} given by (4-7). However, without any extra work we can consider a general local Hecke operator, and consider an arbitrary p-adic field.

Thus, we let F be a p-adic field with valuation v, uniformizer ϖ , ring of integers \mathcal{O}_F , maximal ideal $\mathfrak{p} = \varpi \mathcal{O}_F$, and $q_v = |\mathcal{O}_F/\mathfrak{p}|$. Fix an unramified unitary character $\omega_v : F^* \to \mathbb{C}^*$. For an integral ideal $\mathfrak{n}_v \subseteq \mathcal{O}_F$, define

$$M(\mathfrak{n}_v) = \{g \in M_2(\mathcal{O}_F) \mid (\det g) \mathcal{O}_F = \mathfrak{n}_v\}$$

and

$$f^{\mathfrak{n}_{v}}(g) = \begin{cases} \overline{\omega_{v}(z)} & \text{if } g = zm \in Z(F)M(\mathfrak{n}_{v}), \\ 0 & \text{if } g \notin Z(F)M(\mathfrak{n}_{v}). \end{cases}$$
(4-14)

If γ is hyperbolic in G(F), then replacing it by a conjugate if necessary, we can assume that it is diagonal. In this case, $G_{\gamma}(F) = M(F)$ is the set of invertible diagonal matrices. We may integrate over $\overline{G}(F)$ using the Iwasawa coordinates

$$\int_{\overline{G}(F)} \phi(g) dg = \int_{\overline{M}(F)} \int_{N(F)} \int_{K_v} \phi(mnk) dm dn dk,$$

where $K_v = G(\mathcal{O}_F)$. Therefore if ϕ is M(F)-invariant,

$$\int_{\overline{G_{\nu}(F)}\setminus \overline{G}(F)} \phi(g) dg = \int_{N(F)} \int_{K_{\nu}} \phi(nk) dn dk.$$
 (4-15)

We normalize the measures dn and dk by taking meas $(N(\mathcal{O}_F)) = \text{meas}(K_v) = 1$.

Proposition 4.4. For F as above, suppose γ is hyperbolic in G(F). Assuming $\gamma \in M(\mathfrak{n}_v)$, and letting $\Delta_{\gamma} \in \mathcal{O}_F$ be its discriminant, we have $\Phi(\gamma, f^{\mathfrak{n}_v}) = |\Delta_{\gamma}|_v^{-1/2}$. In particular, if Δ_{γ} is a unit, then $\Phi(\gamma, f^{\mathfrak{n}_v}) = 1$.

Proof. We may assume that $\gamma = {\alpha \choose \beta}$ for some distinct $\alpha, \beta \in \mathcal{O}_F$. By (4-15) and the fact that $f^{\mathfrak{n}_v}$ is right K_v -invariant,

$$\Phi(\gamma, f^{\mathfrak{n}_v}) = \int_F f^{\mathfrak{n}_v} \left(\begin{pmatrix} 1 & -t \\ & 1 \end{pmatrix} \begin{pmatrix} \alpha & \\ & \beta \end{pmatrix} \begin{pmatrix} 1 & t \\ & 1 \end{pmatrix} \right) dt = \int_F f^{\mathfrak{n}_v} \left(\begin{pmatrix} \alpha & t(\alpha - \beta) \\ & \beta \end{pmatrix} \right) dt.$$

Choose $j \ge 0$ so that $\alpha - \beta \in \varpi^j \mathcal{O}_F^*$. By hypothesis, $\alpha, \beta \in \mathcal{O}_F$ and $\alpha\beta\mathcal{O}_F = \mathfrak{n}_v$, so the integrand is nonzero if and only if $t(\alpha - \beta) \in \mathcal{O}_F$, which is equivalent to $t \in \varpi^{-j} \mathcal{O}_F$. Therefore

$$\Phi(\gamma, f^{\mathfrak{n}_v}) = \operatorname{meas}(\varpi^{-j}\mathcal{O}_F) = q_v^j = |\alpha - \beta|_v^{-1}.$$

Now let $D = \det \gamma$ and $r = \operatorname{tr} \gamma$. Note that

$$4D = 4\alpha\beta = (\alpha + \beta)^2 - (\alpha - \beta)^2 = r^2 - (\alpha - \beta)^2. \tag{4-16}$$

Therefore

$$\Phi(\gamma, f^{\mathfrak{n}_v}) = |\alpha - \beta|_v^{-1} = |r^2 - 4D|_v^{-1/2},$$

as claimed. \Box

4.5. Local orbital integrals at primes $\ell \nmid N$: elliptic case. If γ is elliptic over a field F of characteristic 0, then $E = F[\gamma]$ is a quadratic field extension of F, and

$$G_{\gamma}(F) = E^*$$

(see [23, Proposition 26.1]). The center Z(F) is isomorphic to F^* .

Proposition 4.5. Let F be a local field of characteristic 0, and suppose γ is elliptic in G(F). Then $G_{\gamma}(F)/Z(F)$ is compact.

Proof. If $F = \mathbb{R}$, then $G_{\gamma}(\mathbb{R}) = \mathbb{R}[\gamma]^* \cong \mathbb{C}^*$, and the map $z \mapsto z/|z|$ gives rise to $\mathbb{C}^*/\mathbb{R}^* \cong SO(2)/\{\pm 1\}$, which is compact.

Now suppose that F is nonarchimedean, with valuation v and integer ring \mathcal{O}_F . Let $E = F[\gamma]$, and choose a prime element $\pi \in \mathcal{O}_E$. Then letting $e \in \{1, 2\}$ be the ramification index of E/F,

$$G_{\gamma}(F)/Z(F) \cong E^*/F^* = \bigcup_{j=0}^{e-1} \pi^j \mathcal{O}_E^*/\mathcal{O}_F^*,$$
 (4-17)

which is compact.

Consider a p-adic field F, with all notation as in the previous subsection. For γ elliptic in G(F), the above leads to the following natural choice of G(F)-invariant measure on the quotient space $\overline{G_{\gamma}(F)} \setminus \overline{G}(F)$. We assign the compact group $\overline{G_{\gamma}(F)}$ a total volume of 1. We assign $\overline{G}(F)$ the Haar measure for which $\overline{G}(\mathcal{O}_F)$ has measure 1. Together these choices determine the quotient measure via

$$\int_{\overline{G_{\gamma}(F)}\setminus \overline{G}(F)} \int_{\overline{G_{\gamma}(F)}} \phi(xy) \, dx \, dy = \int_{\overline{G}(F)} \phi(g) \, dg.$$

In fact, by our normalization, if ϕ is left $G_{\gamma}(F)$ -invariant, then

$$\int_{\overline{G_{V}(F)}\setminus \overline{G}(F)} \phi(y) \, dy = \int_{\overline{G}(F)} \phi(g) \, dg, \tag{4-18}$$

when γ is elliptic in G(F).

For such γ , $E = F[\gamma]$ is a quadratic extension of F. Fix an F-integral basis $\{1, \varepsilon\}$ for the ring of integers \mathcal{O}_E , so

$$\mathcal{O}_F = \mathcal{O}_F + \mathcal{O}_F \,\varepsilon. \tag{4-19}$$

We will need some facts about orders and lattices in E. Recall that an F-order in E is a subring containing \mathcal{O}_F which has rank 2 as an \mathcal{O}_F -module.

Proposition 4.6. Let $\mathfrak{O}_{E/F}$ denote the set of all F-orders in E. For $r \geq 0$ and ε as above, define

$$\mathcal{O}_r = \mathcal{O}_F + \mathfrak{p}^r \varepsilon,$$

where \mathfrak{p} is the maximal ideal of \mathcal{O}_F , so in particular $\mathcal{O}_0 = \mathcal{O}_E$. Then

$$\mathfrak{O}_{E/F} = \{ \mathcal{O}_r \mid r \ge 0 \}.$$

Furthermore, letting e = e(E/F) be the ramification index, for r > 0 we have

$$[\mathcal{O}_E^* : \mathcal{O}_r^*] = \begin{cases} q_v^r & \text{if } e = 2, \\ q_v^r + q_v^{r-1} & \text{if } e = 1. \end{cases}$$
 (4-20)

Proof. See also [40, Sections 6.6 and 6.7] for the case $F = \mathbb{Q}_p$. Here we loosely follow Okada [42, Section 2.3]. Clearly $\mathcal{O}_r \in \mathfrak{O}_{E/F}$. Conversely, let $\mathcal{O} \in \mathfrak{O}_{E/F}$. The elements of \mathcal{O} are integral over E [41, Proposition I.2.2] so $\mathcal{O} \subseteq \mathcal{O}_E$. Hence there exists $\alpha \in \mathcal{O} \subseteq \mathcal{O}_E$ such that

$$\mathcal{O} = \mathcal{O}_F + \mathcal{O}_F \alpha.$$

Since $\alpha \notin \mathcal{O}_F$, by topological considerations we see that there exists $r \ge 0$ such that $\alpha \in \mathcal{O}_F + \varpi^r \mathcal{O}_E = \mathcal{O}_r$ but $\alpha \notin \mathcal{O}_F + \varpi^{r+1} \mathcal{O}_E = \mathcal{O}_{r+1}$. Hence

$$\mathcal{O}_{r+1} \subsetneq \mathcal{O} \subseteq \mathcal{O}_r$$
.

We see easily that $\mathcal{O}_r/\mathcal{O}_{r+1} \cong \mathfrak{p}^r/\mathfrak{p}^{r+1} \cong \mathcal{O}_F/\mathfrak{p}$ as \mathcal{O}_F -modules. Since the latter is 1-dimensional as a vector space over $\mathcal{O}_F/\mathfrak{p}$, it has no nonzero proper submodules. It follows that $\mathcal{O} = \mathcal{O}_r$.

For the second part, consider the sequence

$$1 \to \mathcal{O}_F^*/(1+\mathfrak{p}^r) \to \mathcal{O}_E^*/(1+\mathfrak{p}^r\mathcal{O}_E) \to \mathcal{O}_E^*/\mathcal{O}_r^* \to 1,$$

where the maps are the obvious ones. It is straightforward to check that the sequence is exact. Therefore

$$[\mathcal{O}_E^*:\mathcal{O}_r^*] = \frac{|\mathcal{O}_E^*/(1+\mathfrak{p}^r\mathcal{O}_E)|}{|\mathcal{O}_F^*/(1+\mathfrak{p}^r)|}.$$

Let e = e(E/F), so that $\mathfrak{p}\mathcal{O}_E = \mathfrak{P}^e$, where \mathfrak{P} is the maximal ideal of \mathcal{O}_E . Then

$$\begin{split} |\mathcal{O}_E^*/(1+\mathfrak{p}^r\mathcal{O}_E)| &= |\mathcal{O}_E^*/(1+\mathfrak{P}^{er})| \\ &= [\mathcal{O}_E^*:1+\mathfrak{P}] \prod_{j=2}^{er} [1+\mathfrak{P}^{j-1}:1+\mathfrak{P}^j] = (q_E-1) \, q_E^{er-1} \end{split}$$

(see [41, p. 139]). Here,

$$q_E = |\mathcal{O}_E/\mathfrak{P}| = \begin{cases} q_v & \text{if } e = 2, \\ q_v^2 & \text{if } e = 1. \end{cases}$$

Likewise $|\mathcal{O}_F^*/(1+\mathfrak{p}^r)|=(q_v-1)\,q_v^{r-1}$, and (4-20) follows immediately. \square

For the purposes of this subsection, a *lattice* in $F^2 = F \times F$ is an \mathcal{O}_F -submodule of rank 2. The group F^* acts by multiplication on the set of lattices, and the orbits are called *lattice classes*. The map $g \mapsto L = g\binom{\mathcal{O}_F}{\mathcal{O}_F}$ from G(F) to the set of lattices in F^2 induces a bijection between $\overline{G}(F)/\overline{K_v}$ and the set of lattice classes, since K_v is the stabilizer of $\binom{\mathcal{O}_F}{\mathcal{O}_F}$.

With notation as in (4-19), we may identify a lattice $L \subseteq F^2$ with the lattice $(1 \ \varepsilon)L \subseteq E$, so that in particular $\binom{\mathcal{O}_F}{\mathcal{O}_F}$ is identified with \mathcal{O}_E . Given $\eta \in E^*$, it acts by scalar multiplication on the set of lattices in E, and by matrix multiplication (via $E = F[\gamma]$) on the lattices in F^2 . In general, these actions are not compatible with the above identification. However, as shown in [23, Lemma 26.20], after possibly replacing γ (or equivalently, ε) by a G(F)-conjugate, these two actions do coincide for all $\eta \in E^*$. Explicitly, for any $g \in G(F)$,

$$\eta(1\ \varepsilon)g\begin{pmatrix}\mathcal{O}_F\\\mathcal{O}_F\end{pmatrix}=(1\ \varepsilon)\eta g\begin{pmatrix}\mathcal{O}_F\\\mathcal{O}_F\end{pmatrix},$$

where on the left η acts as a scalar via $\eta(1 \varepsilon) = (\eta \eta \varepsilon)$, and on the right it is acting by matrix multiplication. We will assume that γ is chosen in this way, as we may since the value of the orbital integral depends only on γ 's conjugacy class in G(F).

We associate to any lattice $L \subseteq E$ the order

$$\mathcal{O}_L = \{ \mu \in E \mid \mu L \subseteq L \}.$$

This depends only on the lattice class to which L belongs. Since E is local, every lattice in E is principal in the sense that there exists $y \in E^*$ such that $yL = \mathcal{O}_L$. (One may adapt the proof of [23, Proposition 26.13], which follows [33]).

Given an order \mathcal{O} , L is a proper \mathcal{O} -lattice if $\mathcal{O}_L = \mathcal{O}$. Two proper \mathcal{O} -lattices $y\mathcal{O}$ and $z\mathcal{O}$ (for $y, z \in E^*$) are equal if and only if $y/z \in \mathcal{O}^*$. Therefore the set of all proper \mathcal{O} -lattices corresponds bijectively with E^*/\mathcal{O}^* .

Lemma 4.7. Suppose $(\det \gamma) \mathcal{O}_F = \mathfrak{n}_v$ and $g \in G(F)$. Then for $f^{\mathfrak{n}_v}$ given by (4-14), $f^{\mathfrak{n}_v}(g^{-1}\gamma g) \neq 0$ if and only if $\gamma \in \mathcal{O}_L$ for $L = g\binom{\mathcal{O}_F}{\mathcal{O}_F}$.

Proof. We observe that

$$\gamma \in \mathcal{O}_L \iff \gamma L \subseteq L \iff g^{-1} \gamma g \begin{pmatrix} \mathcal{O}_F \\ \mathcal{O}_F \end{pmatrix} \subseteq \begin{pmatrix} \mathcal{O}_F \\ \mathcal{O}_F \end{pmatrix} \iff g^{-1} \gamma g \in M_2(\mathcal{O}_F).$$

Given that $\operatorname{ord}_v(\det \gamma) = \operatorname{ord}_v(\mathfrak{n}_v)$, the above is equivalent to $g^{-1}\gamma g$ belonging to the support $Z(F)M(\mathfrak{n}_v)$ of $f^{\mathfrak{n}_v}$.

Proposition 4.8. Let $f^{\mathfrak{n}_{\upsilon}}$ be given by (4-14). Then for $\gamma \in G(F)$ elliptic, the orbital integral

$$\Phi(\gamma, f^{\mathfrak{n}_{v}}) = \int_{\overline{G_{\gamma}(F)} \setminus \overline{G}(F)} f^{\mathfrak{n}_{v}}(g^{-1}\gamma g) \, dg$$

vanishes unless some conjugate of γ lies in $Z(F)M(\mathfrak{n}_v)$. Taking $\gamma \in M(\mathfrak{n}_v)$, with measure normalized as in (4-18) we have

$$\Phi(\gamma, f^{\mathfrak{n}_v}) = e_{\gamma} \sum_{r=0}^{n_{\gamma}} [\mathcal{O}_E^* : \mathcal{O}_r^*],$$

where $E = F[\gamma]$ is the associated quadratic extension of F with ramification index $e_{\gamma} \in \{1, 2\}$ and ring of integers $\mathcal{O}_E = \mathcal{O}_F + \mathcal{O}_F \varepsilon$,

$$\mathcal{O}_r = \mathcal{O}_F + \mathfrak{p}^r \varepsilon$$

is the order of index q_v^r inside \mathcal{O}_E , and $n_{\gamma} \geq 0$ is defined by $\mathcal{O}_{\gamma} = \mathcal{O}_F + \mathcal{O}_F \gamma = \mathcal{O}_r$ for $r = n_{\gamma}$. In particular, if $\mathcal{O}_{\gamma} = \mathcal{O}_E$ and \mathfrak{p} is inert in E, then $\Phi(\gamma, f^{\mathfrak{n}_v}) = 1$.

Remarks. (1) Let $P_{\gamma}(X) \in \mathcal{O}_F[X]$ be the characteristic polynomial of γ . If P_{γ} is irreducible modulo \mathfrak{p} , then $e_{\gamma} = 1$ and $\mathcal{O}_{\gamma} = \mathcal{O}_E$ [50, p. 18]. Hence $\Phi(\gamma, f^{\mathfrak{n}_v}) = 1$ in this case.

(2) The index $[\mathcal{O}_E^*:\mathcal{O}_r^*]$ is given explicitly in (4-20) when r>0 (and is 1 when r=0).

(3) Let $\mathfrak{d}_{E/F} = \det \left(\frac{1}{1} \frac{\varepsilon}{\varepsilon}\right)^2 \mathcal{O}_F$ be the relative discriminant (with the bar denoting Galois conjugation), write $\gamma = s + b\varepsilon$ for $s, b \in \mathcal{O}_F$, and let $\Delta_{\gamma} = r^2 - 4D$ be the discriminant of γ . Then

$$n_{\gamma} = \operatorname{ord}_{v}(b) = \frac{1}{2}(\operatorname{ord}_{v}(\Delta_{\gamma}) - \operatorname{ord}_{v}(\mathfrak{d}_{E/F})). \tag{4-21}$$

This follows from the fact that the relative discriminant of

$$\mathcal{O}_{\gamma} = \mathcal{O}_F + \mathcal{O}_F \, \gamma = \mathcal{O}_F + \mathcal{O}_F \, b\varepsilon$$

is given on the one hand by

$$\det \begin{pmatrix} 1 & b\varepsilon \\ 1 & b\bar{\varepsilon} \end{pmatrix}^2 \mathcal{O}_F = b^2 \mathfrak{d}_{E/F},$$

and also (using (4-16)) by

$$\det \begin{pmatrix} 1 & \gamma \\ 1 & \overline{\gamma} \end{pmatrix}^2 \mathcal{O}_F = (\gamma - \overline{\gamma})^2 \mathcal{O}_F = \Delta_{\gamma} \mathcal{O}_F.$$

Further, if F is the completion of a number field L at a place v, $\{1, \varepsilon_L\}$ is an integral basis of $L[\gamma]$ over L, and we write $\gamma = s_L + b_L \varepsilon_L$, then (4-21) also holds with b_L in place of b. Indeed the same argument applies in the global case to give $b_L^2 \mathfrak{d}_{L[\gamma]/L} = \Delta_\gamma \mathcal{O}_L$. By the fact By the fact that the global discriminant is the product of the local ones and (due to γ being elliptic in G(F)) there is only one prime of $L[\gamma]$ lying over v, we see that $\operatorname{ord}_v(b_L) = \operatorname{ord}_v(b)$.

(4) If $E = \mathbb{Q}_{\ell}[\sqrt{d}]$ for $d \in \mathbb{Z}$ square-free, then (see [36, Section 6.10], for example)

$$\mathcal{O}_E = \begin{cases} \mathbb{Z}_2 \left[\frac{1}{2} (1 + \sqrt{-3}) \right] & \text{if } \ell = 2, E = \mathbb{Q}_2 [\sqrt{-3}], \\ \mathbb{Z}_\ell [\sqrt{d}] & \text{otherwise.} \end{cases}$$
(4-22)

In particular, if $\ell > 2$ and the valuation $\alpha = v_{\ell}(\Delta_{\gamma})$ of the discriminant of γ is *odd*, then $e_{\gamma} = 2$, $n_{\gamma} = \frac{1}{2}(\alpha - 1)$, and assuming $\gamma \in M(n)_{\ell}$,

$$\Phi(\gamma, f_{\ell}^{n}) = 2 \sum_{r=0}^{(\alpha-1)/2} \ell^{r}.$$
 (4-23)

Proof of Proposition 4.8. The first statement is clear. Now suppose $\gamma \in M(\mathfrak{n}_v)$. By (4-18),

$$\Phi(\gamma, f^{\mathfrak{n}_{v}}) = \int_{\overline{G}(F)} f^{\mathfrak{n}_{v}}(g^{-1}\gamma g) dg.$$

The integrand is right $\overline{K_v}$ -invariant as a function of g. Since $\overline{K_v}$ is open with measure 1, $\overline{G}(F)/\overline{K_v}$ is discrete with the counting measure. Therefore

$$\Phi(\gamma, f^{\mathfrak{n}_{v}}) = \sum_{g \in \overline{G}(F)/\overline{K_{v}}} f^{\mathfrak{n}_{v}}(g^{-1}\gamma g).$$

By our earlier remarks, we can view the sum as a sum over the lattice classes, and by Lemma 4.7, $\Phi(\gamma, f^{n_v})$ is equal to the number of lattice classes preserved by γ .

Since $\gamma \in E$ is integral over \mathcal{O}_F , $\mathcal{O}_{\gamma} = \mathcal{O}_F + \mathcal{O}_F \gamma$ is an order in E (see [23, Lemma 26.10]). We claim that $\gamma \mathcal{O}_r \subseteq \mathcal{O}_r$ if and only if $0 \le r \le n_{\gamma}$, where $q_v^{n_{\gamma}}$ is the index of \mathcal{O}_{γ} . Indeed,

$$\gamma \mathcal{O}_r \subseteq \mathcal{O}_r \iff \gamma \in \mathcal{O}_r \iff \mathcal{O}_\gamma \subseteq \mathcal{O}_r \iff r \leq n_\gamma.$$

It follows that

$$\Phi(\gamma, f^{\mathfrak{n}_v}) = \sum_{r=0}^{n_\gamma} (\text{# of classes of proper } \mathcal{O}_r\text{-lattices}).$$

Recall from earlier that the set of proper \mathcal{O}_r -lattices corresponds bijectively with E^*/\mathcal{O}_r^* . Since we are counting F^* -classes of lattices rather than lattices themselves, we find

$$\Phi(\gamma, f^{\mathfrak{n}_v}) = \sum_{0 \le r \le n_{\gamma}} |E^*/F^*\mathcal{O}_r^*| \quad (\mathcal{O}_{\gamma} = \mathcal{O}_{n_{\gamma}}).$$

Because $\mathcal{O}_F^* \subseteq \mathcal{O}_r^*$, it follows from (4-17) that $|E^*/F^*\mathcal{O}_r^*| = e_{\gamma}[\mathcal{O}_E^* : \mathcal{O}_r^*]$, where $e_{\gamma} \in \{1, 2\}$ is the ramification index of E/F. The result now follows.

Corollary 4.9. For $f^{\mathfrak{n}_v}$ as in (4-14), let $\gamma \in M(\mathfrak{n}_v)$ have characteristic polynomial $P_{\gamma}(X) = X^2 - rX + D \in \mathcal{O}_F[X]$ with discriminant $\Delta_{\gamma} = r^2 - 4D$. Then if γ is hyperbolic in G(F), $\Phi(\gamma, f^{\mathfrak{n}_v}) = |\Delta_{\gamma}|_v^{-1/2}$. If γ is elliptic in G(F) and $P_{\gamma}(X)$ does not have a double root in $\mathcal{O}_F/\mathfrak{p}$, then $\Phi(\gamma, f^{\mathfrak{n}_v}) = 1$.

Consequently, for $\gamma \in M(\mathfrak{n}_v)$ elliptic or hyperbolic in G(F), $\Phi(\gamma, f^{\mathfrak{n}_v}) = 1$ if

$$\Delta_{\gamma} \notin \mathfrak{p}$$
.

Proof. The hyperbolic case is just a restatement of Proposition 4.4. Suppose γ is elliptic. If P_{γ} does not have a double root in $\mathcal{O}_F/\mathfrak{p}$, then it cannot have a simple root either, because otherwise that root would lift to a root in F by Hensel's lemma. By the first remark after Proposition 4.8, $\Phi(\gamma, f^{\mathfrak{n}_v}) = 1$.

Furthermore, suppose $\mathfrak{p} \nmid 2$, and note that $P'_{\gamma}(X) = 2X - r$ vanishes only at $\frac{r}{2} \in \mathcal{O}_F/\mathfrak{p}$. On the other hand,

$$P_{\gamma}\left(\frac{r}{2}\right) = D - \frac{r^2}{4},$$

which shows that P_{γ} has a repeated root modulo \mathfrak{p} if and only if $\mathfrak{p} \mid (r^2 - 4D)$. Hence when $\mathfrak{p} \nmid 2$ and $\Delta_{\gamma} \notin \mathfrak{p}$, $\Phi(\gamma, f^{\mathfrak{n}_{\nu}}) = 1$.

If $\mathfrak{p} \mid 2$ and $(r^2 - 4D) \notin \mathfrak{p}$, then $r \in \mathcal{O}_F^*$, and therefore $P_{\gamma}'(X) = 2X - r$ is nonzero mod \mathfrak{p} . Hence P_{γ} does not have a repeated root, and $\Phi(\gamma, f^{\mathfrak{n}_v}) = 1$ in this case as well.

Although the result of Proposition 4.8 appears complicated, it is not so hard to evaluate it by hand, using the remarks that follow the proposition and standard results about quadratic extensions of *p*-adic fields.

Example 4.10. Let ℓ be a prime not dividing D, and let $\gamma = \begin{pmatrix} 0 & -D \\ 1 & 0 \end{pmatrix}$. Then for f_{ℓ} as in (4-7),

$$\Phi(\gamma, f_{\ell}) = \begin{cases} 2 & \text{if } \ell = 2 \text{ and } D \equiv 1, 5, 7 \text{ mod } 8, \\ 4 & \text{if } \ell = 2 \text{ and } D \equiv 3 \text{ mod } 8, \\ 1 & \text{if } \ell \neq 2. \end{cases}$$

Remark. Some additional examples are given in Section 7.5.

Proof. First suppose $\ell \neq 2$. Since the discriminant -4D of $P_{\gamma}(X) = X^2 + D$ is not divisible by ℓ , $\Phi(\gamma, f_{\ell}) = 1$ by Corollary 4.9.

Now suppose $\ell=2$, so D is odd since $\ell \nmid D$. Recall that the squares of \mathbb{Q}_2^* are exactly the elements of the set $2^{2\mathbb{Z}}(1+8\mathbb{Z}_2)$ [49, Theorem II.4]. Thus -D is a square in \mathbb{Q}_2^* if and only if

$$D \equiv 7 \mod 8$$
.

When this congruence is satisfied, γ is hyperbolic, and by Corollary 4.9,

$$\Phi(\gamma, f_2) = |-4D|_2^{-1/2} = 2.$$

Now suppose that -D is not a square in \mathbb{Q}_2 , i.e., it is not 1 mod 8. We recall some facts about the quadratic extensions of \mathbb{Q}_2 (see, e.g., [36, Chapter 6]). There are exactly seven such extensions, namely $\mathbb{Q}_2[\sqrt{d}]$ for

$$d = -1, \pm 3, \pm 2, \pm 6,$$

with $\mathbb{Q}_2[\sqrt{-3}]$ being the unique unramified quadratic extension. With the exception of d=-3, the ring of integers is $\mathbb{Z}_2[\sqrt[4]{d}]$. For d=-3, the ring of integers is $\mathbb{Z}_2[\frac{1}{2}(1+\sqrt{-3})]$. Under the given hypothesis, $-D\equiv d \mod 8$, where $d\in\{-1,\pm 3\}$. So $-D\equiv dx$ for some $x\in 1+8\mathbb{Z}_2$, and hence $-D\equiv dy^2$ for some $y\in \mathbb{Z}_2^*$. Therefore, writing $E=\mathbb{Q}_2[\sqrt{-D}]$, we have $\mathcal{O}_E=\mathcal{O}_\gamma$ unless d=-3. In the former case, E/\mathbb{Q}_2 is ramified, so by Proposition 4.8,

$$\Phi(\gamma, f_2) = 2 \quad (D \equiv 1, 5 \mod 8).$$

If $D \equiv 3 \mod 8$, then $\mathcal{O}_E = \mathbb{Z}_2 + \mathbb{Z}_2 \varepsilon$ for $\varepsilon = \frac{1}{2}(1 + \sqrt{-3})$. Hence

$$\mathcal{O}_{\gamma} = \mathbb{Z}_2 + \mathbb{Z}_2 \sqrt{-D} = \mathbb{Z}_2 + \mathbb{Z}_2 \sqrt{-3} = \mathbb{Z}_2 + \mathbb{Z}_2 2\varepsilon.$$

So in the notation of Proposition 4.8, $n_{\gamma} = 1$. Since E/\mathbb{Q}_2 is unramified, using (4-20), we have

$$\Phi(\gamma, f_2) = [\mathcal{O}_E^* : \mathcal{O}_E^*] + [\mathcal{O}_E^* : \mathcal{O}_{\gamma}^*] = 1 + 3 = 4.$$

4.6. The set of relevant γ . Here we determine explicitly the finite set of conjugacy classes in $\overline{G}(\mathbb{Q})$ that can have a nonzero contribution to the trace of R(f) for f as in (4-10). Writing $N = \prod_{p|N} p^{N_p}$, define the square-free integers

$$S = \prod_{p \mid N, N_p \text{ even}} p, \quad T = \prod_{p \mid N, N_p \text{ odd}} p.$$

We say that an elliptic element $\gamma \in G(\mathbb{Q}_p)$ is unramified (at p) if $v_p(\det \gamma)$ is even, and ramified otherwise.

Lemma 4.11. Let $\gamma \in G(\mathbb{Q})$ be elliptic, and suppose $\Phi(\gamma, f) \neq 0$ for $f = f^n$ as in (4-10). Then there exists a unique positive divisor $M \mid T$ and a scalar $z \in \mathbb{Q}^*$ such that $\operatorname{tr}(\gamma z) \geq 0$ is an integer and

$$\det(z\gamma) = \mathbf{n}M.$$

In particular, the rational canonical form of $z\gamma$ lies in $M_2(\mathbb{Z})$.

Proof. If $p \mid S$, then γ is unramified at p since f_p is supported in Z_pK_p . For $p \mid T$, the support of f_p has both ramified and unramified elements (see (3-2)). Let M be the product of those primes $p \mid T$ at which γ is ramified. For each prime $\ell \nmid N$, some conjugate of γ must lie in $\operatorname{Supp}(f_\ell^n) = Z_\ell M(n)_\ell$ since otherwise the integrand of $\Phi(\gamma, f)$ vanishes. It follows that $v_p(\det \gamma/nM)$ is even for *all* primes p, where v_p is the p-adic valuation. Hence $\det \gamma \in \pm nM\mathbb{Q}^{*2}$, where \mathbb{Q}^{*2} is the set of squares in \mathbb{Q}^* . Because f_∞ is supported on $G(\mathbb{R})^+$, there is a scalar $z \in \mathbb{Q}^*$ such that $\det(z\gamma) = nM$, as claimed. Because $\Phi(z\gamma, f) = \Phi(\gamma, f) \neq 0$, some $G(\mathbb{A}_{fin})$ -conjugate of $z\gamma$ lies in

$$\prod_{p \mid M} \binom{1}{p} K_p \times \prod_{p \mid (ST/M)} K_p \times \prod_{\ell \nmid N} M(\mathbf{n})_{\ell} \subseteq M_2(\hat{\mathbb{Z}})$$
 (4-24)

(recall that f_p is supported in the group J of (3-2)). In particular, $\operatorname{tr}(z\gamma) \in \hat{\mathbb{Z}} \cap \mathbb{Q} = \mathbb{Z}$. Scaling z by -1 if necessary, we may arrange further that $\operatorname{tr}(z\gamma) \geq 0$.

Lemma 4.12. Let F be a p-adic field, and γ an elliptic element of G(F) with $\operatorname{tr} \gamma \in \mathcal{O}_F$ and $\det \gamma \in \mathfrak{p}$. Then $\operatorname{tr} \gamma \in \mathfrak{p}$.

Proof. Denote the characteristic polynomial of γ by

$$P_{\gamma}(X) = X^2 - dX + \det \gamma,$$

where $d = \operatorname{tr} \gamma$. Notice that $P_{\gamma}(0) \equiv 0 \mod \mathfrak{p}$. Furthermore, $P'_{\gamma}(0) \equiv -d \mod \mathfrak{p}$. If d is nonzero modulo \mathfrak{p} , then by Hensel's lemma, P_{γ} has a root in \mathfrak{p} , contradicting the fact that γ is elliptic in G(F). Hence $d \in \mathfrak{p}$.

Proposition 4.13. For $\gamma \in \overline{G}(\mathbb{Q})$ elliptic, and $f = f^n$ the test function defined in (5-21), $\Phi(\gamma, f) = 0$ unless the conjugacy class of γ has a representative in $G(\mathbb{Q})$ of the form $\binom{0-nM}{rM}$ for some $M \mid T$ and $0 \le r < \sqrt{4n/M}$.

Remark. If the characteristic polynomial of $\gamma = \begin{pmatrix} 0 & -nM \\ 1 & rM \end{pmatrix}$ has a root in \mathbb{Q}_p , then $\Phi(\gamma, f) = 0$ by Proposition 4.3.

Proof. Let \mathfrak{o} be an elliptic conjugacy class in $\overline{G}(\mathbb{Q})$ with $\Phi(\mathfrak{o}, f) \neq 0$. By Lemma 4.11, \mathfrak{o} has a unique representative $\gamma \in G(\mathbb{Q})$ with characteristic polynomial of the form

$$P_{\gamma}(X) = X^2 - dX + nM \in \mathbb{Z}[X],$$

where $d = \operatorname{tr} \gamma \geq 0$ and $M \mid T$. By Proposition 4.3, we know that γ is elliptic in $G(\mathbb{Q}_p)$ for each $p \mid N$ and also in $G(\mathbb{R})$. It follows by Lemma 4.12 that $M \mid d$. Write d = rM. Given that γ is elliptic in $G(\mathbb{R})$, we have $d^2 < 4nM$, i.e.,

$$r^2M < 4n$$
.

So, taking γ in rational canonical form as we may, it has the form

$$\gamma = \begin{pmatrix} 0 & -nM \\ 1 & rM \end{pmatrix}, \quad 0 \le r < \sqrt{4n/M}.$$

4.7. The measure of $\overline{G_{\gamma}(F)} \setminus \overline{G_{\gamma}(\mathbb{A}_F)}$. Let F be a number field with adele ring \mathbb{A}_F , and let γ be an elliptic element of G(F). With G_{γ} the centralizer of γ in G, here we will compute the measure of $\overline{G_{\gamma}(F)} \setminus \overline{G_{\gamma}(\mathbb{A}_F)}$. The result is given in Theorem 4.16 below. A related discussion can be found in [16, Section 5].

The basic idea is straightforward: we know that $G_{\gamma}(\mathbb{A}_F) = \mathbb{A}_F[\gamma]^* = \mathbb{A}_E^*$, where $E = F[\gamma]$ is a quadratic extension of F. (The proof of this fact given in [23, Proposition 26.1] for $F = \mathbb{Q}$ applies to any number field.) The center of $G(\mathbb{A}_F)$ is isomorphic to \mathbb{A}_F^* , so

$$\overline{G_{\gamma}(\mathbb{A}_F)} \cong \mathbb{A}_F^* \backslash \mathbb{A}_E^* \tag{4-25}$$

topologically and algebraically. Finally, $G_{\gamma}(F) \cong F[\gamma]^* = E^*$ by loc. cit., so

$$\overline{G_{\mathcal{V}}(F)} \setminus \overline{G_{\mathcal{V}}(\mathbb{A}_F)} = \mathbb{A}_F^* E^* \setminus \mathbb{A}_F^* \cong (F^* \setminus \mathbb{A}_F^*) \setminus (E^* \setminus \mathbb{A}_F^*) \cong (F^* \setminus \mathbb{A}_F^1) \setminus (E^* \setminus \mathbb{A}_F^1),$$

where the superscript 1 indicates ideles of norm 1, and the latter isomorphism comes from modding out by an embedded copy of \mathbb{R}^+ in $\mathbb{A}_F^*\subseteq \mathbb{A}_E^*$. For any number field L the measure of $L^*\backslash \mathbb{A}_L^1$ is computed in Tate's thesis under suitable normalization, which we may use with L=E, F to obtain the measure of the above space. However, as will be seen, we need to be very careful about the normalization of measures, particularly in the last step.

4.7.1. *Quotient measure.* Recall that if H < G are unimodular locally compact groups with Haar measures μ_H and μ_G and H closed in G, there is a unique left G-invariant quotient measure $\mu_{G/H}$ on G/H satisfying

$$\int_{G/H} \left[\int_{H} f(gh) \, d\mu_{H}(h) \right] d\mu_{G/H}(g) = \int_{G} f(g) \, d\mu_{G}(g) \quad \text{for all } f \in C_{c}(G).$$

Lemma 4.14. Let H, K and T be unimodular locally compact groups, with Haar measures μ_T , μ_H , μ_K , respectively. Assume that H < K, and let $G = T \times K$ and $J = T \times H$. Then relative to the product measures $\mu_G = \mu_T \times \mu_K$ and $\mu_J = \mu_T \times \mu_H$, we have $\mu_{G/J} = \mu_{K/H}$ on the group $G/J \cong K/H$.

Proof. For $f \in C_c(G)$,

$$\int_{K/H} \left[\int_{J} f(xy) d\mu_{J}(y) \right] d\mu_{K/H}(x)
= \int_{K/H} \left[\int_{H} \int_{T} f(xht) d\mu_{T}(t) d\mu_{H}(h) \right] d\mu_{K/H}(x)
= \int_{K} \left[\int_{T} f(kt) d\mu_{T}(t) \right] d\mu_{K}(k) = \int_{G} f(g) d\mu_{G}(g). \qquad \square$$

4.7.2. A volume from Tate's thesis. Let L be a number field with adele ring $\mathbb{A}_L = \prod_v' L_v$, where v ranges over the places of L. In Tate's thesis, measures μ_v on the local multiplicative groups L_v^* are normalized as follows. If v is real,

$$d\mu_v(x) = \frac{dx}{|x|} \tag{4-26}$$

for $x \in \mathbb{R}^*$. If v is complex,

$$d\mu_{v}(z) = 2\frac{dx\,dy}{x^{2} + y^{2}} = \frac{2}{r}dr\,d\theta\tag{4-27}$$

for $z = x + iy = re^{i\theta} \in \mathbb{C}^*$. Finally, at a nonarchimedean place v, μ_v is the Haar measure on L_v^* satisfying

$$\mu_{\nu}(\mathcal{O}_{\nu}^*) = (\mathbb{N}\mathfrak{D}_{\nu})^{-1/2},\tag{4-28}$$

where \mathcal{O}_v is the ring of integers of L_v , \mathfrak{D}_v is the different of L_v and $\mathbb{N}\mathfrak{D}_v = |\mathcal{O}_v/\mathfrak{D}_v|$. Taking the restricted product of the above local measures, we obtain a Haar measure

$$\mu_L = \prod_{v}' \mu_v \quad \text{on } \mathbb{A}_L^*.$$

Let $L_{\infty}^* = \prod_{v \mid \infty} L_v^*$; we embed it into \mathbb{A}_L^* by taking 1's at the nonarchimedean components. We embed \mathbb{R}^+ into L_{∞}^* and hence into \mathbb{A}_L^* via

$$\lambda(t) = (t^{1/n}, t^{1/n}, \dots, t^{1/n}),$$

where $n = n_L = [L : \mathbb{Q}]$. Then if L has r_1 real embeddings and $2r_2$ complex embeddings, for $t \in \mathbb{R}^+$ we have

$$|\lambda(t)|_{\mathbb{A}_L} = \prod_{v \mid \infty} |t|_v^{1/n} = t^{(r_1 + 2r_2)/n} = t$$

(recall that in the ideles we take the square of the usual absolute value at the complex places).

Let $T \cong \mathbb{R}^+$ denote the image of the map λ . We give it the Haar measure dt/t. We have

$$\mathbb{A}_L^* \cong T \times \mathbb{A}_L^1, \tag{4-29}$$

where \mathbb{A}^1_L is the subgroup consisting of ideles of norm 1. There is a unique measure μ^1_L on $\mathbb{A}^1_L\cong\mathbb{A}^*_L/T$ such that

$$\mu_L = \frac{dt}{t} \times \mu_L^1.$$

The multiplicative group L^* embeds diagonally in \mathbb{A}_L^* as a discrete subgroup, and by the product formula, $L^* \subseteq \mathbb{A}_L^1$.

Theorem 4.15 [54, Theorem 4.3.2]. The group L^* is discrete and cocompact in \mathbb{A}^1_L . Giving L^* the counting measure, for μ^1_L as above we have

$$\mu_L^1(L^*\backslash \mathbb{A}_L^1) = \frac{2^{r_1}(2\pi)^{r_2}h(L)R_L}{|d_L|^{1/2}w_L},$$

where h(L), R_L , d_L and w_L are the class number, regulator, discriminant, and number of roots of unity of L, respectively.

Remark. This is the residue of the Dedekind zeta function of L at s=1.

4.7.3. Haar measure for orbital integrals. Let $\gamma \in G(F)$ be an elliptic element. Here we define a Haar measure η on $\overline{G_{\gamma}(\mathbb{A}_F)}$ which is convenient to use for computing the elliptic orbital integrals. Given a nonarchimedean place v of F, γ is necessarily either elliptic or hyperbolic in $G(F_v)$. We select a compact open subgroup H_v of $\overline{G_{\gamma}(F_v)} = Z(F_v) \backslash G_{\gamma}(F_v)$ as follows. If γ is elliptic in $G(F_v)$, then the full group is compact by Proposition 4.5, and we take $H_v = \overline{G_{\gamma}(F_v)}$. If γ is hyperbolic in $G(F_v)$, then $G_{\gamma}(F_v)$ is conjugate to the diagonal subgroup $M(F_v)$. In this case we define H_v to be the subgroup of $\overline{G_{\gamma}(F_v)}$ taken by this conjugation to $\overline{M}(\mathcal{O}_v) \cong \mathcal{O}_v^*$, where \mathcal{O}_v is the ring of integers of F_v .

Next, we choose a local Haar measure η_v on $\overline{G_{\gamma}(F_v)}$ for each place v of F as follows. If $v \nmid \infty$, we normalize η_v so that $\eta_v(H_v) = 1$. If $v \mid \infty$ is a real place of F and γ is elliptic over F_v , we take $\eta_v(\overline{G_{\gamma}(F_v)}) = 1$. If $v \mid \infty$ and γ is hyperbolic over F_v , then $\overline{G_{\gamma}(F_v)} \cong M(F_v)/F_v^* \cong F_v^*$, and we give it the measure $d\eta_v(x) = d\mu_v(x)$ for μ_v as in (4-26) or (4-27).

Note that

$$\overline{G_{\gamma}(\mathbb{A}_F)} = \prod_{v} \overline{G_{\gamma}(F_v)},$$

where the product is restricted relative to the subgroups H_v . We let η denote the Haar measure on $\overline{G_{\nu}(\mathbb{A}_F)}$ which is the restricted product of the above local measures η_v .

²With $F = \mathbb{Q}$, these are the measures that are used in the local orbital integral calculations in the present paper. See Sections 4.4 and 4.5 for finite $\ell \nmid N$ and [23, Section 26.2] for the $\ell = \infty$ calculation yielding (4-12). For $\ell \mid N$, in Section 6 we will use the same measure used in Section 4.5.

As explained in (4-25), for $E = F[\gamma]$ we have

$$\overline{G_{\gamma}(\mathbb{A}_F)} \cong \mathbb{A}_E^*/\mathbb{A}_F^*.$$

So another natural measure on $\overline{G_{\gamma}(\mathbb{A}_F)}$ is the quotient measure $\mu_{E/F}$ coming from the Haar measures μ_E and μ_F on \mathbb{A}_E^* and \mathbb{A}_F^* obtained by taking L=E and L=F respectively in Section 4.7.2.

Let us next determine the constant relating the two measures η and $\mu_{E/F}$. For a place v of F and a place w of E lying over v, we have defined the measures μ_v and μ_w on F_v^* and E_w^* in Section 4.7.2. We let $\mu_v' = \prod_{w \mid v} \mu_w$ be the product measure on $E_v^* = \prod_{w \mid v} E_w^*$, and define $\bar{\mu}_v'$ to be the corresponding quotient measure on $E_v^*/F_v^* \cong \overline{G_\gamma(F_v)}$. Then $\mu_{E/F} = \prod_v' \bar{\mu}_v'$ where v runs over the places of F. For each v we need to find the constant relating η_v to $\bar{\mu}_v'$.

Let v be a nonarchimedean place of F. Suppose γ is hyperbolic in $G(F_v)$, so that $\eta_v(\overline{M}(\mathcal{O}_v)) = 1$. Let w, \overline{w} be the primes of E lying over v. Then

$$E_v := E \otimes F_v \cong E_w \oplus E_{\overline{w}},$$

and $\mu'_v = \mu_w \times \mu_{\overline{w}}$ on

$$G_{\gamma}(F_v) \cong E_v^* \cong E_w^* \times E_{\overline{w}}^* \cong F_v^* \times F_v^*.$$

Hence

$$\mu_v'(\mathcal{O}_w^* \times \mathcal{O}_{\overline{w}}^*) = \mu_w(\mathcal{O}_w^*) \mu_{\overline{w}}(\mathcal{O}_{\overline{w}}^*) = (\mathbb{N}\mathfrak{D}_w)^{-1/2} (\mathbb{N}\mathfrak{D}_{\overline{w}})^{-1/2}$$

by (4-28). (This is in fact equal to \mathbb{ND}_v , but we prefer to leave it unsimplified for global reasons.) Likewise, the diagonally embedded subgroup $F_v^* \subseteq E_v^*$ has measure $\mu_v(\mathcal{O}_v^*) = (\mathbb{ND}_v)^{-1/2}$. Therefore the quotient measure $\overline{\mu}_v'$ on $E_v^*/F_v^* \cong \overline{G}_{\gamma}(F_v)$ gives the open subgroup $(\mathcal{O}_w^* \times \mathcal{O}_{\overline{w}}^*)/\mathcal{O}_v^* \cong \overline{M}(\mathcal{O}_v)$ the measure $\frac{(\mathbb{ND}_v)^{-1/2}(\mathbb{ND}_{\overline{w}})^{-1/2}}{(\mathbb{ND}_v)^{-1/2}}$. Consequently,

$$\eta_v = \frac{(\mathbb{N}\mathfrak{D}_w)^{1/2} (\mathbb{N}\mathfrak{D}_{\overline{w}})^{1/2}}{(\mathbb{N}\mathfrak{D}_v)^{1/2}} \overline{\mu}_v'$$

for such v.

Now suppose γ is elliptic in $G(F_v)$ (again with v nonarchimedean). Then there is a unique valuation w of E extending v, and $E_w = F_v[\gamma]$ is a quadratic extension of F_v . Let \mathcal{O}_w be its ring of integers, with a uniformizer ϖ . Then for the ramification index $e_v = e(w/v) \in \{1, 2\}$,

$$\overline{G_{\gamma}(F_v)} \cong E_w^*/F_v^* = \bigcup_{j=0}^{e_v-1} \varpi^j \mathcal{O}_w^*/\mathcal{O}_v^*$$

as in (4-17). By definition of the local component μ_w of μ_E , $\mu_w(\mathcal{O}_w^*) = (\mathbb{N}\mathfrak{D}_w)^{-1/2}$. The local component of μ_F at v gives $\operatorname{meas}(\mathcal{O}_v^*) = (\mathbb{N}\mathfrak{D}_v)^{-1/2}$. Therefore the

quotient measure $\bar{\mu}'_v$ satisfies

$$\bar{\mu}'_v(\mathcal{O}_w^*/\mathcal{O}_v^*) = \frac{(\mathbb{N}\mathfrak{D}_w)^{-1/2}}{(\mathbb{N}\mathfrak{D}_v)^{-1/2}}.$$

Since $\eta_v(\overline{G_\gamma(F_v)}) = 1$, it follows that

$$\eta_v = \frac{1}{e_v} \frac{(\mathbb{N}\mathfrak{D}_w)^{1/2}}{(\mathbb{N}\mathfrak{D}_v)^{1/2}} \bar{\mu}_v'$$

for such v.

Suppose $F_v = \mathbb{R}$ and γ is elliptic in $G(F_v)$. Then $E_w = \mathbb{C}^*$ and $\overline{G_{\gamma}(F_v)} = \mathbb{C}^*/\mathbb{R}^*$. A set of representatives in \mathbb{C}^* is $\{e^{i\theta} \mid \theta \in [0, \pi)\}$. Since the measure $\mu_v(x) = dx/|x|$ on \mathbb{R}^* matches the factor dr/r in $\mu_w(z) = (2dr d\theta)/r$ given in (4-27), it follows that

$$\bar{\mu}'_{n}(\mathbb{C}^*/\mathbb{R}^*)=2\pi.$$

Since $\eta_v(\overline{G_{\nu}(\mathbb{R})}) = 1$,

$$\eta_v = \frac{1}{2\pi} \bar{\mu}_v'$$

for such v.

If $F_v = \mathbb{R}$ or \mathbb{C} and γ is hyperbolic in $G(F_v)$, then as in the analogous nonarchimedean case,

$$E_v^* = E_w \times E_{\overline{w}} \cong F_v^* \times F_v^*$$

and the quotient measure on $\overline{G_{\gamma}(F_v)} \cong E_v^*/F_v^* \cong F_v^*$ is $\overline{\mu}_v'(x) = \mu_v(x)$. In such cases we have likewise defined $\eta_v = \mu_v$. So $\eta_v = \overline{\mu}_v'$ for such v.

Putting everything together, we have shown that

$$\eta = \bigg[\prod_{v \nmid \infty} \frac{1}{e_v} \frac{\prod_{w \mid v} (\mathbb{N}\mathfrak{D}_w)^{1/2}}{(\mathbb{N}\mathfrak{D}_v)^{1/2}}\bigg] \bigg[\prod_{v \mid \infty, \gamma \text{ elliptic in } G(F_v)} \frac{1}{2\pi}\bigg] \mu_{E/F}.$$

We can simplify using three well-known facts from algebraic number theory (see, e.g., [41, Section III.2]):

- (1) $e_v = 2$ if and only if $\mathfrak{p}_v | \mathfrak{d}_{E/F}$ where $\mathfrak{d}_{E/F}$ is the relative discriminant.
- (2) The absolute discriminant of a local field is the absolute norm of the different.
- (3) The product of the local discriminants is the global discriminant.

It follows that taking d_F , $d_E \in \mathbb{Z}$ to be the discriminants of F and E respectively,

$$\eta = \frac{|d_E|^{1/2}}{|d_F|^{1/2}} \frac{1}{2^{\omega_F(\mathfrak{d}_{E/F})}} \frac{1}{(2\pi)^{\alpha_\gamma}} \mu_{E/F},\tag{4-30}$$

where $\omega_F(\mathfrak{d}_{E/F})$ is the number of distinct prime factors of $\mathfrak{d}_{E/F}$ in \mathcal{O}_F , and α_γ is the number of (real) archimedean places v of F for which γ is elliptic in $G(F_v)$.

4.7.4. The quotient measure. We turn now to the quotient space whose measure we need to compute, namely $\overline{G_{\gamma}(F)} \setminus \overline{G_{\gamma}(\mathbb{A}_F)} \cong E^* \mathbb{A}_F^* \setminus \mathbb{A}_E^* \cong \mathbb{A}_E^* / \mathbb{A}_F^* E^*$. We have defined the quotient measure $\mu_{E/F}$ on $\mathbb{A}_F^* / \mathbb{A}_F^*$. By (4-29), we have

$$\mathbb{A}_F^* = T \times \mathbb{A}_F^1, \quad \mathbb{A}_E^* = T \times \mathbb{A}_E^1.$$

We regard \mathbb{A}_F^* as a subset of \mathbb{A}_E^* , so T is the set

$$T = \{(a, a, \dots, a) \in E_{\infty}^* \mid a > 0\} \subseteq \mathbb{A}_E^*.$$

We will use Lemma 4.14 to relate $\mu_{E/F}$ to the quotient measure on $\mathbb{A}^1_E/\mathbb{A}^1_F$ coming from the measures μ^1_E and μ^1_F defined below (4-29). Recall that T is given the measure $d\mu_T(t) = dt/t$, where $t^{1/n_E} = a$ for $n_E = [E:\mathbb{Q}]$. In terms of the parameter a,

$$d\mu_T(t) = n_E \frac{da}{a}$$
.

Notice that this is *not* the measure given to \mathbb{R}^+ upon taking L = F in (4-29), which is $n_F(da/a) = (n_F/n_E) d\mu_T(t)$. In other words, for μ_T normalized as above, μ_F^1 is defined by

$$\mu_F = \frac{1}{[E:F]} \mu_T \times \mu_F^1.$$

Therefore

$$\mu_F = \mu_T \times \frac{1}{[E:F]} \mu_F^1 = \mu_T \times \frac{1}{2} \mu_F^1.$$

Hence by Lemma 4.14, the quotient measure $\mu_{E/F}$ on $\mathbb{A}_E^*/\mathbb{A}_F^* \cong \mathbb{A}_E^1/\mathbb{A}_F^1$ is the same as the quotient measure coming from μ_E^1 and $\frac{1}{2}\mu_F^1$. We denote this quotient measure by $\mu_{E/F}^1$.

Finally, taking the quotient by the discrete subgroup E^* we have

$$\mu_{E/F}(\mathbb{A}_E^*/E^*\mathbb{A}_F^*) = \mu_{E/F}^1 \left(\frac{(\mathbb{A}_E^1/E^*)}{(\mathbb{A}_F^1 E^*/E^*)} \right) = \frac{\mu_E^1(\mathbb{A}_E^1/E^*)}{\frac{1}{2}\mu_F^1(\mathbb{A}_F^1/F^*)}.$$
 (4-31)

As a technical point, the measure on the disjoint union

$$\mathbb{A}_F^1 E^* = \bigcup_{\alpha \in E^*/F^*} \mathbb{A}_F^1 \alpha$$

is simply $\frac{1}{2}\mu_F^1$ on each component since E^* is given the counting measure. This explains why the quotient measure on $\mathbb{A}_F^1E^*/E^*$ is the same as $\frac{1}{2}\mu_F^1$ on \mathbb{A}_F^1/F^* . Applying Theorem 4.15 and (4-30) to (4-31), we immediately obtain the following.

Theorem 4.16. Let $\gamma \in G(F)$ be an elliptic element, and let η be the measure introduced in Section 4.7.3. Then for $E = F[\gamma]$,

$$\eta(\overline{G_{\gamma}(F)}\setminus\overline{G_{\gamma}(\mathbb{A}_F)}) = \frac{2^{r_1(E)}(2\pi)^{r_2(E)}h(E)R_E}{2^{r_1(F)}(2\pi)^{r_2(F)}h(F)R_F} \cdot \frac{w_F}{w_E} \cdot \frac{2}{2^{\omega_F(\mathfrak{d}_{E/F})}(2\pi)^{\alpha_\gamma}},$$

with notation as in Theorem 4.15, where $\omega_F(\mathfrak{d}_{E/F})$ is the number of distinct prime

ideals of \mathcal{O}_F dividing the relative discriminant $\mathfrak{d}_{E/F}$, and α_{γ} is the number of (real) archimedean places v of F for which γ is elliptic in $G(F_v)$.

In the special case where $F = \mathbb{Q}$ and $E = \mathbb{Q}[\gamma]$ is quadratic imaginary, we have $\alpha_{\gamma} = 1$, $w_F = 2$, $h(F) = R_E = R_F = 1$, so

$$\eta(\overline{G_{\gamma}(\mathbb{Q})}\backslash \overline{G_{\gamma}(\mathbb{A})}) = \frac{2h(E)}{w_E 2^{\omega(d_E)}},\tag{4-32}$$

where $\omega(d_E)$ is the number of distinct prime factors of the discriminant d_E .

With the above in place, the proof of Theorem 4.2 is complete.

5. The case $N = S^2T^3$: proof of Theorem 1.1

Henceforth, we will focus on the case where $N = S^2T^3$ for S and T relatively prime square-free integers. In order to prove Theorem 1.1, by Theorem 4.2 we just need to compute the orbital integrals at the primes dividing N. We begin in Sections 5.1 and 5.2 by reviewing the construction of supercuspidals of conductor p^2 (depth zero case) and of conductor p^3 (simple case), giving explicit formulas for the local test functions to be used. In Section 5.3 we outline the global setup, and then compute the required orbital integrals in Section 6 to complete the proof.

5.1. Depth zero supercuspidal representations. Let F be a p-adic field, with ring of integers \mathcal{O} , maximal ideal $\mathfrak{p} = \varpi \mathcal{O}$, and residue field $\mathbb{k} = \mathcal{O}/\mathfrak{p}$ of size q. The supercuspidal representations of G(F) of minimal conductor are the so-called depth zero supercuspidals, with conductor \mathfrak{p}^2 . They have the form $\sigma = \text{c-Ind}_{ZK}^{G(F)}(\rho)$, where ρ is a (q-1)-dimensional representation of $K = G(\mathcal{O})$ inflated from a cuspidal representation of $G(\mathbb{k})$, and c-Ind denotes compact induction. Some of their properties are summarized below (see, e.g., [29] for more detail).

Temporarily, write $G = G(\mathbb{k})$. Let L be the unique quadratic extension of \mathbb{k} . The multiplicative group L^* embeds as a nonsplit torus $\mathbb{T} \subseteq G$, with \mathbb{k}^* mapping onto the center $Z \subseteq G$. A character $\nu : L^* \to \mathbb{C}^*$ is *primitive* (or regular) if $\nu \neq \nu^q$, or equivalently, if ν is not of the form $\chi \circ N^L_{\mathbb{k}}$ for a character χ of \mathbb{k}^* , where $N^L_{\mathbb{k}}$ is the norm map. There are q(q-1) primitive characters of L^* . Given a character ω of \mathbb{k}^* , let $[\omega]$ denote the set of primitive characters ν satisfying $\nu|_{\mathbb{k}^*} = \omega$. By [29, Proposition 2.3], the cardinality of $[\omega]$ is

$$P_{\omega} = \begin{cases} q - 1 & \text{if } q \text{ is odd and } \omega^{(q-1)/2} \text{ is trivial,} \\ q + 1 & \text{if } q \text{ is odd and } \omega^{(q-1)/2} \text{ is nontrivial,} \\ q & \text{if } q \text{ is even.} \end{cases}$$
 (5-1)

Let $U = \begin{pmatrix} 1 & \mathbb{K} \\ 0 & 1 \end{pmatrix}$ be the upper triangular unipotent subgroup of G. A representation of G is *cuspidal* if it does not contain a U-fixed vector. Fix a nontrivial additive character

$$\psi: \mathbb{k} \to \mathbb{C}^*$$
.

We will always take $\psi(x) = e(x/p) = e^{2\pi i x/p}$ if $\mathbb{k} = \mathbb{Z}/p\mathbb{Z}$. We may view ψ as a character of U in the obvious way.

Given a primitive character ν of \mathbb{T} , there is a unique irreducible cuspidal representation ρ_{ν} of dimension q-1 satisfying

$$\operatorname{Ind}_{ZU}^G(\Bbbk)(\nu \otimes \psi) = \rho_{\nu} \oplus \operatorname{Ind}_{\mathbb{T}}^G \nu.$$

Every cuspidal representation arises in this way, and $\rho_{\nu} \cong \rho_{\nu'}$ if and only if $\nu' \in$ $\{\nu, \nu^q\}.$

We have the following well-known formula for the character of ρ_{ν} . For $x \in G(\mathbb{k})$,

$$\operatorname{tr} \rho_{\nu}(x) = \begin{cases} (q-1)\nu(x) & \text{if } x \in Z, \\ -\nu(z) & \text{if } x = zu, \ z \in Z, \ u \in U, \ u \neq 1, \\ -\nu(x) - \nu^{q}(x) & \text{if } x \in \mathbb{T}, \ x \notin Z, \\ 0 & \text{if no conjugate of } x \text{ belongs to } \mathbb{T} \cup ZU. \end{cases}$$
(5-2)

Because $v(c^{-1}xc) = v(x^q)$ for all $c \in N_G(\mathbb{T}) - \mathbb{T}$, there is no ambiguity evaluating tr $\rho_{\nu}(y)$ using the third row above if y is conjugate in $G(\mathbb{k})$ to $x \in \mathbb{T}$.

Working now in the group G(F), given the surjection $K \to G(\mathbb{k})$ obtained by reduction modulo \mathfrak{p} , we may view ρ_{ν} as a representation of K. Its central character is given by $z \mapsto \nu(z(1+\mathfrak{p}))$ for $z \in \mathcal{O}^*$. By choosing a complex number $\nu(\varpi)$ of norm 1, we may extend ρ_{ν} to a representation of ZK, and then

$$\sigma_{\nu} = \text{c-Ind}_{ZK}^{G(F)}(\rho_{\nu})$$

is an irreducible unitary supercuspidal representation of conductor p². Its formal degree under the normalization meas(K) = 1 is

$$d_{\sigma_{\nu}} = \dim \rho_{\nu} = q - 1. \tag{5-3}$$

The only equivalences among the representations σ_{ν} are $\sigma_{\nu} \cong \sigma_{\nu^q}$ (provided $\nu^q(\varpi)$ is defined to be the same complex number as $v(\varpi)$).

We define the test function $f_{\mathfrak{p}}:G(F)\to\mathbb{C}$ by

$$f_{\mathfrak{p}}(g) = \begin{cases} \overline{\operatorname{tr} \rho_{\nu}(g)} & \text{if } g \in ZK, \\ 0 & \text{otherwise,} \end{cases}$$
 (5-4)

where tr ρ_{ν} is given in (5-2).

Proposition 5.1. Suppose σ_{ν} has trivial central character. Then its root number is given by

$$\epsilon_{\nu} = \epsilon \left(\frac{1}{2}, \sigma_{\nu}, \psi\right) = \begin{cases} -(-1)^{(q+1)/r} & \text{if } q \text{ is odd,} \\ -1 & \text{if } q \text{ is even,} \end{cases}$$
 (5-5)

 $\epsilon_{\nu} = \epsilon(\frac{1}{2}, \sigma_{\nu}, \psi) = \begin{cases} 0.5 \text{ day,} \\ -1 \end{cases}$ if q is even, (5-5) where r is the order of ν in the character group of L^* . Suppose further that q is odd and $4 \nmid (q-1)$ so that $\alpha^2 = -1$ for some $\alpha \in L^* - \mathbb{k}^*$. Then

$$\epsilon_{\nu} = -\nu(\alpha). \tag{5-6}$$

Remark. Under the hypothesis, $\nu|_{\mathbb{R}^*}$ is trivial, which is equivalent to r|(q+1) when q is odd.

Proof. The root number coincides with the Atkin–Lehner sign of the representation [48, 3.2.2 Theorem]. We will show that it is a Gauss sum for ν , which can be evaluated explicitly. The Atkin–Lehner sign ϵ_{ν} is defined by

$$\sigma_{\nu}\left(\begin{pmatrix} 1 \\ \varpi^2 \end{pmatrix}\right)\varphi = \epsilon_{\nu}\,\varphi,$$

where φ is a new vector in the space of σ_{ν} . Note that $\epsilon_{\nu}^2 = 1$ since $\sigma((_{\varpi^2}^{-1})^2) = \sigma((_{\varpi^2}^{-2}))$ acts trivially under the hypothesis of trivial central character.

A model for ρ_{ν} on the space $\mathbb{C}[\mathbb{R}^*]$ of complex-valued functions on \mathbb{R}^* is described in [29], following [47]. In terms of this model, the new space $(\text{c-Ind}_{ZK}^{G(F)}(\rho_{\nu}))^{K_1(\mathfrak{p}^2)}$ is spanned by the function $\varphi: G(F) \to \mathbb{C}[\mathbb{R}^*]$ supported on the coset $ZK\binom{\varpi}{1}K_1(\mathfrak{p}^2)$ and defined by

$$\varphi\left(zk\begin{pmatrix} \overline{\omega} \\ 1 \end{pmatrix}\right) = \rho_{\nu}(zk)\,w \quad (z \in Z, \ k \in K),$$

where $w \in \mathbb{C}[\mathbb{k}^*]$ is the constant function 1 [29, Proposition 3.1]. In particular,

$$\varphi\bigg(\!\!\left(\!\!\begin{array}{c} \varpi \\ 1 \end{array}\!\!\right)\!\!\!\bigg)(1)=w(1)=1.$$

Therefore the Atkin-Lehner eigenvalue is given by

$$\epsilon_{\nu} = \left[\sigma_{\nu}\left(\begin{pmatrix} \sigma \\ \varpi^{2} \end{pmatrix}\right)\varphi\right]\left(\begin{pmatrix} \sigma \\ 1 \end{pmatrix}\right)(1) = \varphi\left(\begin{pmatrix} \sigma \\ 1 \end{pmatrix}\begin{pmatrix} \sigma^{2} \\ \varpi^{2} \end{pmatrix}\right)(1)$$

$$= \varphi\left(\begin{pmatrix} \sigma \\ \varpi \end{pmatrix}\begin{pmatrix} 1 \\ 1 \end{pmatrix}\begin{pmatrix} \sigma \\ 1 \end{pmatrix}\right)(1)$$

$$= \nu(\varpi)\left(\rho_{\nu}\left(\begin{pmatrix} 1 \\ 1 \end{pmatrix}\right)w\right)(1)$$

$$= \left(\rho_{\nu}\left(\begin{pmatrix} 1 \\ -1 \end{pmatrix}\begin{pmatrix} -1 \\ 1 \end{pmatrix}\right)w\right)(1),$$

since we are assuming $\nu|_{F^*}=1$. Let $f_a\in\mathbb{C}[\Bbbk^*]$ be the characteristic function of $a\in \Bbbk^*$, so that $w=\sum_{a\in \Bbbk^*}f_a$. Using [29, (2-11)] we see that $\rho_{\nu}(\binom{-1}{1})w=w$, and just below (2-16) of the same reference, we have

$$\left(\rho_{\nu}\left(\begin{pmatrix}1\\-1\end{pmatrix}\right)f_{a}\right)(1) = -\frac{1}{q}\nu(a^{-1})\sum_{\substack{u \in L^{*}\\N(u)=a}}\psi\left(\operatorname{tr}_{\mathbb{k}}^{L}(u)\right)\nu(u) \quad \text{for all } a \in \mathbb{k}^{*}.$$

We are assuming that $\nu|_{\mathbb{R}^*} = 1$, so $\nu(a^{-1}) = 1$, and summing over $a \in \mathbb{R}^*$ we have

$$\epsilon_{v} = -\frac{1}{q} \sum_{u \in L^{*}} \psi(\operatorname{tr}_{\mathbb{k}}^{L}(u)) v(u).$$

This Gauss sum can be evaluated explicitly by an elementary calculation, giving (5-5); see [2, Theorem 11.6.1] for details.

Now suppose q is odd and $4 \nmid (q-1)$, and let t be a generator of the cyclic group L^* , so in particular $t^{(q^2-1)/2} = -1$. If ν has order r, there exists j with $\gcd(j,r)=1$ such that $\nu(t)=e(j/r)$. Taking $\alpha=t^{(q^2-1)/4}$,

$$\nu(\alpha) = e\left(\frac{j(q+1)(q-1)}{4r}\right) = (-1)^{\frac{j(q+1)}{r}\frac{q-1}{2}} = (-1)^{\frac{j(q+1)}{r}},$$

since $\frac{1}{2}(q-1)$ is odd by hypothesis. The above is equal to $(-1)^{(q+1)/r} = -\epsilon_{\nu}$, since r is odd when j is even, and 2|(q+1). This proves (5-6).

Corollary 5.2. Fix $\epsilon \in \{\pm 1\}$. Then the number of depth zero supercuspidal representations of G(F) with trivial central character and root number ϵ is

$$\begin{cases} \frac{1}{4}(q-1) & \text{if } q \equiv 1 \mod 4, \\ \frac{1}{4}(q+1) & \text{if } q \equiv 3 \mod 4 \text{ and } \epsilon = 1, \\ \frac{1}{4}(q-3) & \text{if } q \equiv 3 \mod 4 \text{ and } \epsilon = -1, \\ 0 & \text{if } q \text{ is even and } \epsilon = 1, \\ \frac{1}{2}q & \text{if } q \text{ is even and } \epsilon = -1. \end{cases}$$

Proof. With notation as in (5-1), the number of supercuspidals with a given central character ω is $\frac{1}{2}P_{\omega}$. (We divide by 2 to account for the fact that ν and ν^q induce the same supercuspidal.) So the assertion for q even is immediate from (5-1) and (5-5).

Let q be odd, and let t be a generator of the cyclic group L^* . Then t^{q+1} is a generator of \mathbb{R}^* . The characters of L^* are the maps ν_m defined by

$$v_m(t) = e\left(\frac{m}{q^2 - 1}\right)$$

for $0 \le m < q^2 - 1$. We consider only those characters satisfying $\nu_m|_{\mathbb{R}^*} = 1$, i.e., $(q-1)|_m$. Notice that ν_m is imprimitive if and only if $\nu_m^{q-1} = 1$, which holds if and only if $(q+1)|_m$. So we consider the values m = k(q-1) (for $1 \le k < (q+1)$) which are not multiples of q+1, i.e., $k \ne \frac{1}{2}(q+1)$.

The order of v_m is

$$\frac{q^2 - 1}{\gcd(m, q^2 - 1)} = \frac{q + 1}{\gcd(k, q + 1)}.$$
 (5-7)

By (5-5), σ_{ν_m} has root number

$$\epsilon_{\nu_m} = -(-1)^{\gcd(k,q+1)} = -(-1)^k,$$
 (5-8)

since q+1 is even. Notice that the removed value $\frac{1}{2}(q+1)$ of k is odd if and only if $q \equiv 1 \mod 4$. So in this case, among the remaining q-1 values of k, half are odd and half are even. If $q \equiv 3 \mod 4$, then $\frac{1}{2}(q-1)+1=\frac{1}{2}(q+1)$ of the remaining values of k are odd, and $\frac{1}{2}(q-1)-1=\frac{1}{2}(q-3)$ are even.

To count supercuspidal representations, we divide the number of relevant k's by 2 since the distinct characters ν_m and ν_m^q induce the same representation.

5.2. Simple supercuspidal representations. With notation as in the previous section, we recall here the construction of the supercuspidal representations of G(F) of conductor \mathfrak{p}^3 . The central character of any such representation is at most tamely ramified. So we begin by fixing a character $\omega_{\mathfrak{p}}$ of the center $Z = Z(F) \cong F^*$ of G(F), trivial on $1 + \mathfrak{p}$.

Define the following compact open subgroup of G(F):

$$K' = \begin{pmatrix} 1 + \mathfrak{p} & \mathcal{O} \\ \mathfrak{p} & 1 + \mathfrak{p} \end{pmatrix}.$$

Fix a nontrivial character

$$\psi: \mathbb{k} \to \mathbb{C}^*,$$

which we also regard as a character of \mathcal{O} trivial on \mathfrak{p} . Given $t \in \mathbb{R}^*$, define a character $\chi = \chi_t : K' \to \mathbb{C}^*$ by

$$\chi\left(\begin{pmatrix} a & b \\ c \overline{\omega} & d \end{pmatrix}\right) = \psi(b + tc). \tag{5-9}$$

The matrix

$$g_t = g_{\chi} = \begin{pmatrix} & t \\ \varpi & \end{pmatrix}$$

normalizes K', and furthermore

$$\chi(g_{\chi}^{-1}kg_{\chi}) = \chi(k) \quad \text{for all } k \in K'.$$
 (5-10)

Given χ as above, let

$$H' = ZK' \cup g_{\chi} ZK'. \tag{5-11}$$

Although it is not reflected in the notation H', this set depends on both t and the fixed choice of ϖ . Given that $g_{\chi}^2 = t\varpi$, we may extend χ to a character χ_{ζ} of H' via

$$\chi_{\zeta}(g_{\chi}^{d}zk) = \zeta^{d}\omega_{\mathfrak{p}}(z)\chi(k) \tag{5-12}$$

for $z \in Z$ and $k \in K'$, where ζ is a fixed complex number satisfying

$$\zeta^2 = \omega_{\mathfrak{p}}(t\varpi). \tag{5-13}$$

Proposition 5.3. The compactly induced representation $\sigma_{\chi}^{\zeta} = \text{c-Ind}_{H'}^{G(F)}(\chi_{\zeta})$ is an irreducible supercuspidal representation of conductor \mathfrak{p}^3 , with root number

$$\epsilon\left(\frac{1}{2},\sigma_{\chi}^{\zeta},\psi\right)=\zeta.$$

Conversely, every irreducible admissible representation of G(F) of conductor \mathfrak{p}^3 with central character trivial on $1 + \mathfrak{p}$ arises in this way.

Proof. See [31]. For a more recent treatment using the above notation (but on GL_n), see [27, Sections 4 and 5 and Proposition 7.2]. The root number is computed in [1, Corollary 3.12].

We will also use the notation

$$\sigma_t^{\zeta} = \sigma_{\chi}^{\zeta}$$

for t, χ as in (5-9), though it should be borne in mind that the representation depends also on the choice of additive character ψ and uniformizer ϖ . When $F = \mathbb{Q}_p$, we will always take $\varpi = p$ and

$$\psi(x) = e(x/p) = e^{2\pi i x/p}$$
 for $x \in \mathbb{Z}/p\mathbb{Z}$.

Henceforth we assume that $\omega_{\mathfrak{p}}$, and hence also σ_t^{ζ} , is unitary. Under the normalization meas $(\overline{G(\mathcal{O})}) = 1$, the formal degree of σ_{χ}^{ζ} is

$$d_{\chi} = \frac{1}{2}(q^2 - 1). \tag{5-14}$$

This is seen, e.g., from (6.4) of [27] and the last line of the proof of Corollary 6.5 of the same paper.

We define the matrix coefficient $f_{\mathfrak{p}}: G(F) \to \mathbb{C}$ by

$$f_{\mathfrak{p}}(g) = d_{\chi} \left(\overline{\sigma_{t}^{\zeta}(g) \frac{\phi}{\|\phi\|}, \frac{\phi}{\|\phi\|}} \right),$$

where $\phi \in \text{c-Ind}_{H'}^{G(F)}(\chi_{\zeta})$ is the function

$$\phi(g) = \begin{cases} \chi_{\zeta}(g) & \text{if } g \in H', \\ 0 & \text{otherwise.} \end{cases}$$
 (5-15)

Note that

$$\|\phi\|^2 = \int_{\overline{G}(F)} |\phi(g)|^2 dg = \text{meas}(\overline{H}').$$
 (5-16)

Likewise,

$$\langle \sigma_t^{\zeta}(g) \phi, \phi \rangle = \int_{\overline{G}(F)} \phi(xg) \, \overline{\phi(x)} \, dx = \int_{\overline{H'}} \phi(xg) \, \overline{\chi_{\zeta}(x)} \, dx$$

$$= \begin{cases} \operatorname{meas}(\overline{H'}) \chi_{\zeta}(g) & \text{if } g \in H', \\ 0 & \text{otherwise.} \end{cases}$$
(5-17)

By (5-14), (5-16), and (5-17), we have

$$f_{\mathfrak{p}}(g) = \begin{cases} \frac{1}{2}(q^2 - 1)\overline{\chi_{\zeta}(g)} & \text{if } g \in H', \\ 0 & \text{otherwise.} \end{cases}$$
 (5-18)

5.3. Global setup. Fix square-free integers S, T > 0 with ST > 1 and gcd(S, T) = 1, and let k > 2. Set $N = S^2T^3$, and let ω' be a Dirichlet character of modulus N satisfying

$$\omega'(-1) = (-1)^k. \tag{5-19}$$

Let ω be the Hecke character attached to ω' in (4-1). We assume in addition that for each $p \mid N$, ω_p is trivial on $1 + p\mathbb{Z}_p$, since this is true of the central character of every supercuspidal representation of conductor $\leq p^3$. Equivalently, the conductor of ω' divides ST.

Proposition 5.4. If $N = 2^2$ or 2^3 and k is odd, there is no such character.

Proof. If *N* is a power of 2, then by (4-3) and (5-19), $(-1)^k = \omega'(-1) = \omega_2(-1) = 1$ since ω_2 is trivial on $\mathbb{Z}_2^* = 1 + 2\mathbb{Z}_2$. So *k* must be even.

Under the stated hypotheses, for each $p \mid S$, ω_p is trivial on $1 + p\mathbb{Z}_p$. We may thus view ω_p as a character of $(\mathbb{Z}_p/p\mathbb{Z}_p)^* = \mathbb{F}_p^*$. For each such p, fix a primitive character ν_p of $\mathbb{F}_{p^2}^*$ such that $\nu_p \mid_{\mathbb{F}_p^*} = \omega_p$. Recall that the number $P_{\omega_p} > 0$ of such primitive characters is given in (5-1). We define $\nu_p(p) = \omega_p(p)$ and extend multiplicatively so that ν_p can also be viewed as a character of \mathbb{Q}_p^* , which allows us to view ρ_{ν_p} as a representation of Z_pK_p with central character ω_p . We let

$$\sigma_p = \sigma_{\nu_p} = \text{c-Ind}_{Z_p K_p}^{G(\mathbb{Q}_p)}(\rho_{\nu_p})$$

be the associated supercuspidal representation of $G(\mathbb{Q}_p)$. The number of isomorphism classes of supercuspidal representations of conductor p^2 and central character ω_p is $\frac{1}{2}P_{\omega_p}$.

For each prime $p \mid T$, fix a simple supercuspidal representation $\sigma_p = \sigma_{t_p}^{\zeta_p}$ of $G(\mathbb{Q}_p)$ with central character ω_p , where $t_p \in (\mathbb{Z}/p\mathbb{Z})^*$ and $\zeta_p^2 = \omega_p(t_p p)$. When the prime p is understood, we sometimes write t, ζ instead of t_p , ζ_p . By (4-2),

$$\zeta_p^2 = \omega_p(t_p p) = \omega_p(t_p) \prod_{\ell \mid N, \ell \neq p} \omega_\ell(p^{-1}). \tag{5-20}$$

In particular, when $N = p^3$ for p prime, $\zeta_p^2 = \omega_p(t_p)$.

Having made the above choices, we let $\hat{\sigma} = (\sigma_p)_{p|N}$ denote this tuple of local representations. Then $S_k(\hat{\sigma}) \subseteq S_k^{\text{new}}(S^2T^3, \omega')$.

Now consider the test function

$$f = f^{\mathbf{n}} = f_{\infty} \prod_{p \mid N} f_p \prod_{\ell \nmid N} f_{\ell}^{\mathbf{n}}$$
(5-21)

as in (4-10) with $N = S^2T^3$, where, for $p \mid S$ (resp. $p \mid T$), f_p is the chosen test function given in (5-4) (resp. in (5-18)).

The above setup is slightly different from that used in (4-10) and Proposition 4.1 since f_p is not a single matrix coefficient when $p \mid S$, but a certain sum of matrix

coefficients, and without the formal degree coefficient. Nevertheless, the conclusions of Proposition 4.1 do hold for the above test function, as the next result shows.

Proposition 5.5. With f defined above, $\operatorname{tr}(T_n | S_k(\hat{\sigma})) = n^{(k/2)-1} \operatorname{tr} R(f)$.

Remark. This is not special to depth zero supercuspidals. By [29, Proposition 1.2], the proof below applies with any unramified (even power conductor) supercuspidals σ_p at $p \mid S$, using $d_{\sigma_p} = \dim \rho$ in place of p-1, where $\sigma_p = \text{c-Ind}_{ZK}^G(\rho)$. (Ramified supercuspidals may be induced from a *character* of an appropriately chosen open compact-mod-center subgroup, so for these, one can use a test function analogous to (5-18).)

Proof. In the proof of Proposition 4.1, we used the fact [23, Corollary 10.26] that for $\sigma = \sigma_p$, the operator $\sigma(d_\sigma\langle \overline{\sigma(g)w,w}\rangle)$ is the orthogonal projection of the space of σ onto $\mathbb{C}w$. For f_p in (5-4), by [29, Proposition 1.1], there is an orthonormal set $\{w_1,\ldots,w_{p-1}\}$ of vectors in the space of σ such that

$$f_p(g) = \sum_{j=1}^{p-1} \langle \overline{\sigma(g) w_j, w_j} \rangle.$$

So $\sigma(d_{\sigma}f_p) = \sigma((p-1)f_p)$ is the orthogonal projection onto $\operatorname{Span}\{w_1,\ldots,w_{p-1}\}$. Therefore using this local test function in the proof of Proposition 4.1 would give us a block sum of p-1 copies of the matrix for $\operatorname{n}^{1-(k/2)}T_{\mathrm{n}}$. To get the correct trace, we would need to divide by p-1, which is achieved by simply taking f_p instead of $(p-1)f_p$.

Noting that $f_p(1) = \dim \rho_v = p - 1 = d_{\sigma_p}$ for $p \mid S$, the identity term in the formula for tr R(f) is

$$\overline{\omega'(\mathbf{n}^{1/2})} \, \frac{1}{12} (k-1) \prod_{p \mid S} (p-1) \prod_{p \mid T} \frac{1}{2} (p^2 - 1), \tag{5-22}$$

as seen between the brackets in Theorem 4.2. We remark that this is not always an integer when n = 1. For example consider the case where S = 1. For $p \ge 3$ prime,

$$v_2(p^2 - 1) = v_2(p - 1) + v_2(p + 1) \ge 3,$$

with equality holding precisely when $p \equiv 3, 5 \mod 8$. (Here, v_2 is the 2-adic valuation.) It follows easily that when n = 1, the identity term $\frac{1}{12}(k-1) \prod_{p|T} \frac{1}{2}(p^2-1)$ fails to be an integer in exactly the following situations:

- T = 2 and $k \not\equiv 1 \mod 8$.
- T = 3 and $k \not\equiv 1 \mod 3$.
- T = 2p for some $p \equiv 3, 5 \mod 8$, and k is even.

So in such instances, when S = n = 1 the elliptic contribution to $|H_k(\hat{\sigma})|$ in Theorem 4.2 must be nonzero for this simple reason.

The list of relevant matrices in the trace formula of Theorem 4.2 can be refined in certain situations.

Proposition 5.6. Let $N = S^2T^3$ as above, let $f = f^n$ be the test function defined in (5-21), let $M \mid T$, and $0 \le r < \sqrt{4n/M}$. Then $\Phi\left(\binom{0-nM}{r}, f\right) = 0$ in each of the following situations:

- r = 0 and k is odd.
- There exists $p \mid N$ such that $X^2 rMX + nM$ has a root in \mathbb{Q}_p .
- There exists $p \mid M$ such that $-pt_p/nM$ is not a square modulo p, where t_p is the parameter of the local representation $\sigma_{t_p}^{\zeta_p}$.
- There exists p|(T/M) such that $X^2 rMX + nM \equiv (X z)^2 \mod p$ has no solution $z \in (\mathbb{Z}/p\mathbb{Z})^*$.

Remark. For the case n = 1, we can refine the list of relevant γ even further (see Proposition 7.9 below).

Proof. The first bullet point follows from (4-13).

Let $\gamma = \binom{-nM}{rM}$, and suppose that $\Phi(\gamma, f) \neq 0$. Then by Proposition 4.3, γ is elliptic in $G(\mathbb{Q}_p)$, which gives the second bullet point.

For the third bullet point, suppose $p \mid M$. Write $\det \gamma = up$ for some $u \in \mathbb{Z}_p^*$. Assuming the local orbital integral $\Phi(\gamma, f_p)$ is nonzero, $f_p(g^{-1}\gamma g) \neq 0$ for some $g \in G(\mathbb{Q}_p)$. Then $g^{-1}\gamma g$ belongs to the ramified component of $\operatorname{Supp}(f_p)$, i.e., writing $t = t_p$,

$$g^{-1}\gamma g = z \begin{pmatrix} t \\ p \end{pmatrix} \begin{pmatrix} a & b \\ pc & d \end{pmatrix} \in Zg_{\chi_p}K'$$

for some $b, c \in \mathbb{Z}_p$, $a, d \in 1 + p\mathbb{Z}_p$, and $z \in \mathbb{Z}_p^*$. Taking determinants, we have

$$up = -tpz^2(ad - pbc),$$

and hence

$$u \equiv -tz^2 \bmod p. \tag{5-23}$$

This shows that -t/u is a quadratic residue modulo p.

Finally, if $p \mid (N/M)$, then det $\gamma \in \mathbb{Z}_p^*$ so if $\Phi(\gamma, f_p) \neq 0$, some conjugate $g^{-1}\gamma g$ lies in the unramified component of Supp (f_p) :

$$g^{-1}\gamma g = z \begin{pmatrix} a & b \\ pc & d \end{pmatrix} \in ZK'$$

for z, a, b, c, d as above. Taking determinants, $\det \gamma \equiv z^2 \mod p$. Taking the trace, $\operatorname{tr} \gamma \equiv 2z \mod p$. Hence $P_{\gamma}(X) \equiv X^2 - 2zX + z^2 \equiv (X - z)^2 \mod p$.

6. Local orbital integrals at primes $p \mid N$ for $N = S^2T^3$

Our goal here is to compute

$$\Phi(\gamma, f_p) = \int_{\overline{G_{\gamma}(\mathbb{Q}_p)} \setminus \overline{G}(\mathbb{Q}_p)} f_p(g^{-1}\gamma g) \, dg$$

taking for f_p the test functions given in (5-4) and (5-18), and for γ the matrices given in Theorem 4.2, and using the quotient measure defined in Section 4.5, so

$$\Phi(\gamma, f_p) = \int_{\overline{G}(\mathbb{Q}_p)} f_p(g^{-1}\gamma g) \, dg.$$

With these calculations, Theorem 1.1 will follow immediately from Theorem 4.2.

We will use the strategy adopted by Palm in [43, Proposition 9.11.3] which avoids the use of lattices or buildings. There are errors in the statement and proof of his proposition, so we cannot simply quote the result. However, the basic method is sound and can be adapted to give the result in the cases of interest to us here.

The following lemma will allow us to rewrite the integral in such a way as to exploit the structure of the support of f_p .

Lemma 6.1 [43, Lemma 6.4.10]. Let G be a unimodular locally compact group, and suppose I_1 , I_2 are two open compact subgroups of G, each given total Haar measure 1. Then for any choice of Haar measure on G we have

$$\int_{G} \phi(g) \, dg = \sum_{x \in I_{1} \backslash G/I_{2}} \operatorname{meas}_{G}(I_{1} x I_{2}) \int_{I_{1}} \int_{I_{2}} \phi(i_{1} x i_{2}) \, di_{2} \, di_{1}$$
 (6-1)

for all $\phi \in C_c(G)$.

Proof. For $\phi \in C_c(G)$, we see that

$$\int_{G} \phi(g) \, dg = \int_{G} \int_{I_{1}} \int_{I_{2}} \phi(i_{1}gi_{2}) \, di_{2} \, di_{1} \, dg$$

by changing the order of integration and using the bi-invariance of dg. The inner double integral defines a compactly supported function F of $g \in G$ which is constant on double cosets I_1gI_2 , and is therefore a finite linear combination of characteristic functions of such double cosets. The identity (6-1) clearly holds for the characteristic function of a double coset. By linearity it holds for F as well, so

$$\int_{G} \phi(g) \, dg = \int_{G} F(g) \, dg$$

$$= \sum_{x \in I_{1} \backslash G/I_{2}} \operatorname{meas}_{G}(I_{1}xI_{2}) \int_{I_{1}} \int_{I_{2}} F(i_{1}xi_{2}) \, di_{2} \, di_{1}$$

$$= \sum_{x \in I_{1} \backslash G/I_{2}} \operatorname{meas}_{G}(I_{1}xI_{2}) F(x)$$

$$= \sum_{x \in I_{1} \backslash G/I_{2}} \operatorname{meas}_{G}(I_{1}xI_{2}) \int_{I_{1}} \int_{I_{2}} \phi(i_{1}xi_{2}) \, di_{2} \, di_{1}. \qquad \Box$$

6.1. Preliminaries when $p \mid T$. Throughout much of this section, we will work over a p-adic field F with notation as in Section 5.2, and write G for G(F), and \overline{G} for G/Z. Having fixed a simple supercuspidal representation σ_t^{ζ} of G with unitary central character ω_p , we take f_p to be the test function given in (5-18).

Applying Lemma 6.1 to (4-18), we have

$$\Phi(\gamma, f_{\mathfrak{p}}) = \int_{\overline{G}} f_{\mathfrak{p}}(g^{-1}\gamma g) dg$$

$$= \sum_{x \in \overline{K}' \setminus \overline{G}/\overline{K}'} \operatorname{meas}_{\overline{G}}(\overline{K}' x \overline{K}') \int_{\overline{K}'} \int_{\overline{K}'} f_{\mathfrak{p}}(h_2^{-1} x^{-1} h_1^{-1} \gamma h_1 x h_2) dh_1 dh_2,$$

where each dh_i is normalized to have total measure 1. Since $f_{\mathfrak{p}}|_{K'}$ is a character, h_2 has no effect, and we obtain

$$\Phi(\gamma, f_{\mathfrak{p}}) = \sum_{x \in \overline{K}' \setminus \overline{G}/\overline{K}'} \operatorname{meas}_{\overline{G}}(\overline{K}' x \overline{K}') \int_{\overline{K}'} f_{\mathfrak{p}}(x^{-1} h^{-1} \gamma h x) dh.$$
 (6-2)

In order to compute the above, we need a few preparations. First, recall the affine Bruhat decomposition

$$G = K'MK' \cup K'MwK' = K'MK' \cup K'Mg_{\chi}K',$$

where $w = \binom{1}{1}^{-1}$ and M is the diagonal subgroup [7, Proposition 17.1]. Accordingly, we may take as a set of representatives $x \in \overline{K}' \setminus \overline{G}/\overline{K}'$ the elements x = m and $x = mg_{\chi}$ for

$$m \in \left\{ \begin{pmatrix} y \\ 1 \end{pmatrix}, \begin{pmatrix} y \\ \varpi^j \end{pmatrix}, \begin{pmatrix} \varpi^n \\ y \end{pmatrix} \middle| j > 0, n > 0, y \in (\mathcal{O}/\mathfrak{p})^* \right\}.$$
 (6-3)

For each such x we need to compute the integral in (6-2), which we denote by

$$K_{\gamma}(x) = \int_{\overline{K'}} f_{\mathfrak{p}}(x^{-1}h^{-1}\gamma hx) dh.$$

By (5-10),

$$f_{\mathfrak{p}}(g_{\chi}^{-1}gg_{\chi}) = f_{\mathfrak{p}}(g)$$
 for all g .

Therefore $K_{\gamma}(xg_{\chi}) = K_{\gamma}(x)$. Furthermore, since g_{χ} normalizes K', the measure of $\overline{K}'x\overline{K}'$ is unchanged if x is replaced by xg_{χ} . It follows that

$$\Phi(\gamma, f_{\mathfrak{p}}) = 2 \sum_{\substack{x \text{ in } (6-3)}} \text{meas}_{\overline{G}}(\overline{K}' x \overline{K}') K_{\gamma}(x). \tag{6-4}$$

Lemma 6.2. Let $x = {\varpi^n \choose y}$ or ${y \choose \varpi^n}$ for $n \ge 0$ and $y \in \mathcal{O}^*$. Then with measure on \overline{G} normalized so that meas(\overline{K}) = 1,

$$\operatorname{meas}_{\overline{G}}(\overline{K}'x\overline{K}') = \frac{q^n}{q^2 - 1}.$$
 (6-5)

Proof. We may assume that y = 1 since, for example,

$$\begin{aligned} \operatorname{meas} \left(\overline{K}' \begin{pmatrix} \overline{\varpi}^n \\ y \end{pmatrix} \overline{K}' \right) &= \operatorname{meas} \left(\overline{K}' \begin{pmatrix} \overline{\varpi}^n \\ 1 \end{pmatrix} \overline{K}' \begin{pmatrix} 1 \\ y \end{pmatrix} \right) \\ &= \operatorname{meas} \left(\overline{K}' \begin{pmatrix} \overline{\varpi}^n \\ 1 \end{pmatrix} \overline{K}' \right). \end{aligned}$$

Likewise, since g_{χ} normalizes K' and $g_{\chi}^{-1} {m \choose 1} g_{\chi} = {n \choose \varpi^n}$, we may assume that $x = {m \choose 1}$.

We claim that for $n \ge 0$,

$$K'\begin{pmatrix} \overline{\omega}^n \\ 1 \end{pmatrix} K' = \bigcup_{b \in \mathcal{O}/\mathfrak{p}^n} \begin{pmatrix} \overline{\omega}^n & b \\ 0 & 1 \end{pmatrix} K', \tag{6-6}$$

a disjoint union. The union is disjoint since

$$\begin{pmatrix} \varpi^n & b_1 \\ 0 & 1 \end{pmatrix}^{-1} \begin{pmatrix} \varpi^n & b_2 \\ 0 & 1 \end{pmatrix} = \begin{pmatrix} 1 & \frac{b_2 - b_1}{\varpi^n} \\ 0 & 1 \end{pmatrix},$$

which is in K' if and only if $b_1 \equiv b_2 \mod \mathfrak{p}^n$. The inclusion \supseteq in (6-6) follows from

$$\begin{pmatrix} \varpi^n & b \\ 0 & 1 \end{pmatrix} = \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} \begin{pmatrix} \varpi^n \\ & 1 \end{pmatrix} \in K' \begin{pmatrix} \varpi^n \\ & 1 \end{pmatrix}.$$

The reverse inclusion follows from

$$\begin{pmatrix} a & b \\ c & d \end{pmatrix} \begin{pmatrix} \varpi^n \\ & 1 \end{pmatrix} = \begin{pmatrix} \varpi^n & bd^{-1} \\ 0 & 1 \end{pmatrix} \begin{pmatrix} a - cbd^{-1} & 0 \\ c\varpi^n & d \end{pmatrix}.$$

By the decomposition (6-6),

$$\operatorname{meas}\!\left(\overline{K}'\begin{pmatrix}\varpi^n\\1\end{pmatrix}\overline{K}'\right) = q^n\operatorname{meas}(\overline{K}') = \frac{q^n}{q^2-1},$$

since $\operatorname{meas}(\overline{K}') = 1/(q^2 - 1)$ when $\operatorname{meas}(\overline{K}) = 1$, as shown in the proof of [27, Corollary 6.5].

If $x = {y \choose 1}$, then $K_{\gamma}(x) = f_{\mathfrak{p}}(\gamma^{y})$ where $\gamma^{y} = {y^{-1} \choose 1}\gamma{y \choose 1}$, since $f_{\mathfrak{p}}$ is a character of K', ${y \choose 1}$ normalizes K', and we give \overline{K}' measure 1. Thus, in view of the above lemma, (6-4) now becomes

$$\Phi(\gamma, f_{\mathfrak{p}}) = \frac{2}{q^{2} - 1} \sum_{y \in (\mathcal{O}/\mathfrak{p})^{*}} f_{\mathfrak{p}}(\gamma^{y})
+ 2 \sum_{n=1}^{\infty} \frac{q^{n}}{q^{2} - 1} \sum_{y \in (\mathcal{O}/\mathfrak{p})^{*}} \left[K_{\gamma} \left(\begin{pmatrix} \varpi^{n} \\ y \end{pmatrix} \right) + K_{\gamma} \left(\begin{pmatrix} y \\ \varpi^{n} \end{pmatrix} \right) \right].$$
(6-7)

To compute $K_{\nu}(x)$, we fix coordinates on \overline{K}' with the following.

Lemma 6.3. Let G, H, K be compact topological groups, with G = HK and $H \cap K = \{1\}$. Let dh and dk be the respective Haar measures on H, K of total measure 1. Then the Haar measure on G of measure 1 is given by

$$\int_{G} f(g) dg = \int_{H} \int_{K} f(hk) dk dh.$$

Proof. This is a special case of [23, Lemma 7.13].

We will use the Iwahori decomposition [7, (7.3.1)] of K'. Letting $M(1+\mathfrak{p}) = \binom{1+\mathfrak{p}}{1+\mathfrak{p}}$, $N(\mathcal{O}) = \binom{1}{0}$, and $N'(\mathfrak{p}) = \binom{1}{\mathfrak{p}}$, the decomposition

$$K' = N(\mathcal{O}) \cdot N'(\mathfrak{p}) \cdot M(1+\mathfrak{p})$$

is a (topological) direct product, and the same is true for any ordering of the three factors. We will take $meas(\overline{K}') = meas(K') = 1$, so that applying the above lemma, this Haar measure on \overline{K}' is given by both of the following:

$$\int_{\overline{K'}} \phi(k) \, dk = \int_{\mathcal{O}} \int_{\mathcal{O}} \int_{M(1+\mathfrak{p})} \phi\left(\begin{pmatrix} 1 & 0 \\ \varpi \, c & 1 \end{pmatrix} \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} m\right) dm \, db \, dc \tag{6-8}$$

$$= \int_{\mathcal{O}} \int_{\mathcal{O}} \int_{M(1+\mathfrak{p})} \phi\left(\begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} \begin{pmatrix} 1 & 0 \\ \varpi c & 1 \end{pmatrix} m\right) dm \, dc \, db, \tag{6-9}$$

where dm, db, dc each have total measure 1.

6.2. The case where $p \mid T$ and γ is ramified. The aim here is to compute $\Phi(\gamma, f_p)$ when $p \mid N$ and γ is ramified at p, i.e., $v_p(\det \gamma)$ is odd. As above we work over a p-adic field F with uniformizer ϖ , and a fixed supercuspidal representation σ_t^{ζ} of G(F) as in Section 5.2. We can assume that $v_p(\det \gamma) = 1$, and further, by Lemma 4.12, that $v_p(\operatorname{tr} \gamma) \geq 1$. So we will consider matrices in the F-rational canonical form

$$\gamma = \begin{pmatrix} 0 & -u\varpi \\ 1 & v\varpi \end{pmatrix} = w \begin{pmatrix} 1 & v\varpi \\ 0 & u\varpi \end{pmatrix} \tag{6-10}$$

for $u \in \mathcal{O}^*$, $v \in \mathcal{O}$, and $w = {1 \choose 1}^{-1}$.

Proposition 6.4. For γ as in (6-10) and $f_{\mathfrak{p}}$ as in (5-18), $\Phi(\gamma, f_{\mathfrak{p}}) = 0$ unless -t/u is a square modulo \mathfrak{p} . If the latter condition holds and $y^2 \equiv -t/u \mod \mathfrak{p}$, then

$$\Phi\left(\begin{pmatrix} 0 & -u\varpi \\ 1 & v\varpi \end{pmatrix}, f_{\mathfrak{p}}\right) = \bar{\zeta}\left(\overline{\psi(yv)}\,\omega_{\mathfrak{p}}(y) + \delta(\mathfrak{p}\nmid 2)\,\overline{\psi(-yv)}\,\omega_{\mathfrak{p}}(-y)\right),$$

where ψ is the nontrivial character of \mathcal{O}/\mathfrak{p} used in (5-9). Thus, in the case of trivial central character (so $\zeta^2 = 1$), we have

$$\Phi(\gamma, f_{\mathfrak{p}}) = \begin{cases} 2\zeta \operatorname{Re}(\psi(yv)) & \text{if } \mathfrak{p} \nmid 2, \\ \overline{\psi(yv)}\zeta & \text{if } \mathfrak{p} \mid 2. \end{cases}$$

When the central character is trivial, $F = \mathbb{Q}_p$, and $v \in \mathbb{Z}$, this gives

$$\Phi(\gamma, f_p) = \begin{cases} 2\zeta \cos\left(\frac{2\pi yv}{p}\right) & \text{if } p \neq 2, \\ (-1)^v \zeta & \text{if } p = 2. \end{cases}$$
 (6-11)

Proof. We need to compute each term of (6-7). First, note that for $y \in (\mathcal{O}/\mathfrak{p})^*$,

$$\gamma^{y} = \begin{pmatrix} 0 & -u\varpi/y \\ y & v\varpi \end{pmatrix} = \begin{pmatrix} t \\ \varpi \end{pmatrix} \begin{pmatrix} y/\varpi & v \\ -u\varpi/ty \end{pmatrix}$$

does not belong to the support of $f_{\mathfrak{p}}$. Hence $f_{\mathfrak{p}}(\gamma^{y}) = K_{\gamma}({y \choose 1}) = 0$.

Next, suppose $x = {y \choose \varpi^j}$ with j > 0 and $y \in \mathcal{O}^*$. Then we use the measure in (6-8):

$$K_{\nu}(x)$$

$$= \!\!\int_{\mathcal{O}} \!\!\int_{\mathcal{O}} \!\!\int_{M(1+\mathfrak{p})} \!\! f_{\mathfrak{p}} \! \left(x^{-1} m^{-1} \begin{pmatrix} 1 & -b \\ 0 & 1 \end{pmatrix} \! \begin{pmatrix} 1 & 0 \\ -\varpi c & 1 \end{pmatrix} \! \gamma \begin{pmatrix} 1 & 0 \\ \varpi c & 1 \end{pmatrix} \! \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} \! mx \right) \! dm \, db \, dc.$$

Note that m commutes with x, and lies in the kernel of $f_{\mathfrak{p}}$. Therefore the integration over $M(1+\mathfrak{p})$ has no effect, and

$$K_{\gamma}(x) = \int_{\mathcal{O}} \int_{\mathcal{O}} f_{\mathfrak{p}} \left(x^{-1} \begin{pmatrix} 1 & -b \\ 0 & 1 \end{pmatrix} \begin{pmatrix} 1 & 0 \\ -\varpi c & 1 \end{pmatrix} \gamma \begin{pmatrix} 1 & 0 \\ \varpi c & 1 \end{pmatrix} \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} x \right) db \, dc.$$

Likewise

$$\begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} x = \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} \begin{pmatrix} y \\ \varpi^j \end{pmatrix} = \begin{pmatrix} y \\ \varpi^j \end{pmatrix} \begin{pmatrix} 1 & b\varpi^j/y \\ 1 \end{pmatrix}.$$

Note that the matrix on the right lies in K' since j > 0, and in fact it is in the kernel of f_p . Therefore the integral over b also has no effect, and

$$K_{\gamma}(x) = \int_{\mathfrak{p}} f_{\mathfrak{p}} \left(x^{-1} \begin{pmatrix} 1 & 0 \\ -r & 1 \end{pmatrix} \gamma \begin{pmatrix} 1 & 0 \\ r & 1 \end{pmatrix} x \right) dr, \tag{6-12}$$

where dr gives \mathfrak{p} the measure 1.

Taking $\gamma = w \begin{pmatrix} 1 & v\varpi \\ 0 & u\varpi \end{pmatrix}$ as in (6-10),

$$\begin{pmatrix} 1 & 0 \\ -r & 1 \end{pmatrix} w \begin{pmatrix} 1 & v\varpi \\ 0 & u\varpi \end{pmatrix} \begin{pmatrix} 1 & 0 \\ r & 1 \end{pmatrix} = w \begin{pmatrix} 1 & r \\ 0 & 1 \end{pmatrix} \begin{pmatrix} 1 + rv\varpi & v\varpi \\ ru\varpi & u\varpi \end{pmatrix}$$

$$= w \begin{pmatrix} 1 + r^2u\varpi + rv\varpi & (v + ru)\varpi \\ ru\varpi & u\varpi \end{pmatrix}.$$

Writing the above as $w\begin{pmatrix} a & b \\ c & d \end{pmatrix} \in w\begin{pmatrix} 1+\mathfrak{p}^2 & \mathfrak{p} \\ \mathfrak{p}^2 & \varpi\mathcal{O}^* \end{pmatrix}$,

$$x^{-1} \begin{pmatrix} 1 & 0 \\ -r & 1 \end{pmatrix} \gamma \begin{pmatrix} 1 & 0 \\ r & 1 \end{pmatrix} x = \begin{pmatrix} y^{-1} \\ \varpi^{-j} \end{pmatrix} w \begin{pmatrix} a & b \\ c & d \end{pmatrix} \begin{pmatrix} y \\ \varpi^{j} \end{pmatrix}$$

$$= w \begin{pmatrix} \varpi^{-j} \\ y^{-1} \end{pmatrix} \begin{pmatrix} ay & b\varpi^{j} \\ cy & d\varpi^{j} \end{pmatrix}$$

$$= w \begin{pmatrix} ay/\varpi^{j} & b \\ c & d\varpi^{j}/y \end{pmatrix}$$

$$= g_{\chi} \begin{pmatrix} \varpi^{-1} \\ -t^{-1} \end{pmatrix} \begin{pmatrix} ay/\varpi^{j} & b \\ c & d\varpi^{j}/y \end{pmatrix}$$

$$= g_{\chi} \begin{pmatrix} ay/\varpi^{j+1} & b/\varpi \\ -c/t & -d\varpi^{j}/ty \end{pmatrix}.$$

Since the determinant is $u\varpi$, this belongs to the support of f_p if and only if the matrix on the right belongs to \mathcal{O}^*K' . But this would require j+1=0, which is impossible since j>0. Hence

$$K_{\gamma}\left(\begin{pmatrix} y & \\ & \varpi^{j} \end{pmatrix}\right) = 0$$

for all j > 0 and all $y \in \mathcal{O}^*$.

Lastly, consider $x = {\varpi^n \choose y}$ for n > 0 and $y \in \mathcal{O}^*$. We proceed in just the same way, only this time using the coordinates given in (6-9). Taking -b in place of b for convenience, and eliminating the integral over $M(1 + \mathfrak{p})$ with the same justification as before,

$$K_{\gamma}(x) = \int_{\mathcal{O}} \int_{\mathcal{O}} f_{\mathfrak{p}} \left(x^{-1} \begin{pmatrix} 1 & 0 \\ -\varpi c & 1 \end{pmatrix} \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} \gamma \begin{pmatrix} 1 & -b \\ 0 & 1 \end{pmatrix} \begin{pmatrix} 1 & 0 \\ \varpi c & 1 \end{pmatrix} x \right) db \, dc.$$

Now

$$\begin{pmatrix} 1 & 0 \\ c\varpi & 1 \end{pmatrix} x = \begin{pmatrix} 1 & 0 \\ c\varpi & 1 \end{pmatrix} \begin{pmatrix} \varpi^n \\ y \end{pmatrix} = \begin{pmatrix} \varpi^n \\ y \end{pmatrix} \begin{pmatrix} 1 \\ c\varpi^{1+n}/y & 1 \end{pmatrix}.$$

The matrix on the right lies in the kernel of f_p . Therefore

$$K_{\gamma}(x) = \int_{\mathcal{O}} f_{\mathfrak{p}}\left(x^{-1} \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} \gamma \begin{pmatrix} 1 & -b \\ 0 & 1 \end{pmatrix} x\right) db. \tag{6-13}$$

Taking $\gamma = w \begin{pmatrix} 1 & v\varpi \\ 0 & u\varpi \end{pmatrix}$, we have

$$\begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} w \begin{pmatrix} 1 & v\varpi \\ u\varpi \end{pmatrix} \begin{pmatrix} 1 & -b \\ 0 & 1 \end{pmatrix} = w \begin{pmatrix} 1 & 0 \\ -b & 1 \end{pmatrix} \begin{pmatrix} 1 & -b + v\varpi \\ 0 & u\varpi \end{pmatrix}$$

$$= w \begin{pmatrix} 1 & -b + v\varpi \\ -b & b^2 - v\varpi b + u\varpi \end{pmatrix}.$$

Thus letting $P_{\nu}(X)$ denote the characteristic polynomial of γ ,

$$x^{-1} \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} \gamma \begin{pmatrix} 1 & -b \\ 0 & 1 \end{pmatrix} x = \begin{pmatrix} \varpi^{-n} \\ y^{-1} \end{pmatrix} w \begin{pmatrix} 1 & -b + v\varpi \\ -b & P_{\gamma}(b) \end{pmatrix} \begin{pmatrix} \varpi^{n} \\ y \end{pmatrix}$$

$$= w \begin{pmatrix} y^{-1} \\ \varpi^{-n} \end{pmatrix} \begin{pmatrix} \varpi^{n} & y(-b + v\varpi) \\ -b\varpi^{n} & yP_{\gamma}(b) \end{pmatrix}$$

$$= w \begin{pmatrix} \varpi^{n}/y & -b + v\varpi \\ -b & yP_{\gamma}(b)/\varpi^{n} \end{pmatrix}$$

$$= g_{\chi} \begin{pmatrix} \varpi^{-1} \\ -t^{-1} \end{pmatrix} \begin{pmatrix} \varpi^{n}/y & -b + v\varpi \\ -b & yP_{\gamma}(b)/\varpi^{n} \end{pmatrix}$$

$$= g_{\chi} \begin{pmatrix} \varpi^{n-1}/y & v - b/\varpi \\ b/t & -yP_{\gamma}(b)/t\varpi^{n} \end{pmatrix}. \tag{6-14}$$

Since the determinant is $u\varpi$, the above belongs to the support of $f_{\mathfrak{p}}$ if and only if the matrix on the right belongs to \mathcal{O}^*K' . This means in particular that n=1 and $b \in \mathfrak{p}$. Make the change of variables $b=c\varpi$, $db=|\varpi|dc=q^{-1}dc$. Then

$$P_{\gamma}(b) = \varpi (u - vc\varpi + c^2\varpi),$$

and

$$K_{\gamma}\left(\begin{pmatrix}\varpi\\ y\end{pmatrix}\right) = q^{-1} \int_{\mathcal{O}} f_{\mathfrak{p}}\left(g_{\chi}y^{-1}\begin{pmatrix}1 & yv - cy\\ cy\varpi/t & -y^2(u - vc\varpi + c^2\varpi)/t\end{pmatrix}\right) dc.$$

From the definition of K', the integrand is nonzero if and only if $y^2 \equiv -t/u \mod \mathfrak{p}$. (We have already seen in Proposition 5.6 that -t/u must be a square mod \mathfrak{p} .) Assuming this to be the case, then from (5-9), (5-12), and (5-18), we have

$$K_{\gamma}(x) = q^{-1} \overline{\omega_{\mathfrak{p}}(y^{-1})\zeta} d_{\chi} \int_{\mathcal{O}} \overline{\psi(yv - cy + cy)} dc$$

$$= q^{-1} \omega_{\mathfrak{p}}(y) \overline{\zeta} d_{\chi} \overline{\psi(yv)} = \frac{q^{2} - 1}{2q} \omega_{\mathfrak{p}}(y) \overline{\zeta} \overline{\psi(yv)}. \tag{6-15}$$

To recap, for $\gamma = \begin{pmatrix} 0 & -u\varpi \\ 1 & v\varpi \end{pmatrix}$, $K_{\gamma}(x) = 0$ unless $x = \begin{pmatrix} \varpi \\ y \end{pmatrix}$ for $y^2 \equiv -t/u \mod \mathfrak{p}$, so (6-7) becomes

$$\Phi(\gamma, f_{\mathfrak{p}}) = \frac{2q}{q^2 - 1} \sum_{y \in (\mathcal{O}/\mathfrak{p})^*} K_{\gamma} \left(\begin{pmatrix} \varpi \\ y \end{pmatrix} \right) = \bar{\zeta} \sum_{\varepsilon \in \{\pm 1 \bmod \mathfrak{p}\}} \overline{\psi(\varepsilon y_0 v)} \, \omega_{\mathfrak{p}}(\varepsilon y_0),$$

where y_0 is any fixed solution to $y_0^2 \equiv -t/u \mod \mathfrak{p}$. Note that when $\mathfrak{p}|2$, we can take $\varepsilon = 1$, and if $F = \mathbb{Q}_2$ we can also take $y_0 = 1$.

6.3. The case where $p \mid T$ and γ is unramified. We adopt the same notation used in the previous subsection. Suppose γ is unramified, i.e., $\operatorname{ord}_{\mathfrak{p}}(\det \gamma)$ is even. Scaling if needed, we may assume that $\det \gamma \in \mathcal{O}^*$. For the nonvanishing of $\Phi(\gamma, f_{\mathfrak{p}})$, it is

necessary that some conjugate of γ belong to the unramified component of the support of $f_{\mathfrak{p}}$, namely $ZK'_{\mathfrak{p}}$. Given that $u = \det \gamma \in \mathcal{O}^*$, this means that tr γ must also be integral. So we may take γ in rational canonical form

$$\gamma = \begin{pmatrix} 0 & -u \\ 1 & v \end{pmatrix} \tag{6-16}$$

for some $u \in \mathcal{O}^*$ and $v \in \mathcal{O}$.

Proposition 6.5. For γ elliptic in G(F) and of the form (6-16), $\Phi(\gamma, f_{\mathfrak{p}}) = 0$ unless the characteristic polynomial P_{γ} has a nonzero double root modulo \mathfrak{p} :

$$P_{\gamma}(X) \equiv (X - z)^2 \mod \mathfrak{p} \tag{6-17}$$

for some $z \in (\mathcal{O}/\mathfrak{p})^*$. Under this condition,

$$\Phi(\gamma, f_{\mathfrak{p}}) = \frac{\overline{\omega_{\mathfrak{p}}(z)}}{q} \sum_{n=1}^{\infty} \sum_{c \bmod \mathfrak{p}} \mathcal{N}_{\gamma}(c, n) \sum_{v \in (\mathcal{O}/\mathfrak{p})^*} \psi\left(\frac{yc}{z}\right) \psi\left(-\frac{t}{yz}\right)^{\delta(n=1)}, \quad (6\text{-}18)$$

where ψ is the nontrivial character of \mathcal{O}/\mathfrak{p} used in (5-9), $t \in (\mathcal{O}/\mathfrak{p})^*$ is the parameter of σ_t^{ζ} , and

$$\mathcal{N}_{\gamma}(c, n) = \#\{b \mod \mathfrak{p}^{n+1} \mid P_{\gamma}(b) \equiv c\varpi^n \mod \mathfrak{p}^{n+1}\}.$$

Remarks. (1) Since P_{γ} is irreducible over F, there exists r such that $P_{\gamma}(X) \equiv 0$ mod \mathfrak{p}^r has no solution, and hence $\mathcal{N}_{\gamma}(c,n) = 0$ for all pairs (c,n) with $n \geq r$. So the series is actually a finite sum.

(2) When n = 1 the sum over y is a Kloosterman sum. When n > 1,

$$\sum_{v} \psi\left(\frac{yc}{z}\right) = \begin{cases} q-1 & \text{if } c \equiv 0 \text{ mod } \mathfrak{p}, \\ -1 & \text{otherwise.} \end{cases}$$

(3) When $F = \mathbb{Q}_p$, the integer $\mathcal{N}_{\gamma}(c, n)$ is given explicitly in [26, Lemma 9.6], and presumably there is a version of that lemma for an arbitrary p-adic field. In particular, $\mathcal{N}_{\gamma}(c, n) = 0$ unless $n \leq \operatorname{ord}_{p}(\Delta_{\gamma}) - 1$, and for such n,

$$N_{\nu}(c,n) < p^{\lfloor (n+1)/2 \rfloor}$$

assuming γ is elliptic in $G(\mathbb{Q}_p)$ and satisfies (6-17). This gives the following bound for the orbital integral: setting $\delta = \operatorname{ord}_p(\Delta_{\gamma})$,

$$\begin{split} |\Phi(\gamma, f_p)| &\leq \sum_{n=1}^{\delta - 1} (p - 1)(p^{1/2})^{n+1} = p(p - 1) \sum_{n=0}^{\delta - 2} (p^{1/2})^n \\ &= p(p - 1) \frac{(p^{1/2})^{\delta - 1} - 1}{p^{1/2} - 1} \\ &= p(p^{1/2} + 1)(p^{-1/2}p^{\delta/2} - 1) \leq 2p|\Delta_{\gamma}|_p^{-1/2}. \end{split}$$

This illustrates the general bound given in [21, (1.8) and Theorem 3.11], according to which

$$|\Phi(\gamma, f_{\mathfrak{p}})| \leq C \cdot (d_{\sigma_{\mathfrak{p}}})^{\eta} |\Delta_{\gamma}|_{\mathfrak{p}}^{-1/2},$$

where C > 0 and $\eta < 1$ depend only on G(F).

Proof of Proposition 6.5. The first statement was proven in Proposition 5.6. Suppose $\Phi(\gamma, f_{\mathfrak{p}}) \neq 0$ for γ as in (6-16). We will compute each term of (6-7). It is not hard to check that $f_{\mathfrak{p}}(\gamma^y) = 0$ and $K_{\gamma}((\gamma^y)) = 0$, since the matrices involved do not intersect the support of $f_{\mathfrak{p}}$. Therefore

$$\Phi(\gamma, f_{\mathfrak{p}}) = 2\sum_{n=1}^{\infty} \frac{q^n}{q^2 - 1} \sum_{y \in (\mathcal{O}/\mathfrak{p})^*} K_{\gamma} \left(\begin{pmatrix} \varpi^n \\ y \end{pmatrix} \right). \tag{6-19}$$

Now fix $n \ge 1$ and $y \in \mathcal{O}^*$ and let $x = {\binom{\varpi^n}{y}}$. As in (6-13), we have

$$K_{\gamma}(x) = \int_{\mathcal{O}} f_{\mathfrak{p}} \left(x^{-1} \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} \gamma \begin{pmatrix} 1 & -b \\ 0 & 1 \end{pmatrix} x \right) db.$$

By a quick calculation (see (6-14) with u, v in place of $u\varpi, v\varpi$),

$$x^{-1} \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} \gamma \begin{pmatrix} 1 & -b \\ 0 & 1 \end{pmatrix} x = \begin{pmatrix} b & -y P_{\gamma}(b)/\varpi^n \\ \varpi^n/y & v-b \end{pmatrix}. \tag{6-20}$$

Since the determinant is $u \in \mathcal{O}^*$, this belongs to the support of $f_{\mathfrak{p}}$ only if it belongs to \mathcal{O}^*K' . In particular, $b \in \mathcal{O}^*$ and $P_{\gamma}(b) \equiv 0 \mod \mathfrak{p}^n$. Therefore $b \in z + \mathfrak{p}$ for z as in (6-17). From (6-17) we see that $v \equiv 2z \mod \mathfrak{p}$ so $v - b \in z + \mathfrak{p}$ as well. Therefore, pulling out a factor of z from the above matrix,

$$K_{\gamma}\left(\begin{pmatrix} \varpi^n \\ y \end{pmatrix}\right) = \overline{\omega_{\mathfrak{p}}(z)} \int_{z+\mathfrak{p}} f_{\mathfrak{p}}\left(\begin{pmatrix} 1 & -y P_{\gamma}(b)/z\varpi^n \\ \varpi^n/yz & 1 \end{pmatrix}\right) db.$$

Writing $P_{\gamma}(b) \equiv c \varpi^n \mod \mathfrak{p}^{n+1}$ for some $c \in \mathcal{O}/\mathfrak{p}$, by (5-18) the integrand becomes

$$f_{\mathfrak{p}}\left(\begin{pmatrix}1&-yc/z\\\varpi^n/yz&1\end{pmatrix}\right) = \frac{1}{2}(q^2-1)\,\overline{\psi\left(-\frac{yc}{z} + \frac{t\varpi^{n-1}}{yz}\right)}.$$

This depends (via c) only on the coset $b + \mathfrak{p}^{n+1}$ (in fact it depends only on $b + \mathfrak{p}^n$ but we will not use this). Each such coset has measure $q^{-(n+1)}$. Therefore if we let

$$\mathcal{N}_{\gamma}(c, n) = \#\{b \mod \mathfrak{p}^{n+1} \mid P_{\gamma}(b) \equiv c\varpi^n \mod \mathfrak{p}^{n+1}\}\$$

for $c \in \mathcal{O}/\mathfrak{p}$, we find that

$$K_{\gamma}(x) = \overline{\omega_{\mathfrak{p}}(z)} \frac{q^2 - 1}{2q^{n+1}} \sum_{c \bmod \mathfrak{p}} \psi\left(\frac{yc}{z}\right) \psi\left(-\frac{t}{yz}\right)^{\delta(n=1)} \mathcal{N}_{\gamma}(c, n).$$

Inserting this into (6-19) gives the result.

Example 6.6. For $M \in \mathbb{Z}_2^*$ and f_2 as in (5-18),

$$\Phi\left(\begin{pmatrix} -M \\ 1 \end{pmatrix}, f_2\right) = \begin{cases} 1 & \text{if } M \equiv 1 \mod 4, \\ -3 & \text{if } M \equiv 3 \mod 8, \\ 0 & \text{if } M \equiv 7 \mod 8. \end{cases}$$

Proof. First, $\gamma = \binom{1}{1}^{-M}$ is hyperbolic in $G(\mathbb{Q}_2)$ if and only if -M is a square in \mathbb{Q}_2 , which holds if and only if $M \equiv 7 \mod 8$. In this case, $\Phi(\gamma, f_2) = 0$ by (3-4).

Assuming $M \not\equiv 7 \mod 8$, we may apply Proposition 6.5. We need to determine

$$\mathcal{N}_{\nu}(0,n) = \text{number of solutions to } x^2 \equiv -M \mod 2^{n+1}$$

and

$$\mathcal{N}_{\gamma}(1, n) = \text{ number of solutions to } x^2 \equiv 2^n - M \mod 2^{n+1}.$$

Given any odd integer D, the number of solutions to $x^2 \equiv D \mod 2^j$ is

$$\begin{cases} 1, & j = 1, \\ 2, & j = 2, D \equiv 1 \mod 4, \\ 0, & j = 2, D \equiv 3 \mod 4, \\ 4, & j > 2, D \equiv 1 \mod 8, \\ 0, & j > 2, D \not\equiv 1 \mod 8 \end{cases}$$

[32, Theorem 87]. Therefore

$$\mathcal{N}_{\gamma}(0,n) = \begin{cases} 2 & \text{if } n = 1 \text{ and } M \equiv 3 \mod 4, \\ 0 & \text{if } n = 1 \text{ and } M \equiv 1 \mod 4, \\ 4 & \text{if } n \ge 2 \text{ and } M \equiv 7 \mod 8, \\ 0 & \text{if } n \ge 2 \text{ and } M \not\equiv 7 \mod 8, \end{cases}$$

$$\mathcal{N}_{\gamma}(1,n) = \begin{cases} 0 & \text{if } n = 1 \text{ and } M \equiv 3 \mod 4, \\ 2 & \text{if } n = 1 \text{ and } M \equiv 1 \mod 4, \\ 4 & \text{if } n = 2 \text{ and } M \equiv 3 \mod 8, \\ 0 & \text{if } n \ge 3 \text{ and } M \equiv 7 \mod 8, \\ 0 & \text{if } n \ge 3 \text{ and } M \equiv 7 \mod 8, \end{cases}$$

By definition, $\psi_2(x) = (-1)^x$ for $x \in \mathbb{Z}$, and ω_2 is trivial on $1 + 2\mathbb{Z}_2 = \mathbb{Z}_2^*$. So by (6-18) and the above,

$$\begin{split} \Phi(\gamma, f_2) &= \frac{1}{2} [\mathcal{N}_{\gamma}(0, 1) \, \psi_2(0) \, \psi_2(1) + \mathcal{N}_{\gamma}(1, 1) \, \psi_2(1)^2] \\ &\quad + \frac{1}{2} [\mathcal{N}_{\gamma}(0, 2) \, \psi_2(0) + \mathcal{N}_{\gamma}(1, 2) \, \psi_2(1)] \\ &= \begin{cases} \frac{1}{2} [0 + 2] + \frac{1}{2} [0 + 0] = 1 & \text{if } M \equiv 1 \bmod 4, \\ \frac{1}{2} [-2 + 0] + \frac{1}{2} [0 - 4] = -3 & \text{if } M \equiv 3 \bmod 8. \end{cases} \quad \Box$$

Example 6.7. For f_3 as in (5-18) and any $m \in \mathbb{Z}_3^*$,

$$\Phi\left(\begin{pmatrix} 0 & -m^2 \\ 1 & m \end{pmatrix}, f_3 \right) = \omega_3(-m)t \cdot 2_{t=1},$$

where $2_{t=1}$ is a factor of 2 which is present only when t=1. Here, $t \in \{\pm 1\} = (\mathbb{Z}/3\mathbb{Z})^*$ is the parameter of the fixed simple supercuspidal representation σ_t^{ζ} of $G(\mathbb{Q}_3)$.

Remark. When N = 3, we have $\omega_3(-1) = \omega'(-1) = (-1)^k$, so

$$\omega_3(-m) = \begin{cases} (-1)^k & \text{if } m \in 1 + 3\mathbb{Z}_3, \\ 1 & \text{if } m \in -1 + 3\mathbb{Z}_3. \end{cases}$$

Proof. We will apply Proposition 6.5. First note that

$$P_{\nu}(X) = X^2 - mX + m^2 \equiv (X + m)^2 \mod 3,$$

so we can take z = -m in (6-17). We need to find

$$\mathcal{N}_{\gamma}(c, n) = \#\{b \mid b^2 - mb + m^2 \equiv c3^n \mod 3^{n+1}\}.$$

If $b \in \mathbb{Z}_3^*$ is a double root of P_{γ} modulo 3, then we may write b = -m + 3d, so

$$P_{\gamma}(b) = (-m+3d)^2 - m(-m+3d) + m^2 = 3m^2 + 9(d^2 - md) \in 3\mathbb{Z}_3^*.$$

Thus, $\operatorname{ord}_3(P_{\gamma}(b)) = 1$, which means that $\mathcal{N}_{\gamma}(c, n) = 0$ for all $n \geq 2$, and also $\mathcal{N}_{\gamma}(0, 1) = 0$. Some elementary calculations show that independently of m, $\mathcal{N}_{\gamma}(-1, 1) = 0$ and $\mathcal{N}_{\gamma}(1, 1) = 3$. In view of (6-18), this means

$$\Phi(\gamma, f_3) = \frac{1}{3} (\overline{\omega_3(-m)}) \, \mathcal{N}_{\gamma}(1, 1) \left(\psi_3 \left(\frac{1}{-m} \right) \psi_3 \left(\frac{-t}{-m} \right) + \psi_3 \left(\frac{1}{m} \right) \psi_3 \left(\frac{-t}{m} \right) \right)$$

$$= \overline{\omega_3(-m)} \left(e \left(\frac{1-t}{-3m} \right) + e \left(\frac{1-t}{3m} \right) \right)$$

$$= \omega_3(-m) \left[e \left(\frac{t-1}{3} \right) + e \left(\frac{1-t}{3} \right) \right].$$

When t = 1 (resp. t = -1), the expression in the brackets equals 2 (resp. -1). \Box

6.4. The case where $p \mid S$. When $p \mid S$, the support of f_p is contained in $Z_p K_p$, so the orbital integral vanishes unless γ is unramified. We again work over a p-adic field F, with the usual notation, and fix a depth zero supercuspidal representation σ_{ν} of G = G(F) for ν a primitive character of $\mathbb{F}_{q^2}^*$.

Proposition 6.8. Let $f_{\mathfrak{p}}$ be the test function defined in (5-4), and let $\gamma = \binom{-u}{v}$ be an elliptic element of G(F), where $u \in \mathcal{O}^*$ and $v \in \mathcal{O}$. If there exists $z \in (\mathcal{O}/\mathfrak{p})^*$ such that

$$P_{\gamma}(X) \equiv (X - z)^2 \mod \mathfrak{p},$$

then

$$\Phi(\gamma, f_{\mathfrak{p}}) = -\overline{\omega_{\mathfrak{p}}(z)} + \frac{\overline{\omega_{\mathfrak{p}}(z)}}{q} \sum_{n=1}^{\infty} \left[(q-1) \mathcal{N}_{\gamma}(0, n) - \sum_{c \in (\mathcal{O}/\mathfrak{p})^*} \mathcal{N}_{\gamma}(c, n) \right], \quad (6-21)$$

where $\mathcal{N}_{\gamma}(c, n) = \#\{b \mod \mathfrak{p}^{n+1} \mid P_{\gamma}(b) \equiv c\varpi^n \mod \mathfrak{p}^{n+1}\}.$

If $P_{\nu}(X)$ is irreducible modulo \mathfrak{p} , then

$$\Phi(\gamma, f_{\mathfrak{p}}) = -\overline{\nu(\gamma)} - \overline{\nu^{q}(\gamma)}, \tag{6-22}$$

where we interpret the above to mean $-\overline{v(x)} - \overline{v^q(x)}$ if $x \in \mathbb{F}_{q^2}^*$ has the same minimum polynomial over \mathbb{F}_q as the reduction of $\gamma \mod \mathfrak{p}$, i.e., γ is conjugate to $x \in \mathbb{T}$.

- **Remarks.** (1) The remaining possibility where $P_{\gamma}(X)$ has two distinct roots mod \mathfrak{p} cannot occur due to Hensel's Lemma, since we are assuming that γ is elliptic in G(F).
- (2) See the remarks after Proposition 6.5 regarding $\mathcal{N}_{\gamma}(c,n)$. In particular, the sum in (6-21) is finite, and when $F = \mathbb{Q}_p$ we find $|\Phi(\gamma, f_p)| \le 1 + 4|\Delta_{\gamma}|_p^{-1/2}$.

Proof. In this proof we write \overline{G} for $\overline{G}(F)$, Z for Z(F), and K for $G(\mathcal{O})$. By Lemma 6.1.

$$\Phi(\gamma, f_{\mathfrak{p}}) = \int_{\overline{G}} f_{\mathfrak{p}}(g^{-1}\gamma g) dg$$

$$= \sum_{x \in \overline{K} \setminus \overline{G}/\overline{K}} \operatorname{meas}_{\overline{G}}(\overline{K}x\overline{K}) \int_{\overline{K}} \int_{\overline{K}} f_{\mathfrak{p}}(h_2^{-1}x^{-1}h_1^{-1}\gamma h_1xh_2) dh_1 dh_2,$$

with dh_1 and dh_2 each having total measure 1. The integrand is nonzero only if $x^{-1}h_1^{-1}\gamma h_1x \in ZK$. Therefore, since $f_{\mathfrak{p}}$ is a trace, h_2 has no effect, so

$$\Phi(\gamma, f_{\mathfrak{p}}) = \sum_{x \in \overline{K} \setminus \overline{G}/\overline{K}} \operatorname{meas}_{\overline{G}}(\overline{K}x\overline{K}) \int_{\overline{K}} f_{\mathfrak{p}}(x^{-1}h\gamma h^{-1}x) dh.$$

(For convenience in what follows, we have set $h = h_1^{-1}$.)

By the Cartan decomposition of G, a set of representatives for $\overline{K} \setminus \overline{G}/\overline{K}$ is given by

$$\left\{ \begin{pmatrix} \overline{\omega}^n \\ 1 \end{pmatrix} \middle| n \ge 0 \right\}.$$

Arguing as in [36, Lemma 4.5.6(2)], for $x = {m \choose 1}$,

$$|K \backslash KxK| = \begin{cases} q^{n-1}(q+1) & \text{if } n > 0, \\ 1 & \text{if } n = 0. \end{cases}$$

Therefore $\operatorname{meas}_{\overline{G}}(\overline{K}x\overline{K}) = q^{n-1}(q+1)$ when n > 0, so

$$\Phi(\gamma, f_{\mathfrak{p}}) = f_{\mathfrak{p}}(\gamma) + \sum_{n=1}^{\infty} q^{n-1} (q+1) K_{\gamma}(n), \tag{6-23}$$

where

$$K_{\gamma}(n) = \int_{K} f_{\mathfrak{p}}\left(\begin{pmatrix}\varpi^{-n} & \\ & 1\end{pmatrix}h\gamma h^{-1}\begin{pmatrix}\varpi^{n} & \\ & 1\end{pmatrix}\right)dh \quad (n>0).$$

(We may integrate over K since both K and \overline{K} have measure 1.) Write $h\gamma h^{-1} = {w \choose y \ z} \in K$. Then

$$\begin{pmatrix} \varpi^{-n} \\ 1 \end{pmatrix} \begin{pmatrix} w & x \\ y & z \end{pmatrix} \begin{pmatrix} \varpi^{n} \\ 1 \end{pmatrix} = \begin{pmatrix} w & x/\varpi^{n} \\ y\varpi^{n} & z \end{pmatrix}.$$
 (6-24)

This belongs to the support of $f_{\mathfrak{p}}$ only if $x \in \mathfrak{p}^n$.

In the integrand above, we can freely multiply h by a diagonal element of K since such an element commutes with x and can be eliminated since $f_{\mathfrak{p}}$ is a trace. In particular, we can assume $\det h = 1$. Write $h = \begin{pmatrix} a & b \\ c & d \end{pmatrix}$, $\det h = 1$. Then

$$h\gamma h^{-1} = \begin{pmatrix} * & -b^2 + abv - a^2u \\ * & * \end{pmatrix}.$$

If $a \in \mathfrak{p}$ then we must have $b \in \mathcal{O}^*$ since $\binom{a \ b}{c \ d} \in K$. But then the upper-right entry above cannot belong to \mathfrak{p}^n , so the integrand vanishes by (6-24). Therefore we may assume $a \in \mathcal{O}^*$, i.e., $h \in A$, where

$$A = \begin{pmatrix} \mathcal{O}^* & * \\ * & * \end{pmatrix} \cap K.$$

Let's find the measure of A. Let $K(\mathfrak{p}) = 1 + M_2(\mathfrak{p}) \subseteq A$. This is the kernel of the reduction mod \mathfrak{p} map $K \to G(\mathcal{O}/\mathfrak{p})$. Since $|G(\mathcal{O}/\mathfrak{p})| = (q^2 - 1)(q^2 - q)$, we see that meas $(K(\mathfrak{p})) = 1/((q^2 - 1)(q^2 - q))$. Let $\overline{A} = A \mod K(\mathfrak{p})$. Thinking of \overline{A} as a set of matrices in $G(\mathcal{O}/\mathfrak{p})$, we see that

$$|\bar{A}| = (q-1) q(q^2 - q).$$

(There are (q-1) q possible top rows, and then $q^2 - q$ remaining choices for the bottom row.) Hence

$$\operatorname{meas}(A) = \frac{(q-1) \, q \, (q^2 - q)}{(q^2 - 1)(q^2 - q)} = \frac{q}{q+1}.$$

It is not hard to show that

$$A = \begin{pmatrix} \mathcal{O}^* & \\ & \mathcal{O}^* \end{pmatrix} \begin{pmatrix} 1 & \\ \mathcal{O} & 1 \end{pmatrix} \begin{pmatrix} 1 & \mathcal{O} \\ & 1 \end{pmatrix},$$

and that the corresponding decomposition of any element of A is unique. Therefore by Lemma 6.3 we can use the above as a coordinate system for integration over A. Since, as noted above, the diagonal component has no effect on the value of the integral, we have

$$K_{\gamma}(n) = \frac{q}{q+1} \int_{\mathcal{O}} \int_{\mathcal{O}} f_{\mathfrak{p}} \left(x^{-1} \begin{pmatrix} 1 \\ c \end{pmatrix} \right) \begin{pmatrix} 1 & b \\ 1 \end{pmatrix} \gamma \begin{pmatrix} 1 & -b \\ 1 \end{pmatrix} \begin{pmatrix} 1 \\ -c \end{pmatrix} x \right) db dc,$$

where $x = {w^n \choose 1}$, and db and dc each have total measure 1. The integral over c can be eliminated, since ${1 \choose -c}{0 \choose 1} = {w^n \choose 1} {1 \choose -\varpi^n c}{1 \choose 1}$, and the rightmost matrix belongs to K. Therefore

$$\begin{split} K_{\gamma}(n) &= \frac{q}{q+1} \int_{\mathcal{O}} f_{\mathfrak{p}} \left(\begin{pmatrix} \varpi^{-n} \\ 1 \end{pmatrix} \begin{pmatrix} 1 & b \\ 1 \end{pmatrix} \begin{pmatrix} -u \\ 1 & v \end{pmatrix} \begin{pmatrix} 1 & -b \\ 1 \end{pmatrix} \begin{pmatrix} \varpi^{n} \\ 1 \end{pmatrix} \right) db \\ &= \frac{q}{q+1} \int_{\mathcal{O}^{*}} f_{\mathfrak{p}} \left(\begin{pmatrix} b & -P_{\gamma}(b)/\varpi^{n} \\ 0 & v-b \end{pmatrix} \right) db \end{split}$$

as in (6-20). (As a reminder, db is additive measure.) We have replaced the lower-left entry by 0, using the fact that by definition (see (5-4)), $f_{\mathfrak{p}}$ is sensitive only to the reduction of its argument mod \mathfrak{p} . Further, the integrand is nonzero only if $P_{\gamma}(b) \in \mathfrak{p}^n$. Under this condition, given that the characteristic polynomial of the matrix in the integrand is $P_{\gamma}(X)$, and this cannot have distinct roots mod \mathfrak{p} as γ is elliptic in $G(\mathbb{Q}_p)$, there exists $z \in (\mathcal{O}/\mathfrak{p})^*$ such that $b \equiv v - b \equiv z \mod \mathfrak{p}$. In particular, the matrix (viewed modulo \mathfrak{p}) belongs to ZU, with notation as in (5-2). Write $P_{\gamma}(b) \equiv c\varpi^n \mod \mathfrak{p}^{n+1}$, for $c \in \mathcal{O}/\mathfrak{p}$. The integrand becomes

$$\overline{\omega_{\mathfrak{p}}(z)} f_{\mathfrak{p}} \begin{pmatrix} 1 & -c/z \\ 0 & 1 \end{pmatrix} \end{pmatrix} = \begin{cases} \overline{\omega_{\mathfrak{p}}(z)} (q-1) & \text{if } c = 0, \\ -\overline{\omega_{\mathfrak{p}}(z)} & \text{if } c \in (\mathcal{O}/\mathfrak{p})^*. \end{cases}$$

This depends (via c) only on the coset $b + \mathfrak{p}^{n+1}$, which has measure $q^{-(n+1)}$. Therefore

$$K_{\gamma}(n) = \overline{\omega_{\mathfrak{p}}(z)} \cdot \frac{q}{q+1} \cdot \frac{1}{q^{n+1}} \bigg[(q-1) \, \mathcal{N}_{\gamma}(0,n) - \sum_{c \in (\mathcal{O}/\mathfrak{p})^*} \!\!\! \mathcal{N}_{\gamma}(c,n) \bigg].$$

Plugging the above into (6-23), equation (6-21) follows. (Note that $f_{\mathfrak{p}}(\gamma) = -\overline{\omega_{\mathfrak{p}}(z)}$ in this case, since $\gamma - z = {-z - u \choose 1} \not\equiv 0 \mod \mathfrak{p}$, so γ is conjugate mod \mathfrak{p} to zu for some $1 \neq u \in U$).

By the above discussion $K_{\gamma}(n) = 0$ for all n > 0 if $P_{\gamma}(X)$ is irreducible mod \mathfrak{p} . So in this case (6-23) gives $\Phi(\gamma, f_{\mathfrak{p}}) = f_{\mathfrak{p}}(\gamma) = -\overline{\nu(\gamma)} - \overline{\nu^{q}(\gamma)}$ by (5-2) and (5-4). \square

7. General dimension formula, and examples with $N = S^2T^3$

When n = 1, the list of relevant γ in Theorem 4.2 can be simplified. The result is the following general dimension formula.

Theorem 7.1. Let $N = \prod_{p \mid N} p^{N_p} > 1$ with $N_p \geq 2$ for all $p \mid N$. Fix k > 2 and a tuple $\hat{\sigma} = (\sigma_p)_{p \mid N}$ of supercuspidal representations so that $S_k(\hat{\sigma}) \subseteq S_k^{\text{new}}(N, \omega')$ for a Dirichlet character ω' , as detailed at the beginning of Section 4. Let T be the product of all primes $p \mid N$ with N_p odd. Let $f = f^1$ be the test function defined in (4-10) with n = 1 but with f_p chosen as in (7-10) below for all $p \mid T$. Then

$$\begin{split} \dim S_k(\hat{\sigma}) &= \frac{1}{12}(k-1) \prod_{p \mid N} d_{\sigma_p} + \frac{1}{2} \Phi\left(\begin{pmatrix} & -T \\ 1 & \end{pmatrix}, f\right) \\ &+ \frac{1}{2} \delta_{T \in 2\mathbb{Z}^+} \Phi\left(\begin{pmatrix} & -T/2 \\ 1 & \end{pmatrix}, f\right) + \delta_{T=2} \Phi\left(\begin{pmatrix} 0 & -2 \\ 1 & 2 \end{pmatrix}, f\right) \\ &+ \delta_{T=3} \Phi\left(\begin{pmatrix} 0 & -3 \\ 1 & 3 \end{pmatrix}, f\right) + \delta_{T \in \{1,3\}} \Phi\left(\begin{pmatrix} 0 & -1 \\ 1 & 1 \end{pmatrix}, f\right). \end{split}$$

Here, d_{σ_p} is the formal degree of σ_p relative to the Haar measure fixed in Section 2, and the orbital integrals $\Phi(\gamma, f)$ are given as in Theorem 4.2.

Proof. The case where T=1 is already contained in Theorem 4.2 by taking n=1. The simplifications when T>1 are proven in Proposition 7.9 below.

As with Theorem 1.1, using the results of Section 6 we can compute the above explicitly in any case of interest when $N = S^2T^3$ with S and T square-free relatively prime positive integers. Although there is not a particularly nice formula for all such levels, as an illustration we will work everything out in the two special cases where $N = S^2$ and $N = T^3$. These results are stated in Sections 7.1 and 7.4 respectively. In Section 7.5 we give some examples to illustrate Theorem 1.1 with n > 1.

First, we highlight the following consequence of Theorem 7.1.

Corollary 7.2. *In the setting of Theorem 7.1 above, suppose that the weight k is odd, so* $\omega'(-1) = -1$. *For T as in Theorem 7.1, if T > 3 the elliptic terms vanish, so*

$$\dim S_k(\hat{\sigma}) = \frac{1}{12}(k-1) \prod_{p \mid N} d_{\sigma_p} \quad (k > 2 \text{ odd}, T > 3).$$

Remark. If $N = 2^2$ or $N = 2^3$, then $S_k(\hat{\sigma})$ is undefined when k is odd since by Proposition 5.4 there is no appropriate nebentypus.

Proof. If
$$\gamma = \binom{T}{1}$$
 or $\binom{T}{2}$, then $\Phi(\gamma, f_{\infty}) = 0$ when k is odd, by (4-13). \square

7.1. Dimension formula and root number bias when $N = S^2$. When we set T = 1 and take $N = S^2$, the formula in Theorem 7.1 gives the following.

Theorem 7.3. Let $N = S^2$ for S square-free, k > 2, ω' a Dirichlet character of modulus N and conductor dividing S, and let $\hat{\sigma} = (\sigma_p)_{p|S}$ be a tuple of depth zero supercuspidal representations chosen compatibly with ω' as in Section 5.3, with T = 1. Then the subspace $S_k(\hat{\sigma}) \subseteq S_k^{\text{new}}(S^2, \omega')$ has dimension

$$\dim S_k(\hat{\sigma}) = \frac{1}{12}(k-1) \prod_{p \mid N} (p-1) + A_1 + A_2,$$

where

$$A_{1} = \frac{1}{4}(-1)^{S+1+(k/2)} \, \delta_{k \in 2\mathbb{Z}} \prod_{\substack{\text{odd } p \mid N}} (-\overline{\nu_{p}(\alpha)} - \overline{\nu_{p}^{p}(\alpha)}) \, \delta_{p \equiv 3 \mod 4}, \tag{7-1}$$

where v_p is the primitive character of $\mathbb{F}_{p^2}^*$ defining σ_p and $\alpha \in \mathbb{F}_{p^2}^*$ satisfies $\alpha^2 = -1$, and

$$A_{2} = \frac{1}{3} (\delta_{k \equiv 0, 2 \mod 3}) (-1)^{\delta_{k \equiv 2, 3 \mod 6}} (-\omega_{3}(-1))^{\delta(3|N)} \prod_{\substack{p \mid N, p \neq 3}} (-\overline{\nu_{p}(\beta)} - \overline{\nu_{p}^{p}(\beta)}) \delta_{p \equiv 2 \mod 3}, \quad (7-2)$$

where $\beta \in \mathbb{F}_{p^2}^*$ satisfies $\beta^2 - \beta + 1 = 0$.

Remarks. (1) Note that $A_1 = A_2 = 0$ in each of the following situations: (i) $k \equiv 1 \mod 6$, (ii) there exist primes $p, q \mid N$ (which could be equal) such that $p \equiv 1 \mod 4$ and $q \equiv 1 \mod 3$, (iii) k is odd and $p \equiv 1 \mod 3$ for some $p \mid N$, (iv) $k \equiv 1 \mod 3$ and $p \equiv 1 \mod 4$ for some $p \mid N$.

- (2) By summing the above formula over all tuples $\hat{\sigma}$, one obtains a formula for the dimension of the space $S_k^{\min}(S^2, \omega')$ of twist-minimal newforms. See Proposition 7.7.
- (3) Theorem 1.3 from the introduction follows from the above by taking ω' trivial. We will prove this after first proving the above result.

Proof. Taking T = 1 in Theorem 7.1, we have

$$\dim S_k(\hat{\sigma}) = \frac{1}{12}(k-1)\prod_{p|N}(p-1) + \frac{1}{2}\Phi\left(\begin{pmatrix} & -1\\ 1 & \end{pmatrix}, f\right) + \Phi\left(\begin{pmatrix} & -1\\ 1 & 1\end{pmatrix}, f\right).$$

Consider $\gamma = \binom{1}{1}$. Its discriminant is $\Delta_{\gamma} = -4$, and we adopt the shorthand

$$\Phi(\gamma) = m\Phi_{\infty}\Phi_2 \prod_{\text{odd } p \mid N} \Phi_p$$

for (1-4), where $m=2h(E)/(w(E)2^{\omega(d_E)})$ for $E=\mathbb{Q}[\gamma]$. We find that $m=\frac{1}{4}$ and $\Phi_{\infty}=(-1)^{k/2}\delta_{k\in 2\mathbb{Z}}$. If S is odd, then $\Phi_2=2$ by Example 4.10. If S is even, Φ_2 is given by (6-21). Here, $\mathcal{N}_{\gamma}(c,n)=0$ for all $n\geq 2$, $\mathcal{N}_{\gamma}(0,1)=0$ and $\mathcal{N}_{\gamma}(1,1)=2$. So $\Phi_2=-1+\frac{1}{2}(-2)=-2$. Thus in both cases, $\Phi_2=2(-1)^{S+1}$. Finally, for odd $p\mid S$, γ is elliptic in $G(\mathbb{Q}_p)$ if and only if -1 is not a square in \mathbb{Q}_p , i.e., $p\equiv 3 \mod 4$.

In such cases, $P_{\gamma}(X)$ is irreducible modulo p, so by (6-22), $\Phi_p = -\overline{\nu_p(\gamma)} - \overline{\nu_p^p(\gamma)}$. Multiplying everything together, we see that $\frac{1}{2}\Phi(\binom{1}{1},f)$ gives (7-1).

Now consider $\gamma = \begin{pmatrix} 1 & -1 \\ 1 & 1 \end{pmatrix}$. Then $\Delta_{\gamma} = -3$, so

$$\Phi(\gamma) = m \Phi_{\infty} \Phi_3 \prod_{p \mid N, p \neq 3} \Phi_p.$$

We find that $m = \frac{1}{6}$, and

$$\Phi_{\infty} = -\frac{\sin((k-1)\pi/3)}{\sin(\pi/3)} = \begin{cases} 0 & \text{if } k \equiv 1 \mod 3, \\ 1 & \text{if } k \equiv 0, 5 \mod 6, \\ -1 & \text{if } k \equiv 2, 3 \mod 6. \end{cases}$$

If $3 \nmid N$, then by Proposition 4.8, $\Phi_3 = 2$ since 3 is ramified in $\mathbb{Q}_3[\gamma] = \mathbb{Q}_3[\sqrt{-3}]$ and $\mathcal{O}_{\gamma} = \mathbb{Z}_3\left[\frac{1}{2}(1+\sqrt{-3})\right] = \mathbb{Z}_3[\sqrt{-3}]$ is the full ring of integers. If $3 \mid N$, then Φ_3 is given by (6-21) with z = -1. We find that $\mathcal{N}_{\gamma}(c, n) = 0$ for all $n \geq 2$, $\mathcal{N}_{\gamma}(1, 1) = 3$, and $\mathcal{N}_{\gamma}(0, 1) = \mathcal{N}_{\gamma}(2, 1) = 0$. So

$$\Phi_3 = -\overline{\omega_3(-1)} + \frac{1}{3}\overline{\omega_3(-1)}(-3) = -2\omega_3(-1).$$

For $p \mid N$ with $p \neq 3$, $P_{\gamma}(X) = X^2 - X + 1$ is irreducible in \mathbb{Q}_p if and only if -3 is not a square in \mathbb{Q}_p , or equivalently, $p \equiv 2 \mod 3$ (see [23, Lemma 27.4]). For such p, Φ_p is given by (6-22). Multiplying these factors together gives (7-2), and the theorem follows.

Now suppose ω' is trivial, so k > 2 is even. In this case we can simplify the expressions for A_1 and A_2 to obtain Theorem 1.3, as follows.

Proof of Theorem 1.3. Recall that by (5-6), when $p \equiv 3 \mod 4$ and ω_p is trivial, $-\nu_p(\alpha) = -\nu_p^p(\alpha) = \epsilon_p$ is the root number of σ_p . Likewise, by (5-5), $(-1)^{S+1} = \epsilon_2$ when S is even (and 1 otherwise). So in this case, we simply have

$$A_1 = \frac{1}{4} (\epsilon(k, \hat{\sigma})) D_4(S) \prod_{\text{odd } p \mid S} 2, \tag{7-3}$$

where $\epsilon(k, \hat{\sigma}) = (-1)^{k/2} \prod_{p \mid S} \epsilon_p$ is the common global root number of the newforms in $S_k(\hat{\sigma})$, and $D_4(S) \in \{0, 1\}$ vanishes exactly when S is divisible by a prime $p \equiv 1 \mod 4$.

Turning to (7-2), if $p \equiv 2 \mod 3$, the polynomial $X^2 - X + 1$ is irreducible over \mathbb{F}_p . So $L = \mathbb{F}_{p^2}$ has a root $\beta \in L^* - \mathbb{F}_p^*$. Let t be a generator of the cyclic group L^* . The dual group of L^* is the set $\{\nu_m \mid 1 \leq m \leq p^2 - 1\}$, where $\nu_m = \nu_{p,m}$ is defined by

$$\nu_m(t) = \nu_{p,m}(t) = e\left(\frac{m}{p^2 - 1}\right).$$

Suppose p is odd. As shown in the proof of Corollary 5.2, the list of depth zero supercuspidal representations of $G(\mathbb{Q}_p)$ with trivial central character is

$$\{\sigma_{\nu_{p-1}},\sigma_{\nu_{2(p-1)}},\ldots\sigma_{\nu_{\frac{p-1}{2}(p-1)}}\}.$$

So there exists m = k(p-1) such that the primitive character ν_p of $\mathbb{F}_{p^2}^*$ fixed in Theorem 1.3 is given by

$$v_p = v_{p,m} = v_m$$
.

Hopefully this conflict of notation $(\nu_p = \nu_m)$ will cause no confusion, since m cannot equal p.

Noting that $\beta^3 = -1$, we can take $\beta = t^{(p^2-1)/6}$. Then for m = k(p-1),

$$\nu_m(\beta) = e\left(\frac{k(p-1)(p^2-1)}{6(p^2-1)}\right) = e\left(\frac{k(p-1)}{6}\right) = \begin{cases} 1 & \text{if } 3 \mid k, \\ -\frac{1}{2} \pm i \frac{\sqrt{3}}{2} & \text{otherwise.} \end{cases}$$

Therefore

$$B(\nu_p) := -\overline{\nu_p(\beta)} - \overline{\nu_p(\beta^p)} = -2\operatorname{Re}(\nu_m(\beta)) = \begin{cases} -2 & \text{if } 3 \mid k, \\ 1 & \text{if } 3 \nmid k. \end{cases}$$
(7-4)

When p = 2, there is only one supercuspidal, corresponding to m = k = 1, we can take $t = \beta$, and (7-4) holds as well. By (5-7), $3 \mid k$ if and only if the order of ν_m divides $\frac{1}{3}(p+1)$. So the above coincides with $B(\nu_p)$ defined in Theorem 1.3, and the theorem follows from (7-3) and (7-4).

Next we will use Theorem 1.3 to count the locally supercuspidal newforms of level S^2 with a given global root number. (What we will actually compute is the bias in global root number, but the count for each sign could be determined easily by following the proof of Proposition 7.6.)

To understand the impact of the local root numbers on the product of $B(\nu_p)$ in (1-11), the primes of interest are equivalent to 2 mod 3, so aside from p = 2, we have $p \equiv 5 \mod 6$. It is helpful to look at two typical examples:

$$p = 11 \quad AL \quad + \quad - \quad + \quad - \quad + \quad B(v) \quad 1 \quad 1 \quad -2 \quad 1 \quad 1$$

$$p = 17 \quad AL \quad + \quad - \quad + \quad - \quad + \quad D(v) \quad 1 \quad 1 \quad -2 \quad 1 \quad 1 \quad (7-5)$$

$$p = 17 \quad AL \quad + \quad - \quad D(v) \quad 1 \quad 1 \quad -2 \quad 1 \quad 1 \quad -2 \quad 1 \quad 1 \quad -2 \quad 1 \quad 1$$

The Atkin–Lehner sign (AL) in the second row comes from (5-8), and the third row is from (7-4).

Lemma 7.4. Given S > 1 square-free, let H_S^+ (resp. H_S^-) denote the set of tuples $\hat{\sigma} = (\sigma_p)_{p|S}$ satisfying:

- For each $p \mid S$, σ_p has trivial central character and conductor p^2 .
- $\prod_{p|S} \epsilon_p = 1$ (resp. -1), where ϵ_p is the root number of σ_p .

For v_p the primitive character of $\mathbb{F}_{p^2}^*$ attached to σ_p , and $B(v_p)$ defined in (7-4), define

$$\mathcal{B}(S)^{\pm} = \sum_{\hat{\sigma} \in H_S^{\pm}} \prod_{p \mid S, p \neq 3} B(\nu_p).$$

Suppose $D_3(S) = 1$ (in the notation of Theorem 1.3), and let $\omega(S)$ denote the number of prime factors of S. Then if gcd(S, 6) = 1,

$$\mathcal{B}(S)^{+} = \begin{cases} 2^{\omega(S)-1} & \text{if there exists } p \mid S \text{ with } p \equiv 5 \mod 12, \\ 2^{\omega(S)} & \text{if } \omega(S) \text{ is even and } p \equiv 11 \mod 12 \text{ for all } p \mid S, \\ 0 & \text{if } \omega(S) \text{ is odd and } p \equiv 11 \mod 12 \text{ for all } p \mid S, \end{cases}$$
 (7-6)

and $\mathcal{B}(S)^-$ is the same but with "even" and "odd" interchanged, i.e., $\mathcal{B}(S)^- = 2^{\omega(S)} - \mathcal{B}(S)^+$.

If S is odd and 3 | S, then $\mathcal{B}(S)^{\pm} = \mathcal{B}(\frac{S}{3})^{\pm}$ if S > 3, and $\mathcal{B}(3)^{+} = 1$, $\mathcal{B}(3)^{-} = 0$. If S is even, then $\mathcal{B}(S)^{\pm} = \mathcal{B}(\frac{S}{2})^{\mp}$ if S > 2, and $\mathcal{B}(2)^{+} = 0$, $\mathcal{B}(2)^{-} = 1$.

Proof. Suppose $\gcd(S,6)=1$. We prove (7-6) by induction on $\omega(S)$. For the base case, we take S=p for a prime $p\equiv 5 \mod 6$. As in (7-5), there are $\frac{1}{3}(p+1)$ representations with $B(\nu_p)=1$, of which $\frac{1}{6}(p+1)$ have $\epsilon_p=1$ and $\frac{1}{6}(p+1)$ have $\epsilon_p=-1$. There are $\frac{1}{6}(p-5)$ representations with $B(\nu_p)=-2$, of which $\left\lceil \frac{1}{12}(p-5) \right\rceil$ have $\epsilon_p=1$, and $\left\lfloor \frac{1}{12}(p-5) \right\rfloor$ have $\epsilon_p=-1$. Therefore

$$\mathcal{B}(p)^{+} = \sum_{\sigma_{p} \in H_{p}^{+}} B(v_{p}) = \frac{1}{6}(p+1) - 2\lceil \frac{1}{12}(p-5) \rceil = \begin{cases} 1 & \text{if } p \equiv 5 \text{ mod } 12, \\ 0 & \text{if } p \equiv 11 \text{ mod } 12. \end{cases}$$

Likewise

$$\mathcal{B}(p)^{-} = \sum_{\sigma_{p} \in H_{p}^{-}} B(v_{p}) = \frac{1}{6}(p+1) - 2\lfloor \frac{1}{12}(p-5) \rfloor = \begin{cases} 1 & \text{if } p \equiv 5 \mod 12, \\ 2 & \text{if } p \equiv 11 \mod 12. \end{cases}$$

This proves the base case. Suppose (7-6) holds for some S > 1 with gcd(S, 6) = 1, and $\ell \equiv 5 \mod 6$ is a prime not dividing S. Then the result follows, by considering cases, from the fact that

$$\mathcal{B}(S\ell)^{+} = \mathcal{B}(S)^{+} \cdot \mathcal{B}(\ell)^{+} + \mathcal{B}(S)^{-} \cdot \mathcal{B}(\ell)^{-}$$

and

$$\mathcal{B}(S\ell)^{-} = \mathcal{B}(S)^{+} \cdot \mathcal{B}(\ell)^{-} + \mathcal{B}(S)^{-} \cdot \mathcal{B}(\ell)^{+}.$$

When $3 \mid S$, the claim follows from the fact that there is a unique depth zero supercuspidal representation of $PGL_2(\mathbb{Q}_3)$, and it has root number +1 (see Corollary 5.2). When $2 \mid S$, the claim follows from the fact that there is a unique depth zero supercuspidal representation σ_{ν} of $PGL(\mathbb{Q}_2)$, and it has $B(\nu) = 1$ and root number -1. \square

Lemma 7.5. Let H_S^+ and H_S^- be defined as in Lemma 7.4 above. As in (1-11), define $D_4(S) \in \{0, 1\}$ to be 0 if and only if S is divisible by a prime $p \equiv 1 \mod 4$. Then

$$|H_S^{\pm}| = \begin{cases} \frac{1}{2} \prod_{odd \ p \mid S} \frac{1}{2} (p-1) & \text{if } D_4(S) = 0, \\ \frac{1}{2} \prod_{odd \ p \mid S} \frac{1}{2} (p-1) \pm \frac{1}{2} (-1)^{\delta(2|S)} & \text{if } D_4(S) = 1. \end{cases}$$

Proof. For each odd prime p, there are $\frac{1}{2}(p-1)$ depth zero supercuspidals with trivial central character (see Section 5.1). For p=2, there is only one. Therefore for all square-free S>1, the total number of tuples $\hat{\sigma}=(\sigma_p)_{p\mid S}$ with each σ_p having depth zero and trivial central character is

$$|H_S^+| + |H_S^-| = \prod_{\text{odd } p \mid S} \frac{1}{2}(p-1).$$
 (7-7)

Now suppose S is divisible by a prime $p_0 \equiv 1 \mod 4$. Fix $\epsilon_{\rm fin} = \pm 1$. By the above, the number of tuples $(\sigma_p)_{p|(S/p_0)}$ is $\prod_{{\rm odd}\; p|(S/p_0)} \frac{1}{2}(p-1)$. Having fixed one such tuple, by Corollary 5.2 there are then $\frac{1}{4}(p_0-1)$ choices for σ_{p_0} subject to $\prod_{p|S} \epsilon_{\sigma_p} = \epsilon_{\rm fin}$. This proves the result when $D_4(S) = 0$.

Now suppose $p \equiv 3 \mod 4$ for all odd $p \mid S$. For this case, in view of (7-7), the given formula is equivalent to $|H_S^+| - |H_S^-| = (-1)^{\delta(2 \mid S)}$. We will prove the latter by induction on the number $\omega(S)$ of primes dividing S. If S = 2, the given formula holds since there is just one representation σ_2 , and it has $\epsilon_{\sigma_2} = -1$. If $S = p \equiv 3 \mod 4$, the given formula holds by Corollary 5.2. Having established the base case, suppose now that the given formula holds for some S satisfying $D_4(S) = 1$, and that $p_0 \nmid S$ is a prime satisfying $p_0 \equiv 3 \mod 4$. We construct a tuple $\hat{\sigma} = (\sigma_p)_{p \mid Sp_0}$ by first choosing the components at $p \mid S$, and then at p_0 . Let $P = |H_S^+|$ and $Q = |H_{p_0}^+|$, so $|H_S^-| = P - (-1)^{\delta(2 \mid S)}$ and $|H_{p_0}^-| = Q - 1$ by the inductive hypothesis. Then

$$|H_{Sp_0}^+| = PQ + (P - (-1)^{\delta(2|S)})(Q - 1) = 2PQ - P - (-1)^{\delta(2|S)}Q + (-1)^{\delta(2|S)},$$
 and

$$|H_{Sp_0}^-| = P(Q-1) + (P-(-1)^{\delta(2|S)})Q = 2PQ - P - (-1)^{\delta(2|S)}Q.$$

Subtracting,

$$|H_{Sp_0}^+| - |H_{Sp_0}^-| = (-1)^{\delta(2|Sp_0)},$$

as needed.

Proposition 7.6. For S > 1 square-free, let

$$\Delta(S^2, k)^{\min} = \dim S_k^{\min}(S^2)^+ - \dim S_k^{\min}(S^2)^-.$$

Then for k > 2 even,

$$\Delta(S^2, k)^{\min} = \Delta_M + \Delta_{A_1} + \Delta_{A_2}, \tag{7-8}$$

where

$$\Delta_M = D_4(S)(-1)^{k/2 + \delta(2|S)} \frac{1}{12}(k-1) \prod_{p|S} (p-1), \quad \Delta_{A_1} = \frac{1}{4}(D_4(S)) \prod_{p|S} (p-1)$$

for $D_4(S)$ as in Lemma 7.5 above, and

$$\Delta_{A_2} = \delta(k \equiv 0, 2 \mod 6) \frac{1}{3} (D_3(S)) (-1)^{\delta(k \equiv 6, 8 \mod 12)} \mu(S) \Omega_0(S'),$$

where $D_3(S) \in \{0,1\}$ is 0 if and only if $p \equiv 1 \mod 3$ for some $p \mid S$, $\mu(S) = \prod_{p \mid S} (-1)$ is the Möbius function, and for $S' = S/\gcd(S,6)$,

$$\Omega_0(S') = \begin{cases} 0 & \text{if there exists } p \mid S' \text{ such that } p \equiv 5 \mod 12, \\ 2^{\omega(S')} & \text{if } p \equiv 11 \mod 12 \text{ for all } p \mid S', \end{cases}$$

where $\omega(S') = \sum_{p \mid S'} 1$. (Note that $\Omega_0(1) = 1$.)

Remark. Proposition 1.4, which summarizes the conditions under which $\Delta(S^2, k)^{\min}$ vanishes, is positive, or is negative, follows easily. The claim in the third paragraph of Proposition 1.4 is due to the fact that when $D_4(S) = 1$,

$$\begin{split} |\Delta_{A_1} + \Delta_{A_2}| &\leq \frac{1}{4} \prod_{p \mid S} (p-1) + \frac{1}{3} \prod_{p \mid S'} 2 \\ &= \left[\frac{1}{4} + \frac{1}{3} \prod_{p \mid S'} \frac{2}{p-1} \prod_{p \mid \gcd(S,6)} \frac{1}{p-1} \right] \prod_{p \mid S} (p-1) \leq \frac{7}{12} \prod_{p \mid S} (p-1), \end{split}$$

where the last inequality is strict if S > 2. So if $k \ge 10$, or k = 8 and S > 2, it follows that $|\Delta_{A_1} + \Delta_{A_2}| < |\Delta_M|$, and hence the sign of Δ_M is the sign of the bias. One checks by hand (or LMFDB) that $S_8^{\min}(2^2) = 0$. The case k = 6 follows similarly, replacing the rightmost inequality by $< \frac{5}{12} \prod_{p \mid S} (p-1)$ when S > 6 and $D_4(S) = 1$, and checking the $S \mid 6$ cases by hand.

Proof of Proposition 7.6. We have

$$\Delta(S^2, k)^{\min} = \sum_{\hat{\sigma}: \epsilon(k, \hat{\sigma}) = 1} \dim S_k(\hat{\sigma}) - \sum_{\hat{\sigma}: \epsilon(k, \hat{\sigma}) = -1} \dim S_k(\hat{\sigma}). \tag{7-9}$$

Applying Theorem 1.3 to each summand, we get a sum of three terms as in (7-8). Since the archimedean factor of the global root number is $(-1)^{k/2}$ (see [20, Theorem 14.17] and [9]), the set of tuples $\hat{\sigma}$ with global root number ϵ is $H_S^{(-1)^{k/2}\epsilon}$, with notation as in Lemma 7.4. Therefore the contribution of the main term is

$$\Delta_M = \frac{1}{12}(k-1) \prod_{p \mid S} (p-1)(|H_S^{(-1)^{k/2}}| - |H_S^{-(-1)^{k/2}}|),$$

and using Lemma 7.5 we obtain the formula given for Δ_M .

Likewise, the contribution of the A_1 term of Theorem 1.3 to (7-9) is

$$\begin{split} \Delta_{A_1} &= |H_S^{(-1)^{k/2}}| \frac{1}{4} D_4(S) \cdot 1 \prod_{\text{odd } p \mid S} 2 - |H_S^{-(-1)^{k/2}}| \frac{1}{4} D_4(S) \cdot (-1) \prod_{\text{odd } p \mid S} 2 \\ &= \frac{1}{4} D_4(S) (|H_S^+| + |H_S^-|) \prod_{\text{odd } p \mid S} 2, \end{split}$$

and the given formula follows from (7-7).

In the notation of Theorem 1.3 and Lemma 7.4, the contribution of A_2 to (7-9) is

$$\Delta_{A_2} = \frac{1}{3} (D_3(S) b(k) (-1)^{\delta(3|S)}) (\mathcal{B}(S)^{(-1)^{k/2}} - \mathcal{B}(S)^{-(-1)^{k/2}})$$

= $\frac{1}{3} (D_3(S) b(k) (-1)^{\delta(3|S) + k/2}) (\mathcal{B}(S)^+ - \mathcal{B}(S)^-).$

By considering possibilities for gcd(6, S), it is easy to check using Lemma 7.4 that

$$\mathcal{B}(S)^+ - \mathcal{B}(S)^- = (-1)^{\delta(2|S)} \mu(S') \Omega_0(S').$$

The result then follows from $(-1)^{\delta(2|S)+\delta(3|S)}\mu(S')=\mu(S)$ and the fact that

$$(-1)^{k/2}b(k) = \begin{cases} 1 & \text{if } k \equiv 0, 2 \text{ mod } 12, \\ -1 & \text{if } k \equiv 6, 8 \text{ mod } 12, \\ 0 & \text{if } k \equiv 4 \text{ mod } 6. \end{cases}$$

By similar arguments, we obtain the dimension of the space of twist-minimal forms of level S^2 .

Proposition 7.7. For S > 1 square-free and k > 2 even,

$$\dim S_k^{\min}(S^2) = \frac{1}{12}(k-1) \prod_{odd \ p \mid S} \frac{1}{2}(p-1)^2 + \frac{1}{4}(D_4(S))(-1)^{\delta(2|S) + k/2} \prod_{odd \ p \mid S} 2 + \frac{1}{3}(D_3(S) \ b(k))(-1)^{\delta(3|S)} \prod_{p \mid (S/\gcd(6,S))} 2$$

for

$$b(k) = \begin{cases} 1 & \text{if } 6 \mid k, \\ -1 & \text{if } k \equiv 2 \mod 6, \\ 0 & \text{otherwise.} \end{cases}$$

Remark. Although we have assumed k > 2, the above formula is valid when k = 2 as well. More generally, the dimension of $S_k^{\min}(N, \chi)$ has been computed by Child [8, Section 5.1].

Proof. We have

$$\dim S_k(S^2)^{\min} = d_M + d_{A_1} + d_{A_2},$$

where

$$d_{M} = \frac{1}{12}(k-1) \prod_{p \mid S} (p-1)(|H_{S}^{(-1)^{k/2}}| + |H_{S}^{-(-1)^{k/2}}|),$$

$$d_{A_{1}} = \frac{1}{4}(D_{4}(S))(-1)^{k/2}(|H_{S}^{+}| - |H_{S}^{-}|) \prod_{\text{odd } p \mid S} 2,$$

and

$$d_{A_2} = \frac{1}{3} (D_3(S) b(k) (-1)^{\delta(3|S)}) (\mathcal{B}(S)^+ + \mathcal{B}(S)^-).$$

The result follows upon applying (7-7) to d_M , Lemma 7.5 to d_{A_1} , and the fact that $\mathcal{B}(S)^+ + \mathcal{B}(S)^- = 2^{\omega(S')} = \prod_{p \mid S'} 2$, for $S' = S/\gcd(6, S)$.

7.2. Simplification when n = 1 and T > 1. We return to the general setting of Theorem 4.2 with no constraint on the conductor exponents of the σ_p . Our aim here is to cull the list of matrices that appear in Theorem 4.2 when n = 1 and T > 1. The result is Proposition 7.9, from which Theorem 7.1 follows.

Recall that for $p \mid T$, σ_p is a supercuspidal representation whose conductor is of the form p^n with $n \geq 3$ odd. It is well known (see, e.g., [6, Section A.3.8]) that there is a ramified quadratic extension E/\mathbb{Q}_p with E^* embedded in $G(\mathbb{Q}_p)$ such that σ_p is compactly induced from a character χ of $J_n = E^*U^{(n-1)/2}$, where $U^r = 1 + \binom{p\mathbb{Z}_p}{p\mathbb{Z}_p} \binom{\mathbb{Z}_p}{p\mathbb{Z}_p}^r$ is an open compact subgroup of $G(\mathbb{Q}_p)$ and $\chi|_{F^*} = \omega_p$. In the notation of Section 5.2, U^1 coincides with K', J_3 with H', and in general $J_n \subseteq H'$. We use the local test function defined for $g \in G(\mathbb{Q}_p)$ by

$$f_p(g) = \begin{cases} d_{\sigma_p} \overline{\chi(g)} & \text{if } g \in J_n, \\ 0 & \text{otherwise,} \end{cases}$$
 (7-10)

where d_{σ_p} is the formal degree (depending only on the conductor). This coincides with (5-18) when n = 3.

If $p \mid T$, the support of f_p is the disjoint union of its unramified and ramified elements:

$$\operatorname{Supp}(f_p) = J_n = (J_n \cap ZK') \cup (J_n \cap \pi_E ZK'), \tag{7-11}$$

where π_E is a prime element of E whose square is a prime element of \mathbb{Q}_p . We may decompose f_p as $f_p = f_u + f_r$, a sum of two functions supported on the unramified and ramified elements of J_n respectively. In the paper of Gross [17, p. 1240], discussed in Section 1.3, n = 3 and the local test function used is a multiple of f_u . The following is largely contained in [17, Proposition 5.1].

Proposition 7.8. Let $f^1 = f^n$ for n = 1. Suppose γ is elliptic in $G(\mathbb{Q})$ and unramified at some prime $p \mid T$. Then either γ has p-torsion in $\overline{G}(\mathbb{Q})$ and $p \in \{2, 3\}$, or $\Phi(\gamma, f^1) = 0$. As a result, $\Phi(\gamma, f^1) = 0$ in each of the following situations:

- (1) γ is unramified at some prime $p \mid T$ with p > 3.
- (2) γ is unramified at $3 \mid T$ and $T \neq 3$.

Proof. Write $f = f^1$. Suppose $\Phi(\gamma, f) \neq 0$. By Proposition 4.3, γ is elliptic in $G(\mathbb{R})$ and $\det \gamma > 0$. Hence it belongs to a compact-mod-center subgroup U_{∞} of $G(\mathbb{R})$ (U_{∞} being some conjugate of $\mathbb{R}^* \cdot \mathrm{SO}(2)$). Likewise, at every finite place v, the support of f_v is a compact-mod-center subgroup J_v of G_v (here is where we use n = 1), and γ belongs to some conjugate U_v of J_v . (In fact since $\gamma \in K_v$ a.e., we can take $U_v = K_v$ a.e.) Hence γ belongs to a compact-mod-center subgroup $\prod_v U_v$ of $G(\mathbb{A})$. Identifying γ with its image modulo the center, we have

$$\gamma \in \overline{G}(\mathbb{Q}) \cap \prod_{v} \overline{U}_{v}.$$

This is a *finite* group since $\overline{G}(\mathbb{Q})$ is discrete in $\overline{G}(\mathbb{A})$ [23, Section 7.11]. In particular, γ is a torsion element of $\overline{G}(\mathbb{Q})$, i.e., some power of γ lies in the center $Z(\mathbb{Q})$.

Since γ is unramified at $p \mid T$, some conjugate of γ belongs to the unramified part of the support of f_p , which is a subset of the pro-p group \overline{K}' . (Recall that K' is the pro-p-Sylow subgroup of the Iwahori subgroup of $G(\mathbb{Q}_p)$). It follows that the order of γ in $\overline{G}(\mathbb{Q})$ is a power of p. However, it is known that any torsion element of $\overline{G}(\mathbb{Q})$ has order 1, 2, 3, 4, or 6 [11, Lemma 1]. Since $\gamma \neq 1$, we conclude that $p \leq 3$. This proves (1).

The 3-torsion elements of $\overline{G}(\mathbb{Q})$ comprise a single conjugacy class containing $\binom{0-1}{1-1}$ [11, Lemma 1]. Therefore, if p=3, γ is conjugate in $G(\mathbb{Q})$ to a matrix of the form $\binom{0-z}{z}$ and is hence everywhere unramified. By the above, this means T is not divisible by any prime p>3. It is also odd, because otherwise γ would somehow simultaneously have 3-torsion and 2-power torsion. Hence T=3, which proves (2).

By the same reference, the 4-torsion elements of $\overline{G}(\mathbb{Q})$ are all conjugate to $\binom{1-1}{1-1}$. But such an element is ramified at 2. Hence γ has 2-torsion if p=2.

Proposition 7.9. With notation as in Section 4.1, let T be the product of the primes p for which $\operatorname{ord}_p(N)$ is odd, and for $p \mid T$ take f_p as in (7-10). Then for $\gamma \in \overline{G}(\mathbb{Q})$, $\Phi(\gamma, f^1) = 0$ unless either $\gamma = 1$ or the conjugacy class of γ has a representative in $G(\mathbb{Q})$ of one of the forms given in the table below:

form of T	list of relevant elliptic γ for $n = 1$
even $T \neq 2$	$\binom{-T}{1}$, $\binom{-T/2}{1}$
T=2	$\binom{-2}{1}, \binom{-1}{1}, \binom{0-2}{1-2}$
odd T > 3	$\binom{-T}{1}$
T=3	$\binom{-3}{1}$, $\binom{0}{1}$, $\binom{0}{1}$, $\binom{0}{1}$
T = 1	$\binom{-1}{1}$, $\binom{0}{1}$

Remark. When $T/2 \equiv 7 \mod 8$, the matrix $\binom{-T/2}{1}$ is hyperbolic (rather than elliptic) in $G(\mathbb{Q}_2)$, so its orbital integral vanishes. All other entries in the above

table are elliptic in $G(\mathbb{Q}_p)$ for each $p \mid T$, but for $p \mid S$ this needs to be checked on a case-by-case basis.

Proof. The case where T=1 is already contained in Theorem 4.2, taking n=1. So suppose T>1 and $\Phi(\gamma, f)\neq 0$. By Proposition 4.13, we may take $\gamma=\begin{pmatrix} 0 & -M\\ 1 & rM \end{pmatrix}$ for some $M\mid T$ and $0\leq r<\sqrt{4/M}$. Notice that if M>3 then r=0. Suppose first that $T\neq 3$. By Proposition 7.8, γ must be ramified at all odd primes dividing T, so M=T or M=T/2. If T is odd, this means M=T and we obtain the third row of the above table. Suppose T is even and M=T/2. By Proposition 7.8, γ has 2-torsion in $\overline{G}(\mathbb{Q})$. Note that

$$\gamma^2 = \begin{pmatrix} -M & -rM^2 \\ rM & r^2M^2 - M \end{pmatrix}$$

is a scalar matrix if and only if r=0. Therefore $\gamma=\binom{0-M}{1-0}$. This establishes the top two rows of the table. (When M=T=2, r=1 is admissible, and for $\gamma=\binom{0-2}{1-2}$, $P_{\gamma}(X)=X^2-2X+2$ is an Eisenstein polynomial for the prime 2, which is indeed irreducible in $\mathbb{Q}_2[X]$ [50, p. 19].)

Now suppose T=3. Then M=1 or M=3. In the latter case, $\gamma=\binom{0-3}{1\ 3r}$ for r=0,1. If M=1, then $\gamma=\binom{0-1}{1\ r}$ for r=0,1, and γ is unramified at 3. If r=0, this matrix has 2-torsion, in violation of Proposition 7.8. Hence $\gamma=\binom{0-1}{1\ 1}$. (In this case, $P_{\gamma}(X)=X^2-X+1$ has discriminant -3, which is not a square in \mathbb{Q}_3 , and hence γ is indeed elliptic in $G(\mathbb{Q}_3)$.)

7.3. Global orbital integrals for n = 1, $N = T^3$. Here we will evaluate the global elliptic orbital integrals of Theorem 7.1 explicitly when $N = T^3 > 1$ for T squarefree. We must consider

$$\gamma = \begin{pmatrix} -T \\ 1 \end{pmatrix}, \begin{pmatrix} -T/2 \\ 1 \end{pmatrix}_{(T \text{ even})}, \begin{pmatrix} 0 & -2 \\ 1 & 2 \end{pmatrix}_{(T=2)}, \begin{pmatrix} 0 & -1 \\ 1 & 1 \end{pmatrix}_{(T=3)}, \begin{pmatrix} 0 & -3 \\ 1 & 3 \end{pmatrix}_{(T=3)}$$

as appearing in Proposition 7.9.

We introduce some notation before stating the global results. Given our tuple $\hat{\sigma} = (\sigma_{t_p}^{\zeta_p})_{p|T}$ of simple supercuspidal representations, for k > 2 define

$$\epsilon(k,\hat{\sigma}) = i^k \prod_{p \mid N} \zeta_p. \tag{7-12}$$

This is the common global root number of the cusp forms comprising $H_k(\hat{\sigma})$ (see Proposition 5.3 and [9; 20, Theorem 14.17]). Throughout this section $f = f^1$ as in (5-21).

Proposition 7.10. For $N = T^3$, with notation as above, suppose that for each odd prime factor p of the square-free integer T > 1, $-pt_p/T$ is a square modulo p.

Then for $k \geq 4$ even,

$$\Phi\left(\begin{pmatrix} & -T \\ 1 & \end{pmatrix}, f\right) = \frac{\overline{\epsilon(k, \hat{\sigma})} \, 2_7 \, 4_3 \, h(-T)}{3_{T=3} \, 2^{\omega(T)}} \sum_{\mathbf{y}} \omega'(\mathbf{y}),$$

where numbers with subscripts are present only when T falls into the subscript's equivalence class modulo 8, $3_{T=3}$ is a factor of 3 which is present only when T=3, and y ranges over all integers modulo T that satisfy $y^2 \equiv -pt_p/T \mod p$ for all $p \mid T$. If the central character is trivial, the above simplifies to

$$\Phi\left(\begin{pmatrix} -T\\1 \end{pmatrix}, f\right) = \frac{\epsilon(k, \hat{\sigma})h(-T)w_T}{3_{T=3}},\tag{7-13}$$

where

$$w_T = \begin{cases} \frac{1}{2} & \text{if } T \text{ is even,} \\ 1 & \text{if } T \equiv 1 \mod 4, \\ 2 & \text{if } T \equiv 7 \mod 8, \\ 4 & \text{if } T \equiv 3 \mod 8. \end{cases}$$

Remark. If the first hypothesis is not satisfied or k is odd, then $\Phi(\gamma, f) = 0$; see Proposition 5.6.

Proof. Take $\gamma = \binom{1}{1}^{-T}$, $\Delta_{\gamma} = -4T$, and let M be the odd part of T, so that $T = 2^a M$ for some $a \in \{0, 1\}$. Corresponding to (1-4), write

$$\Phi(\gamma, f) = m\Phi_{\infty}\Phi_2 \prod_{p \mid M} \Phi_p = m(-1)^{k/2} \Phi_2 \prod_{p \mid M} \overline{\zeta_p} \sum_{y_p} \omega_p(y_p),$$

where we have applied (4-13) and Proposition 6.4, with y_p running over the two (since p is odd) solutions to $y_p^2 \equiv -pt_p/T \mod p$. We can exchange the sum and product. To each of the $2^{\omega(M)}$ tuples $(y_p)_{p|M}$, the Chinese remainder theorem assigns a unique integer y modulo T satisfying $y \equiv y_p \mod p$ for all p|T, where we take $y_2 = 1$ if T is even. Further,

$$\omega'(y) = \prod_{p \mid T} \omega_p(y) = \prod_{p \mid T} \omega_p(y_p) = \prod_{p \mid M} \omega_p(y_p).$$

The first equality holds because gcd(y, T) = 1 (see [23, (12.4)]); the second holds since each ω_p is trivial on $1 + p\mathbb{Z}_p$. By Example 4.10 (for T odd) or Proposition 6.4 (for T even),

$$\Phi_2 = \begin{cases} \overline{\zeta_2} & \text{if } T \text{ is even,} \\ 2 & \text{if } T \equiv 1, 5, 7 \mod 8, \\ 4 & \text{if } T \equiv 3 \mod 8. \end{cases}$$

It follows that

$$\Phi(\gamma, f) = \frac{2h(E)}{w_E 2^{\omega(d_E)}} \overline{\epsilon(k, \hat{\sigma})} a_T \sum_{y} \omega'(y)$$

for y as in the statement of the proposition,

$$a_T = \begin{cases} 1 & \text{if } T \text{ is even,} \\ 2 & \text{if } T \equiv 1, 5, 7 \mod 8, \\ 4 & \text{if } T \equiv 3 \mod 8, \end{cases}$$

and $E = \mathbb{Q}(\sqrt{-T})$. Since T > 1, we know that

$$w_E = |\mathcal{O}_E^*| = \begin{cases} 6 & \text{if } T = 3, \\ 2 & \text{otherwise.} \end{cases}$$

So $\frac{1}{2}w_E = 3_{T=3}$ and $2h(E)/w_E = h(-T)/3_{T=3}$. Recall that

$$d_E = \begin{cases} -4T, & -T \equiv 2, 3 \mod 4, \\ -T, & -T \equiv 1 \mod 4. \end{cases}$$

Therefore, placing the congruence condition on T rather than -T,

$$2^{\omega(d_E)} = \begin{cases} 2 \cdot 2^{\omega(T)} & \text{if } T \equiv 1 \mod 4, \\ 2^{\omega(T)} & \text{if } T \equiv 2, 3 \mod 4. \end{cases}$$

Hence using the definition of a_T in the following numerator,

$$\Phi(\gamma, f) = \overline{\epsilon(k, \hat{\sigma})} h(-T) \frac{2_{1,5,7} \cdot 4_3}{3_{T=3} \cdot 2_{1,5} \cdot 2^{\omega(T)}} \sum_{\mathbf{y}} \omega'(\mathbf{y}),$$

where numbers with subscripts are only present when T falls into one of the subscript equivalence classes modulo 8. The general result now follows.

If ω' is trivial, the sum over y equals the number of terms, namely $2^{\omega(M)}$. Then (7-13) follows from

$$\frac{2^{\omega(M)}}{2^{\omega(T)}} = \begin{cases} 1 & \text{if } T \text{ is odd,} \\ \frac{1}{2} & \text{if } T \text{ is even} \end{cases}$$

and the fact that $\epsilon(k, \hat{\sigma}) \in \{\pm 1\}$ is real in this case.

Proposition 7.11. For $N = T^3$, suppose that the square-free integer T = 2M is even, and that for each prime factor p of T, $-pt_p/M$ is a square modulo p. Then for even $k \ge 4$,

$$\Phi\left(\begin{pmatrix} -M \\ 1 \end{pmatrix}, f\right) = h(-M) \frac{\overline{\epsilon(k, \hat{\sigma})}}{\zeta_2} \cdot \frac{z_M}{2_{M=1} \, 3_{M=3} \, 2^{\omega(M)}} \sum_{y} \omega'(y),$$

where $2_{M=1}$ is a factor of 2 which is present only when M=1, $3_{M=3}$ is defined similarly,

$$z_M = \begin{cases} \frac{1}{2} & \text{if } M \equiv 1 \mod 4, \\ -3 & \text{if } M \equiv 3 \mod 8, \\ 0 & \text{if } M \equiv 7 \mod 8, \end{cases}$$

and y ranges over all elements modulo M that satisfy $y^2 \equiv -pt_p/M \mod p$ for each $p \mid M$. If ω' is trivial, the sum over y simply cancels with the factor of $2^{\omega(M)}$. (Again, if the condition on the t_p fails to hold or k is odd, the orbital integral vanishes.)

Proof. We use the same proof as for the previous proposition, with minor modifications. First, by Example 6.6,

$$\Phi\left(\begin{pmatrix} -M \\ 1 \end{pmatrix}, f_2\right) = \begin{cases} 1 & \text{if } M \equiv 1 \mod 4, \\ -3 & \text{if } M \equiv 3 \mod 8, \\ 0 & \text{if } M \equiv 7 \mod 8. \end{cases}$$

Taking $E = \mathbb{Q}[\sqrt{-M}]$ we have

$$2^{\omega(d_E)} = \begin{cases} 2 \cdot 2^{\omega(M)} & \text{if } M \equiv 1 \mod 4, \\ 2^{\omega(M)} & \text{if } M \equiv 3 \mod 4 \end{cases}$$

as in the previous proof, and $2h(E)/w_E = h(-M)/(3_{M=3} \, 2_{M=1})$ since $\mathbb{Q}[\sqrt{-1}]$ has unit group of order 4 when M=1. Hence (assuming $M \not\equiv 7 \mod 8$)

$$\Phi\left(\begin{pmatrix} -M \\ 1 \end{pmatrix}, f\right) = \frac{h(-M)(-3)_3}{3_{M=3} \, 2_{M=1} \, 2_{1,5} \, 2^{\omega(M)}} \frac{\overline{\epsilon(k, \hat{\sigma})}}{\zeta_2} \sum_{v} \omega'(v),$$

where numerical subscripts refer to the congruence class of M modulo 8.

Proposition 7.12. Suppose $N=2^3$, $\zeta \in \{\pm 1\}$ and $\sigma = \sigma^{\zeta}$ is our fixed simple supercuspidal representation of $G(\mathbb{Q}_2)$ (the parameter t must equal 1 when p=2). Then

$$\Phi\left(\begin{pmatrix} 0 & -2 \\ 1 & 2 \end{pmatrix}, f\right) = \frac{1}{4}\epsilon(k, \sigma) g_8(k),$$

where $g_8(k) = -1$ if $k \equiv 0, 2 \mod 8$, and $g_8(k) = 1$ if $k \equiv 4, 6 \mod 8$.

Remark. In view of Proposition 5.4, we assume that k is even.

Proof. Given that γ has characteristic polynomial $X^2 - 2X + 2$ with discriminant $\Delta_{\gamma} = -4$, we find $E = \mathbb{Q}[\gamma] = \mathbb{Q}[i]$. Hence h(E) = 1, $w_E = |\mathcal{O}_E^*| = 4$, and $d_E = -4$. By (1-4),

$$\Phi(\gamma, f) = m\Phi_{\infty}\Phi_2 = \frac{1}{4}\Phi_{\infty}\Phi_2.$$

Applying Proposition 6.4 with p = 2 and v = 1, we have $\Phi_2 = -\zeta$. So

$$\Phi(\gamma, f) = -\frac{1}{4}\Phi_{\infty}\zeta. \tag{7-14}$$

The complex eigenvalues of γ are $1 \pm i$, so we apply (4-12) with $\theta = \frac{\pi}{4}$ to get

$$\Phi_{\infty} = -\sqrt{2} \sin(\frac{1}{4}(k-1)\pi) = \begin{cases} 1 & \text{if } k \equiv 0, 6 \mod 8, \\ -1 & \text{if } k \equiv 2, 4 \mod 8. \end{cases}$$

Multiplying this by -1 as in (7-14) yields $(-1)^{k/2}g_8(k)$ with g_8 as given.

Proposition 7.13. Suppose T = 3 so $N = 3^3$, and let $\sigma = \sigma_t^{\zeta}$ be our fixed simple supercuspidal representation of $G(\mathbb{Q}_3)$, for $t = \pm 1$. Then

$$\Phi\left(\begin{pmatrix} 0 & -1 \\ 1 & 1 \end{pmatrix}, f\right) = \frac{t}{2_{t-1}} c_3(k),$$

where $c_3(k) = \frac{1}{3} + \left\lfloor \frac{k}{3} \right\rfloor - \frac{k}{3}$.

Proof. Let $E = \mathbb{Q}[\gamma] = \mathbb{Q}[\sqrt{-3}]$. Then h(E) = 1, $d_E = -3$, and $w_E = |\mathcal{O}_E^*| = 6$. By (1-4) and taking m = 1 in Example 6.7 and its remark,

$$\Phi(\gamma, f) = \Phi_{\infty} \cdot \frac{1}{6} ((-1)^k t \cdot 2_{t=1}) = \frac{1}{3} ((-1)^k \Phi_{\infty}) \frac{t}{2_{t=-1}}.$$

By (4-12), we find that

$$(-1)^k \Phi(\gamma, f_{\infty}) = (-1)^{k+1} \frac{\sin(\frac{(k-1)\pi}{3})}{\sin(\pi/3)} = \begin{cases} 1 & \text{if } k \equiv 0 \mod 3, \\ 0 & \text{if } k \equiv 1 \mod 3, \\ -1 & \text{if } k \equiv 2 \mod 3. \end{cases}$$
(7-15)

Using the above, we see that $\frac{1}{3}((-1)^k\Phi(\gamma, f_\infty)) = \frac{1}{3} + \left\lfloor \frac{k}{3} \right\rfloor - \frac{k}{3}$.

Proposition 7.14. Suppose $N = 3^3$, and let $\sigma = \sigma_t^{\zeta}$ be a fixed simple supercuspidal representation of $G(\mathbb{Q}_3)$. Then

$$\Phi\left(\begin{pmatrix} 0 & -3 \\ 1 & 3 \end{pmatrix}, f\right) = \begin{cases} 0 & \text{if } t = 1, \\ \epsilon(k, \sigma) g_6(k) & \text{if } t = -1, \end{cases}$$

where

$$g_6(k) = \begin{cases} 0 & \text{if } k \equiv 1 \mod 6, \\ -\frac{1}{6} & \text{if } k \equiv 0, 2 \mod 6, \\ \frac{1}{2} & \text{if } k \equiv 3 \mod 6, \\ \frac{1}{3} & \text{if } k \equiv 4 \mod 6, \\ -\frac{1}{2} & \text{if } k \equiv 5 \mod 6. \end{cases}$$

Proof. Let $\gamma = \binom{0-3}{1-3}$, so $\Delta_{\gamma} = -3$. We have $E = \mathbb{Q}[\gamma] = \mathbb{Q}[\sqrt{-3}]$, so the measure factor is $\frac{1}{6}$ as in the previous proof. Therefore as in (1-4), we may write

$$\Phi(\gamma, f) = \frac{1}{6} \Phi_{\infty} \Phi_3. \tag{7-16}$$

By Proposition 6.4, $\Phi_3 = 0$ unless -t is a square modulo 3, i.e., unless t = -1. Assuming this holds, we have

$$\Phi_3 = \bar{\zeta} \cdot (\overline{\psi(1)}\omega_3(1) + \overline{\psi(-1)}\omega_3(-1))$$

= $\bar{\zeta} \cdot (e^{-2\pi i/3} + (-1)^k e^{2\pi i/3}) = -\bar{\zeta} [i\sqrt{3}]_{k \text{ odd}},$

where the factor of $i\sqrt{3}$ is present only when k is odd.

By (4-4) with N=3, $\omega_3(3)=1$. So $\zeta^2=\omega_3(t)=\omega_3(-1)=(-1)^k$, so $\zeta=\pm(i^k)$. In particular, the global root number $\varepsilon(\sigma,\zeta)=i^k\zeta$ is real and $\bar{\zeta}=(-1)^k\zeta$.

The complex roots of $P_{\gamma}(X) = X^2 - 3X + 3$ are $\frac{1}{2}(3 \pm i\sqrt{3}) = \sqrt{3}(\frac{1}{2}(\sqrt{3} \pm i))$, so in (4-12) we can take $\theta = \frac{\pi}{6}$ and $\Phi(\gamma, f_{\infty}) = -2\sin(\frac{1}{6}(k-1)\pi)$. Hence (7-16) becomes

$$\Phi(\gamma, f) = \frac{1}{3} (-1)^k \zeta \sin(\frac{1}{6}(k-1)\pi) [i\sqrt{3}]_{k \text{ odd}} = \begin{cases} \zeta/3 & \text{if } k \equiv 4 \text{ mod } 12, \\ -i\zeta/2 & \text{if } k \equiv 3, 5 \text{ mod } 12, \\ \zeta/6 & \text{if } k \equiv 2, 6 \text{ mod } 12, \\ 0 & \text{if } k \equiv 1, 7 \text{ mod } 12, \\ -\zeta/6 & \text{if } k \equiv 0, 8 \text{ mod } 12, \\ i\zeta/2 & \text{if } k \equiv 9, 11 \text{ mod } 12, \\ -\zeta/3 & \text{if } k \equiv 10 \text{ mod } 12. \end{cases}$$

Upon factoring out $\epsilon(k, \sigma) = i^k \zeta$, we obtain $g_6(k)$ as given.

7.4. Dimension formulas when $N = T^3$. Here we put everything together to compute $|H_k(\hat{\sigma})| = \dim S_k(\hat{\sigma})$ for $\hat{\sigma} = (\sigma_p)_{p|N}$ a tuple of simple supercuspidal representations of $G(\mathbb{Q}_p)$ as in Theorem 7.1 with S = 1.

We begin with the case $N = 2^3$, where the central character is necessarily trivial due to (4-2) and Proposition 5.4.

Theorem 7.15. Let $N = 2^3$, fix $\zeta \in \{\pm 1\}$, and let $\sigma = \sigma_{\zeta}$ be the associated simple supercuspidal representation of $G(\mathbb{Q}_2)$ with trivial central character. Then

$$|H_k(\sigma)| = \begin{cases} 0 & \text{if } k \text{ is odd,} \\ \left\lfloor \frac{k}{8} \right\rfloor & \text{if } k \equiv 0, 2 \text{ mod } 8, \\ \left\lfloor \frac{k}{8} \right\rfloor + \frac{1}{2} (1 + \epsilon(k, \sigma)) & \text{if } k \equiv 4, 6 \text{ mod } 8, \end{cases}$$

where $\epsilon(k, \sigma) = (-1)^{k/2} \zeta$ is the global root number.

Proof. When k is odd, the assertion follows from Proposition 5.4. Suppose k is even. By Theorem 7.1,

$$|H_k(\sigma)| = \frac{1}{12}(k-1) \cdot \frac{3}{2} + \frac{1}{2}\Phi\left(\begin{pmatrix} & -2 \\ 1 & \end{pmatrix}, f\right) + \frac{1}{2}\Phi\left(\begin{pmatrix} & -1 \\ 1 & \end{pmatrix}, f\right) + \Phi\left(\begin{pmatrix} 0 & -2 \\ 1 & 2 \end{pmatrix}, f\right).$$

Applying the results of Section 7.3 using h(-2) = h(-1) = 1, we find

$$|H_k(\sigma)| = \frac{1}{8}(k-1) + \frac{1}{4}((-1)^{k/2}\zeta) + \frac{1}{8}((-1)^{k/2}) + \frac{1}{4}((-1)^{k/2}\zeta)g_8(k)$$

for

$$g_8(k) = \begin{cases} -1 & \text{if } k \equiv 0, 2 \mod 8, \\ 1 & \text{if } k \equiv 4, 6 \mod 8. \end{cases}$$

The result follows upon simplifying each of the cases.

Theorem 7.16. Let $N=3^3$, fix $t \in \{\pm 1\}$, a character ω_3 of \mathbb{Q}_3^* trivial on $1+3\mathbb{Z}_3$, $\zeta \in \mathbb{C}$ with $\zeta^2 = \omega_3(t)$ (see (4-2)), and let $\sigma = \sigma_t^{\zeta}$ be the associated simple supercuspidal representation of $G(\mathbb{Q}_3)$ with central character ω_3 . Then for k > 2, setting $\epsilon = i^k \zeta$, we have

$$|H_k(\sigma)| = \begin{cases} \left\lfloor \frac{k}{3} \right\rfloor + \frac{1}{2}(\epsilon - 1) & \text{if } k \equiv 0 \bmod 3 \text{ and } t = -1, \\ \left\lfloor \frac{k}{3} \right\rfloor & \text{if } k \equiv 1 \bmod 6 \text{ or } t = 1, \\ \left\lfloor \frac{k}{3} \right\rfloor + \frac{1}{2}(\epsilon + 1) & \text{if } k \equiv 2 \bmod 6 \text{ and } t = -1, \\ \left\lfloor \frac{k}{3} \right\rfloor + \epsilon & \text{if } k \equiv 4 \bmod 6 \text{ and } t = -1, \\ \left\lfloor \frac{k}{3} \right\rfloor + \frac{1}{2}(1 - \epsilon) & \text{if } k \equiv 5 \bmod 6 \text{ and } t = -1. \end{cases}$$

Remarks. (1) If t = -1, then $\zeta^2 = \omega_3(-1) = (-1)^k$, so $\zeta = \pm i^k$, as noted earlier. Therefore $\epsilon \in \{\pm 1\}$ when t = -1. When t = 1 and k is odd, $\epsilon = \pm i$.

(2) There is one more newform with $\epsilon = -1$ than with $\epsilon = 1, i$, or -i when $k \equiv 5 \mod 6$, i.e., the root number has a slight bias toward -1 in this case. For example, when k = 5 and ω' is the Dirichlet character of conductor 3, there are five newforms of level 27, with respective root numbers 1, -1, -1, i, -i. These newforms are discussed further in Section 7.5.

Proof. By Theorem 7.1,

$$\begin{split} |H_k(\sigma)| &= \frac{1}{12}(k-1) \cdot \frac{8}{2} + \frac{1}{2}\Phi\bigg(\!\binom{-3}{1},\, f\bigg) + \Phi\bigg(\!\binom{0}{1}-1\!\binom{-1}{1},\, f\bigg) + \Phi\bigg(\!\binom{0}{1}-3\!\binom{-3}{1},\, f\bigg) \\ &= \frac{1}{3}(k-1) + \frac{2\epsilon}{3}\delta_{t=-1} \cdot \delta_{k\in 2\mathbb{Z}} + \frac{t}{2_{t=-1}}c_3(k) + \epsilon g_6(k)\, \delta_{t=-1}, \end{split}$$

where we have applied Propositions 7.10, 7.13, and 7.14, and $c_3(k)$, $g_6(k)$ are recalled below. (For nonvanishing of $\Phi(\binom{1}{1}^{-3})$, f), the hypothesis in Proposition 7.10 requires that -t be a square modulo 3, i.e., t=-1, and k even. Then $\bar{\epsilon}=\epsilon$ and the sum over y in that result is $1+(-1)^k=2$.)

If t = 1, then because $c_3(k) = \frac{1}{3}(1 - k) + \lfloor \frac{k}{3} \rfloor$, the above simplifies to $\lfloor \frac{k}{3} \rfloor$, as needed.

Now suppose t = -1, and write k = a + 6c for some $0 \le a \le 5$. If k is odd, then

$$|H_k(\sigma)| = \frac{1}{3}(k-1) - \frac{1}{2}\left(\frac{1}{3}(1-k) + \left\lfloor \frac{k}{3} \right\rfloor\right) + \epsilon g_6(k) = \frac{1}{2}(k-1) - \frac{1}{2}\left\lfloor \frac{k}{3} \right\rfloor + \epsilon g_6(k).$$

Using the fact that $g_6(k) = 0$, $\frac{1}{2}$, $-\frac{1}{2}$ when a = 1, 3, 5 respectively, we get

$$|H_k(\sigma)| = \begin{cases} 2c = \left\lfloor \frac{k}{3} \right\rfloor & \text{if } a = 1, \\ 2c + 1 + \frac{1}{2}(\epsilon - 1) = \left\lfloor \frac{k}{3} \right\rfloor + \frac{1}{2}(\epsilon - 1) & \text{if } a = 3, \\ 2c + 1 + \frac{1}{2}(1 - \epsilon) = \left\lfloor \frac{k}{3} \right\rfloor + \frac{1}{2}(1 - \epsilon) & \text{if } a = 5. \end{cases}$$

If k is even, then there is one extra term, namely $\frac{2\epsilon}{3}$, so

$$|H_k(\sigma)| = \frac{1}{2}(k-1) - \frac{1}{2} \lfloor \frac{k}{3} \rfloor + \epsilon \left(\frac{2}{3} + g_6(k)\right).$$

Here, $g_6(k) = -\frac{1}{6}, -\frac{1}{6}, \frac{1}{3}$ when a = 0, 2, 4 respectively. Upon simplifying,

$$|H_k(\sigma)| = \begin{cases} 2c + \frac{1}{2}(\epsilon - 1) = \left\lfloor \frac{k}{3} \right\rfloor + \frac{1}{2}(\epsilon - 1) & \text{if } a = 0, \\ 2c + \frac{1}{2}(1 + \epsilon) = \left\lfloor \frac{k}{3} \right\rfloor + \frac{1}{2}(1 + \epsilon) & \text{if } a = 2, \\ 2c + 1 + \epsilon = \left\lfloor \frac{k}{3} \right\rfloor + \epsilon & \text{if } a = 4. \end{cases}$$

Theorem 7.17. Suppose $N = T^3$ with T > 3 square-free, $M = \frac{T}{2}$, $k \ge 4$ is even, and $\hat{\sigma} = (\sigma_{t_p}^{\zeta_p})_{p|N}$ is a tuple of simple supercuspidal representations with trivial central characters. Then

$$|H_{k}(\hat{\sigma})| = \frac{1}{12}(k-1) \prod_{p|T} \frac{1}{2}(p^{2}-1) + \Delta_{1}(\hat{t}) \epsilon(k,\hat{\sigma}) b_{T} h(-T) + \Delta_{2}(\hat{t}) \frac{\epsilon(k,\hat{\sigma}) j_{M} h(-M)}{\zeta_{2} 3_{M-3}}, \quad (7-17)$$

where $\epsilon(k, \hat{\sigma}) \in \{\pm 1\}$ is the common global root number of the newforms in $H_k(\hat{\sigma})$ given in (7-12),

$$b_T = \begin{cases} \frac{1}{4} & \text{if } T \text{ is even,} \\ \frac{1}{2} & \text{if } T \equiv 1 \mod 4, \\ 1 & \text{if } T \equiv 7 \mod 8, \\ 2 & \text{if } T \equiv 3 \mod 8, \end{cases} \quad j_M = \begin{cases} \frac{1}{4} & \text{if } M \equiv 1 \mod 4, \\ -\frac{3}{2} & \text{if } M \equiv 3 \mod 8, \\ 0 & \text{if } M \equiv 7 \mod 8, \end{cases}$$

h(d) is the class number of $\mathbb{Q}[\sqrt{-d}]$, and $\Delta_i(\hat{t}) \in \{0, 1\}$ is nonzero if and only if (i) T is even in the case i = 2, and (ii) $-2^{i-1}pt_p/T$ is a square modulo p for each odd $p \mid T$.

Remarks. To keep the formula simple, we have restricted ourselves to the case of trivial central character; the general case is obtained similarly. Even in the general case, one may restrict to k even because by Corollary 7.2,

$$|H_k(\hat{\sigma})| = \frac{1}{12}(k-1) \prod_{p|N} \frac{1}{2}(p^2-1) \quad (T > 3, k \text{ odd}).$$
 (7-18)

Proof. This follows from Theorem 7.1 and Propositions 7.10 and 7.11. \Box

As a corollary, we recover the following dimension formulas of [37].

Corollary 7.18. For T = 2, 3 and $k \ge 4$ even,

$$\dim S_k^{\text{new}}(8) = \left| \frac{k}{4} \right|, \quad \dim S_k^{\text{new}}(27) = k - 1 + \left| \frac{k}{3} \right|.$$

For T > 3 square-free, and k > 4 even,

$$\dim S_k^{\text{new}}(T^3) = \frac{1}{12}(k-1) \prod_{p|T} (p-1)^2 (p+1). \tag{7-19}$$

Remarks. As shown in [37], the formula is also valid for k = 2. When k is odd and ω' has conductor dividing T, dim $S_k^{\text{new}}(T^3, \omega')$ is also equal to (7-19). This follows from (7-18).

Proof. For T = 2, by Theorem 7.15,

$$|H_k(2^3)| = |H_k(\sigma^+)| + |H_k(\sigma^-)| = \begin{cases} 2\lfloor \frac{k}{8} \rfloor & \text{if } k \equiv 0, 2 \text{ mod } 8, \\ 2\lfloor \frac{k}{8} \rfloor + 1 & \text{if } k \equiv 4, 6 \text{ mod } 8. \end{cases}$$

This is easily seen to be the same as $\left|\frac{k}{4}\right|$.

For T=3, for fixed k we add the formula in Theorem 7.16 over all $t, \zeta \in \{\pm 1\}$, or equivalently, $t, \epsilon \in \{\pm 1\}$. Writing the t=1 contribution first, we obtain

$$|H_k(3^3)| = 2\lfloor \frac{k}{3} \rfloor + \begin{cases} 2\lfloor \frac{k}{3} \rfloor - 1 & \text{if } k \equiv 0 \mod 3, \\ 2\lfloor \frac{k}{3} \rfloor & \text{if } k \equiv 1 \mod 3, \\ 2\lfloor \frac{k}{3} \rfloor + 1 & \text{if } k \equiv 2 \mod 3. \end{cases}$$

The above is easily seen to equal $k-1+\left|\frac{k}{3}\right|$, as required.

For T > 3 we have

dim
$$S_k^{\text{new}}(T^3) = |H_k(T^3)| = \sum_{\hat{\sigma}} |H_k(\hat{\sigma})|,$$

where $\hat{\sigma}$ ranges over the $\prod_{p|T} 2(p-1)$ tuples (t_p, ζ_p) , with trivial central character. By (7-17), this equals

$$\begin{split} \frac{1}{12}(k-1) \prod_{p\mid T} \frac{1}{2}(p^2-1) \, 2(p-1) + \sum_{\hat{\sigma}} \Delta_1(\hat{t}) \, \epsilon(k,\hat{\sigma}) \, b_T \, h(-T) \\ + \sum_{\hat{\sigma}} \Delta_2(\hat{t}) \frac{\epsilon(k,\hat{\sigma}) j_M \, h(-M)}{\zeta_2 \, 3_{M=3}}. \end{split}$$

Clearly, from (7-12), exactly half of the $\hat{\sigma}$ satisfying $\Delta_1(\hat{t}) = 1$ have $\epsilon(k, \hat{\sigma}) = +1$, and half have $\epsilon(k, \hat{\sigma}) = -1$. So the first sum over $\hat{\sigma}$ vanishes. Likewise if T is even, $\epsilon(k, \hat{\sigma})/\zeta_2 = +1$ (resp. -1) exactly half of the time since T is divisible by at least one prime different from 2, so the second sum also vanishes.

Next, we compute the dimension of the subspace of forms with a given root number, which recovers the main result (1-9) of [46].

Corollary 7.19 [46]. For T > 3 square-free and $k \ge 4$ even, the subspace of $S_k^{\text{new}}(T^3)$ with root number ± 1 has dimension

$$|H_k^{\pm}(T^3)| = \frac{1}{24}(k-1) \prod_{p|T} (p-1)^2 (p+1) \pm \frac{1}{2} (c_T h(-T)) \prod_{p|T} (p-1),$$

where $c_T = b_T$ if T is odd, and $c_T = 2b_T$ if T is even, i.e.,

$$c_T = \begin{cases} \frac{1}{2} & \text{if } T \equiv 1, 2 \mod 4, \\ 1 & \text{if } T \equiv 7 \mod 8, \\ 2 & \text{if } T \equiv 3 \mod 8. \end{cases}$$
 (7-20)

Proof. Given $\hat{\sigma} = (\sigma_{t_p}^{\zeta_p})_{p \mid T}$, let $\hat{t} = (t_p)_{p \mid T}$ and $\hat{\zeta} = (\zeta_p)_{p \mid T}$. The root number is determined by $\hat{\zeta}$ and k. Let A_k^{\pm} be the set of all tuples $\hat{\zeta}$ for which $(-1)^{k/2} \prod_{p \mid T} \zeta_p = \pm 1$. Then

$$|A_k^+| = |A_k^-| = \frac{1}{2} \prod_{p|T} 2.$$
 (7-21)

By (7-17), we see that

$$\begin{split} &|H_k^{\pm}(T^3)| \\ &= \sum_{\hat{\zeta} \in A_k^{\pm}} \sum_{\hat{t}} |H_k(\hat{\sigma})| \\ &= \sum_{\hat{\zeta} \in A_k^{\pm}} \sum_{\hat{t}} \left(\frac{1}{12} (k-1) \prod_{p \mid T} \frac{1}{2} (p^2 - 1) \pm b_T h(-T) \Delta_1(\hat{t}) \pm \zeta_2 \frac{j_M h(-M)}{3_{M=3}} \Delta_2(\hat{t}) \right), \end{split}$$

where M is the odd part of T. Recall that $\Delta_2(\hat{t}) = 0$ if T odd. If T is even, upon summing over $\zeta_2 = \pm 1$ the last term will be eliminated, so we can ignore it henceforth. For any given odd prime p, exactly half of the elements $t_p \in (\mathbb{Z}/p\mathbb{Z})^*$ have the property that $-pt_p/T$ is a square. Therefore, the number of tuples \hat{t} for which $\Delta_1(\hat{t}) \neq 0$ is $\prod_{p|M} \frac{1}{2}(p-1)$. The total number of tuples \hat{t} is $\prod_{p|T} (p-1) = \prod_{p|M} (p-1)$. It follows that

$$|H_k^{\pm}(T^3)| = \sum_{\hat{c} \in A_{\pm}^{\pm}} \left(\frac{1}{12} (k-1) \prod_{p \mid T} \frac{1}{2} (p^2 - 1) (p-1) \pm b_T h(-T) \prod_{p \mid M} \frac{1}{2} (p-1) \right).$$

By (7-21), we obtain

$$|H_k^{\pm}(T^3)| = \frac{1}{24}(k-1)\prod_{p\mid T}(p-1)^2(p+1) \pm \frac{1}{2}(2_Tb_Th(-T))\prod_{p\mid T}(p-1),$$

where 2_T is a factor of 2 which is only present when T is even. We see immediately that $2_T b_T = c_T$ as given.

7.5. Some examples with n > 1. In this section we illustrate Theorem 1.1 with some examples. (A different set of examples is given in the earliest version of this paper posted on the arXiv.) We will compare with the Galois orbits of newforms tabulated in LMFDB [34]. Though $S_k(\hat{\sigma})$ occasionally forms a Galois orbit, typically the orbit is a direct sum of more than one such space. It also happens that a space $S_k(\hat{\sigma})$ decomposes as a direct sum of more than one Galois orbit.

Examples of these phenomena can be found in $S_4^{\min}(23^2)$, where Theorem 1.3 gives dim $S_4(\hat{\sigma}) = \frac{1}{2}(11 + \epsilon) \in \{5, 6\}$, but the twist-minimal Galois orbits can have dimensions 1, 2, 5, 6, 12 or 24.

7.5.1. We first consider an example with odd weight. Take $N=3^3$, k=5, and ω' the Dirichlet character of modulus 27 and conductor 3. We consider simple supercuspidal representations σ_t^{ζ} , where $t \in \{\pm 1\}$ and $\zeta^2 = \omega'(t)$. In LMFDB [34] we find the following data for the space $S_5(27, \omega')$:

LMFDB label	ϵ	dim	$\operatorname{tr} T_4$	$\operatorname{tr} T_7$	(ζ,t)
27.5.b.a	1	1	16	71	(-i, -1)
27.5.b.b	-1	2	-76	34	(i, -1)
27.5.b.c	$\pm i$	2	14	-38	$(1,1)\oplus(-1,1)$

The final column, using the shorthand $(\zeta, t) = S_5(\sigma_t^{\zeta})$, is immediate upon comparing Theorem 7.16 with the ϵ and dim columns. Using Theorem 7.17 we find the following, which refines the above.

Example 7.20. With notation as above,

$$\operatorname{tr}(T_4 | S_5(\sigma_t^{\zeta})) = \frac{1}{2}(37t - 23) + 46i\zeta \cdot \delta_{t=-1},$$

$$\operatorname{tr}(T_7 | S_5(\sigma_t^{\zeta})) = \frac{1}{4}(67 - 143t) + \frac{1}{2}(37i\zeta)\delta_{t=-1}.$$

We will give an indication of the proof of the above formulas. The calculations for n = 7 are a little bit more interesting, so we start with this case. By Theorem 1.1, $tr(T_7 | S_5(\sigma_t^{\zeta}))$

$$=7^{3/2}\bigg[\Phi\bigg(\begin{pmatrix} -21\\1&3\end{pmatrix}\bigg)+\Phi\bigg(\begin{pmatrix} -21\\1&6\end{pmatrix}\bigg)+\Phi\bigg(\begin{pmatrix} -21\\1&9\end{pmatrix}\bigg)+\sum_{r=1}^5\Phi\bigg(\begin{pmatrix} -7\\1&r\end{pmatrix}\bigg)\bigg].$$

We have used (4-13) to eliminate the trace zero matrices, since k is odd. The matrix $\binom{7}{1}$ is unramified at p=3 but has no double characteristic root mod 3. So its orbital integral vanishes by Proposition 5.6. The first three integrals vanish unless

$$y^2 \equiv -\frac{t}{7} \equiv -t \mod 3$$

has a solution, i.e., t=-1. In this case, applying Proposition 6.4 to $\gamma=\binom{-21}{9}$ and p=3, we see that v=3 so the local integral has the value $\overline{\zeta_5}(\omega_3(1)+\omega_3(-1))=0$. Hence this γ can be discarded. We compute the remaining orbital integrals locally as summarized in the following table, where $m=2h(E)/(w(E)2^{\omega(d_E)})$ is the global measure factor for $E=\mathbb{Q}[\gamma]$, and ℓ denotes a prime factor of the discriminant Δ_{γ} other than 3 (if such exists). The global orbital integral is then $\Phi=m\Phi_{\infty}\Phi_3\Phi_{\ell}$. The factor

$$\Phi_{\infty} = -\frac{\sin(4\arctan(\sqrt{|\Delta_{\gamma}|}/\operatorname{tr}\gamma))}{\sin(\arctan(\sqrt{|\Delta_{\gamma}|}/\operatorname{tr}\gamma))}$$

was computed using software.

γ	Δ_{γ}	ℓ	m	Φ_{∞}	Φ_3	Φ_ℓ
$\begin{pmatrix} -21 \\ 1 & 3 \end{pmatrix}$	$-3\cdot5^2$	5	$\frac{1}{6}$	$11\sqrt{3}\cdot7^{-3/2}$	$-i\bar{\zeta}\sqrt{3}\cdot\delta_{t=-1}$	7
$\begin{pmatrix} -21 \\ 1 & 6 \end{pmatrix}$	$-2^4 \cdot 3$	2	$\frac{1}{6}$	$4\sqrt{3}\cdot7^{-3/2}$	$i\bar{\zeta}\sqrt{3}\cdot\delta_{t=-1}$	10
$\begin{pmatrix} & -7 \\ 1 & 1 \end{pmatrix}$	-3^{3}		$\frac{1}{6}$	$13\cdot 7^{-3/2}$	4	
$\begin{pmatrix} -7 \\ 1 & 2 \end{pmatrix}$	$-2^3 \cdot 3$	2	$\frac{1}{2}$	$20\cdot 7^{-3/2}$	$\frac{1}{2}(1-3t)$	2
$\begin{pmatrix} -7 \\ 1 & 4 \end{pmatrix}$	$-2^2 \cdot 3$	2	$\frac{1}{6}$	$-8\cdot7^{-3/2}$	$-\frac{1}{2}(3t+1)$	4
$\begin{pmatrix} -7 \\ 1 & 5 \end{pmatrix}$	-3		$\frac{1}{6}$	$-55\cdot7^{-3/2}$	$\frac{1}{2}(3t+1)$	

The formula for tr T_7 in Example 7.20 follows upon simplifying. Most of the entries in the above table are straightforward, but we highlight a few. For example, $\gamma = \binom{-21}{6}$ is elliptic in $G(\mathbb{Q}_2)$, and by the quadratic formula,

$$\mathbb{Z}_2[\gamma] = \mathbb{Z}_2\left[\frac{1}{2}(6+2^2\sqrt{-3})\right] = \mathbb{Z}_2 + \mathbb{Z}_2 2^2 \varepsilon,$$

where $\varepsilon = \frac{1}{2}(1+\sqrt{-3})$. So $n_{\gamma} = 2$ and $\Phi_2(\gamma) = 1 + (2+1) + (4+2) = 10$ by Proposition 4.8 and (4-20).

The matrix $\gamma = \binom{-7}{1}$ is unramified at p = 3, so $\Phi_3(\gamma)$ is computed using Proposition 6.5. We find (using software) that $\mathcal{N}_{\gamma}(0,1) = \mathcal{N}_{\gamma}(0,2) = 3$, $\mathcal{N}_{\gamma}(1,2) = 6$, $\mathcal{N}_{\gamma}(1,3) = 9$, and $\mathcal{N}_{\gamma}(c,n) = 0$ for all other pairs (c,n). Since $P_{\gamma}(X) \equiv (X+1)^2 \mod 3$, we take z = -1, so, using the third remark after Proposition 6.5, for $t = \pm 1$ we have

$$\Phi_3\left(\binom{-7}{1\ 1}\right) = \frac{-1}{3} \left[3\left(e\left(\frac{t}{3}\right) + e\left(\frac{-t}{3}\right)\right) + 3(2) + 6(-1) + 9(-1) \right] = 4.$$

Finally, $\gamma = \binom{-7}{2}$ is unramified at p = 3 and $\mathcal{N}_{\gamma}(-1, 1) = 3$ is the only nonzero value of $\mathcal{N}_{\gamma}(c, n)$. We take z = 1 in Proposition 6.5 to get

$$\Phi_{3}\left(\begin{pmatrix} -7\\1 & 2 \end{pmatrix}\right) = \frac{1}{3} \cdot 3 \left[e\left(\frac{-1-t}{3}\right) + e\left(\frac{1+t}{3}\right) \right] = 2\cos\frac{2\pi(1+t)}{3} = \begin{cases} -1 & \text{if } t = 1, \\ 2 & \text{if } t = -1. \end{cases}$$

This equals $\frac{1}{2}(1-3t)$ for $t=\pm 1$. The remaining entries in the above T_7 table are found in a similar fashion.

For tr T_4 , in the identity term we have $\omega'(\sqrt{4}) = -1$. So

$$\operatorname{tr}(T_4 | S_5(\sigma_t^{\zeta})) = 8 \left[-\frac{4}{3} + \Phi\left(\begin{pmatrix} -12\\1 & 3 \end{pmatrix} \right) + \Phi\left(\begin{pmatrix} -12\\1 & 6 \end{pmatrix} \right) + \Phi\left(\begin{pmatrix} -4\\1 & 1 \end{pmatrix} \right) + \Phi\left(\begin{pmatrix} -4\\1 & 2 \end{pmatrix} \right) + \Phi\left(\begin{pmatrix} -4\\1 & 3 \end{pmatrix} \right) \right].$$

The last term can be eliminated since it is unramified at p = 3 and it has no characteristic root modulo 3. The remaining orbital integrals are computed locally as follows, and the formula for tr T_4 in Example 7.20 follows upon simplification.

γ	Δ_{γ}	ℓ	m	Φ_{∞}	Φ_3	Φ_ℓ
$\binom{-12}{1}$	$-3 \cdot 13$	13	1	$5\sqrt{3}\cdot 8^{-1}$	$-i\bar{\zeta}\sqrt{3}\cdot\delta_{t=-1}$	2
$\begin{pmatrix} -12 \\ 1 & 6 \end{pmatrix}$	$-2^2 \cdot 3$	2	$\frac{1}{6}$	$-\sqrt{3}$	$i\bar{\zeta}\sqrt{3}\cdot\delta_{t=-1}$	4
$\begin{pmatrix} -4 \\ 1 & 1 \end{pmatrix}$	$-3\cdot5$	5	$\frac{1}{2}$	$7 \cdot 8^{-1}$	$\frac{1}{2}(3t-1)$	2
$\begin{pmatrix} -4 \\ 1 & 2 \end{pmatrix}$	$-2^2 \cdot 3$	2	$\frac{1}{6}$	1	$\frac{1}{2}(3t+1)$	4

7.5.2. Let $N=2^311^2$ and k=6, and let σ^{ζ} be a simple supercuspidal representation of $\operatorname{PGL}_2(\mathbb{Q}_2)$ and σ_{ν} a depth zero supercuspidal representation of $\operatorname{PGL}_2(\mathbb{Q}_{11})$. Here, $\zeta \in \{\pm 1\}$, and ν is one of the five primitive characters of L^* listed in (7-5), where $L=\mathbb{F}_{11^2}$ and we take the generator t of L^* to be a root of the polynomial $X^2+7X+2\in\mathbb{F}_{11}[X]$. Let $\hat{\sigma}$ be the associated tuple. Then by Theorem 7.1,

$$\dim S_6(\hat{\sigma}) = \frac{25}{4} + \frac{1}{2}\Phi\left(\begin{pmatrix} & -2 \\ 1 & \end{pmatrix}, f^1\right) + \frac{1}{2}\Phi\left(\begin{pmatrix} & -1 \\ 1 & \end{pmatrix}, f^1\right) + \Phi\left(\begin{pmatrix} & -2 \\ 1 & 2 \end{pmatrix}, f^1\right).$$

Over \mathbb{F}_{11} , $X^2 + 2 = (x+3)(x-3)$, so $\binom{1}{1}$ is hyperbolic in $G(\mathbb{Q}_{11})$ by Hensel's lemma, and therefore its orbital integral vanishes. Using Example 6.6 and the argument at (7-3),

$$\frac{1}{2}\Phi\left(\begin{pmatrix} & -1 \\ 1 & \end{pmatrix}\right) = \frac{1}{2}m\Phi_{\infty}\Phi_{2}\Phi_{11} = \frac{1}{2}\cdot\frac{1}{4}\cdot(-1)^{6/2}\cdot1\cdot2\epsilon_{11} = -\frac{1}{4}\epsilon_{11}.$$

Taking $\gamma = \binom{-2}{2}$, $P_{\gamma}(X) = X^2 - 2X + 2$ is irreducible over \mathbb{F}_{11} , so by (6-22),

$$\Phi_{11} = -\overline{\nu(\gamma)} - \overline{\nu^{11}(\gamma)}.$$

For $L^* = \langle t \rangle$ as above, we find (using software) that t^{51} has minimum polynomial $P_{\gamma}(X)$. Therefore, if $\nu = \nu_m$ for $m = 10w \in \{10, 20, 30, 40, 50\}$ as in (7-5) where $\mu_m(t) = e\left(\frac{m}{120}\right)$, we have

$$\nu_m(\gamma) = e\left(\frac{51m}{120}\right) = e\left(\frac{17w}{4}\right) = e\left(\frac{w}{4}\right) = i^w.$$

Using this, $\Phi_{11}(\gamma)$ is given by

As in the proof of Proposition 7.12, $m = \frac{1}{4}$, $\Phi_{\infty} = 1$ (since k = 6), and $\Phi_2 = -\zeta$. Hence $\Phi(\gamma) = -\frac{1}{4}\zeta \,\Phi_{11}$ for Φ_{11} as above. Thus

$$\dim S_{6}(\hat{\sigma}) = \frac{25}{4} - \frac{1}{4}\epsilon_{11} - \frac{1}{4}\zeta \Phi_{11} = \begin{cases} 6 & \text{if } \epsilon_{11} = 1, \text{ or } \zeta = 1 \text{ and } \nu = \nu_{20}, \\ & \text{or } \zeta = -1 \text{ and } \nu = \nu_{40}, \\ 7 & \text{if } \zeta = 1 \text{ and } \nu = \nu_{40}, \\ & \text{or } \zeta = -1 \text{ and } \nu = \nu_{20}. \end{cases}$$
(7-23)

We would like to match the above spaces to Galois orbits of twist-minimal newforms in $S_6^{\text{new}}(2^311^2)$. In the table below, the first five columns show LMFDB [34] data, with AL entries corresponding to the Atkin–Lehner signs at p=2, 11. These are equal to ζ and ϵ_{11} respectively. The dim column gives the size of the orbit.

LMFDB label	dim	$\operatorname{tr} T_7$	AL 2	AL 11	(ζ, ν)
968.6.a.f	6	-124	_	_	$(-1, \nu_{40})$
968.6.a.g	6	124	+	_	$(1, \nu_{20})$
968.6.a.h	6	-88	+	+	$(1, \nu_{30})$
968.6.a.i	6	88	_	+	$(-1, \nu_{30})$
968.6.a.j	7	-62	_	_	$(-1, \nu_{20})$
968.6.a.k	7	62	+	_	$(1, \nu_{40})$
968.6.a.1	6	-206	+	+	$(1, \nu_{10}) \oplus (1, \nu_{50})$
968.6.a.m	6	206	_	+	$(-1, \nu_{10}) \oplus (-1, \nu_{50})$

In the final column we have adopted the notation $S_6(\hat{\sigma}) = (\zeta, \nu)$. This column was obtained as follows. Comparing (7-22) and (7-23) with the AL and dim columns, we immediately infer the entries with $\epsilon_{11} = -1$, i.e., with ν_{20} and ν_{40} . We can distinguish the remaining entries by looking at Hecke eigenvalues. For this we apply Theorem 1.1 to compute $\operatorname{tr}(T_7 | S_6(\hat{\sigma}))$. The result is the following.

Example 7.21. Let $N=2^311^2$ and $\hat{\sigma}=(\sigma^{\zeta},\sigma_{\nu})$ be a tuple of supercuspidal representations of conductors 2^3 and 11^2 respectively, as above. Then

$$tr(T_7 | S_6(\hat{\sigma})) = -98\zeta \epsilon_{11} - 5\zeta X_{11} - 31Y_{11},$$

where ϵ_{11} , X_{11} and Y_{11} are given as follows:

ν	ν_{10}	ν_{20}	ν_{30}	ν_{40}	v_{50}
ϵ_{11}	+	_	+	_	+
X_{11}	1	1	-2	1	1
<i>Y</i> ₁₁	$\sqrt{3}$	-1	0	1	$-\sqrt{3}$

For example, in the notation used above,

$$\operatorname{tr}(T_7 | (1, \nu_{10})) = -103 - 31\sqrt{3}, \quad \operatorname{tr}(T_7 | (1, \nu_{50})) = -103 + 31\sqrt{3}.$$

We sketch the proof as follows. By Theorem 1.1,

 $\operatorname{tr}(T_7 | S_6(\hat{\sigma}))$

$$=7^{2} \left[\frac{1}{2} \Phi \left(\begin{pmatrix} & -7 \\ 1 & \end{pmatrix} \right) + \frac{1}{2} \Phi \left(\begin{pmatrix} & -14 \\ 1 & \end{pmatrix} \right) + \sum_{r=1}^{5} \Phi \left(\begin{pmatrix} & -7 \\ 1 & r \end{pmatrix} \right) + \sum_{r=1}^{3} \Phi \left(\begin{pmatrix} & -14 \\ 1 & 2r \end{pmatrix} \right) \right].$$

All but three of the orbital integrals vanish for simple reasons. The matrices $\binom{1}{1}^{-7}$, $\binom{-7}{1}$, $\binom{-7}{1}$, $\binom{-7}{1}$, and $\binom{-14}{1}$ are hyperbolic in $G(\mathbb{Q}_{11})$, since their characteristic polynomials have two distinct roots modulo 11. The matrices $\binom{-7}{1}$, $\binom{-7}{1}$ are unramified at p=2 but do not have characteristic roots modulo 2. So the associated orbital integrals vanish by Proposition 5.6, and

$$\operatorname{tr}(T_7 \mid S_6(\hat{\sigma})) = 7^2 \left[\frac{1}{2} \Phi\left(\begin{pmatrix} & -14 \\ 1 & \end{pmatrix} \right) + \Phi\left(\begin{pmatrix} & -7 \\ 1 & 4 \end{pmatrix} \right) + \Phi\left(\begin{pmatrix} & -14 \\ 1 & 6 \end{pmatrix} \right) \right].$$

The formula in Example 7.21 follows upon computing each of these terms locally. The local results are shown in the following table, with notation as in the previous N=27 example. The global orbital integral for a given row is $\Phi=m\Phi_{\infty}\Phi_{2}\Phi_{11}\Phi_{\ell}$.

$$\begin{array}{|c|c|c|c|c|c|c|c|c|}\hline \gamma & \Delta_{\gamma} & \ell & m & \Phi_{\infty} & \Phi_{2} & \Phi_{11} & \Phi_{\ell} \\\hline \begin{pmatrix} 1^{-14} \end{pmatrix} & -2^{3} \cdot 7 & 7 & 1 & -1 & \zeta & 2\epsilon_{11} & 2 \\ \begin{pmatrix} -14 \\ 1 & 6 \end{pmatrix} & -2^{2} \cdot 5 & 5 & \frac{1}{2} & \frac{5}{7^{2}} & -\zeta & X_{11} & 2 \\ \begin{pmatrix} -7 \\ 1 & 4 \end{pmatrix} & -2^{2} \cdot 3 & 3 & \frac{1}{6} & \frac{31}{7^{2}} & -3 & Y_{11} & 2 \\ \hline \end{array}$$

The Φ_{11} column was determined as follows. As described earlier, $\mathbb{F}^*_{11^2} = \langle t \rangle$ where $t^2 + 7t + 2 = 0$. For each γ as above, there is a power t^j whose minimum polynomial over \mathbb{F}_{11} is $P_{\gamma}(X)$. The power j was found with software, and is given as follows:

In each case, (6-22) implies that

$$\Phi_{11} = -\overline{\nu(\gamma)} - \overline{\nu^{11}(\gamma)} = -\overline{\nu(t^j)} - \overline{\nu(t^{11j})}.$$

By definition, $v_m(t) = e\left(\frac{m}{120}\right)$, so if $v = v_m$ for m = 10w,

$$\Phi_{11}(\gamma) = -e\left(-\frac{jw}{12}\right) - e\left(-\frac{11jw}{12}\right),\,$$

which can be evaluated by hand or using software to obtain the Φ_{11} column in the above table.

Acknowledgements

I am grateful to Charles Li for his very helpful input on the quotient measure that is computed in Section 4.7. I thank the referees for many insightful comments which led to improved exposition, and Andrew Booker, David Bradley, Min Lee, and Kimball Martin for helpful conversations. Partial support for this research was provided by an AMS-Simons Research Enhancement Grant for Primarily Undergraduate Institution Faculty.

References

- [1] M. Adrian and B. Liu, "Some results on simple supercuspidal representations of $GL_n(F)$ ", *J. Number Theory* **160** (2016), 117–147. MR Zbl
- [2] B. C. Berndt, R. J. Evans, and K. S. Williams, Gauss and Jacobi sums, Wiley, New York, 1998.MR Zbl
- [3] V. Blomer, "The relative trace formula in analytic number theory", pp. 51–73 in *Relative trace formulas*, edited by W. Müller et al., Springer, 2021. MR Zbl
- [4] A. R. Booker and A. Strömbergsson, "Numerical computations with the trace formula and the Selberg eigenvalue conjecture", *J. Reine Angew. Math.* **607** (2007), 113–161. MR Zbl
- [5] A. R. Booker, M. Lee, and A. Strömbergsson, "Twist-minimal trace formulas and the Selberg eigenvalue conjecture", *J. Lond. Math. Soc.* (2) **102**:3 (2020), 1067–1134. MR Zbl
- [6] C. Breuil and A. Mézard, "Multiplicités modulaires et représentations de $\operatorname{GL}_2(\mathbf{Z}_p)$ et de $\operatorname{Gal}(\overline{\mathbf{Q}}_p/\mathbf{Q}_p)$ en l=p", Duke Math. J. **115**:2 (2002), 205–310. MR Zbl
- [7] C. J. Bushnell and G. Henniart, *The local Langlands conjecture for* GL(2), Grundl. Math. Wissen. **335**, Springer, 2006. Zbl
- [8] K. Child, "Twist-minimal trace formula for holomorphic cusp forms", *Res. Number Theory* **8**:1 (2022), art. id. 11, 27 pp. MR Zbl
- [9] J. W. Cogdell, "Lectures on *L*-functions, converse theorems, and functoriality for GL_n", pp. 1–96 in *Lectures on automorphic L-functions*, edited by J. W. Cogdell et al., Fields Inst. Monogr. 20, Amer. Math. Soc., Providence, RI, 2004. MR
- [10] L. V. Dieulefait, A. Pacetti, and P. Tsaknias, "On the number of Galois orbits of newforms", J. Eur. Math. Soc. (JEMS) 23:8 (2021), 2833–2860. MR Zbl
- [11] G. P. Dresden, "There are only nine finite groups of fractional linear transformations with integer coefficients", *Math. Mag.* **77**:3 (2004), 211–218. MR Zbl
- [12] S. S. Gelbart, Automorphic forms on adele groups, Annals of Mathematics Studies 83, Princeton Univ. Press, 1975. MR Zbl
- [13] S. Gelbart, "An elementary introduction to the Langlands program", *Bull. Amer. Math. Soc.* (N.S.) 10:2 (1984), 177–219. MR Zbl
- [14] S. Gelbart, Lectures on the Arthur–Selberg trace formula, University Lecture Series 9, Amer. Math. Soc., Providence, RI, 1996. MR Zbl
- [15] S. Gelbart and H. Jacquet, "Forms of GL(2) from the analytic point of view", pp. 213–251 in *Automorphic forms, representations and L-functions, I* (Corvallis, Ore., 1977), edited by A. Borel and W. Casselman, Proc. Sympos. Pure Math. **33**, Amer. Math. Soc., Providence, RI, 1979. MR Zbl

- [16] J. Gordon, "Orbital integrals and normalizations of measures", pp. 247–314 in On the Langlands program: endoscopy and beyond, edited by W. T. Gan et al., Lect. Notes Ser. Inst. Math. Sci. Natl. Univ. Singap. 43, World Sci. Publ., Hackensack, NJ, 2024. MR
- [17] B. H. Gross, "Irreducible cuspidal representations with prescribed local behavior", Amer. J. Math. 133:5 (2011), 1231–1258. MR Zbl
- [18] Y. Hu, "The Petersson/Kuznetsov trace formula with prescribed local ramifications", Amer. J. Math. 146:5 (2024), 1193–1252. MR Zbl
- [19] Y. Hu, I. Petrow, and M. P. Young, "A generalized PGL(2) Petersson/Bruggeman/Kuznetsov formula for analytic applications", preprint, 2024. Zbl arXiv 2411.05672
- [20] H. Iwaniec and E. Kowalski, *Analytic number theory*, American Mathematical Society Colloquium Publications 53, Amer. Math. Soc., Providence, RI, 2004. MR Zbl
- [21] J.-L. Kim, S. W. Shin, and N. Templier, "Asymptotic behavior of supercuspidal representations and Sato-Tate equidistribution for families", Adv. Math. 362 (2020), art. id. 106955, 57 pp. MR Zbl
- [22] A. W. Knapp and J. D. Rogawski, "Applications of the trace formula", pp. 413–431 in Representation theory and automorphic forms (Edinburgh, 1996), edited by T. N. Bailey and A. W. Knapp, Proc. Sympos. Pure Math. 61, Amer. Math. Soc., Providence, RI, 1997. MR Zbl
- [23] A. Knightly and C. Li, *Traces of Hecke operators*, Mathematical Surveys and Monographs 133, Amer. Math. Soc., Providence, RI, 2006. MR Zbl
- [24] A. Knightly and C. Li, "Petersson's trace formula and the Hecke eigenvalues of Hilbert modular forms", pp. 145–187 in *Modular forms on Schiermonnikoog* (Netherlands, 2006), edited by B. Edixhoven et al., Cambridge Univ. Press, Cambridge, 2008. Zbl
- [25] A. Knightly and C. Li, "Modular L-values of cubic level", Pacific J. Math. 260:2 (2012), 527–563. MR Zbl
- [26] A. Knightly and C. Li, Kuznetsov's trace formula and the Hecke eigenvalues of Maass forms, Mem. Amer. Math. Soc. 1055, 2013. MR Zbl
- [27] A. Knightly and C. Li, "Simple supercuspidal representations of GL(n)", Taiwanese J. Math. 19:4 (2015), 995–1029. MR Zbl
- [28] A. Knightly and C. Li, "Traces of Hilbert modular Hecke operators", in preparation.
- [29] A. Knightly and C. Ragsdale, "Matrix coefficients of depth-zero supercuspidal representations of GL(2)", *Involve* 7:5 (2014), 669–690. MR Zbl
- [30] R. E. Kottwitz, "Tamagawa numbers", Ann. of Math. (2) 127:3 (1988), 629-646. MR Zbl
- [31] P. C. Kutzko, "On the supercuspidal representations of Gl₂, II", Amer. J. Math. 100:4 (1978), 705–716. MR
- [32] E. Landau, Elementary number theory, Chelsea Publishing, New York, 1958. MR Zbl
- [33] S. Lang, *Elliptic functions*, 2nd ed., Graduate Texts in Mathematics 112, Springer, 1987. MR Zbl
- [34] The LMFDB collaboration, "The L-functions and modular forms database", electronic reference (accessed 15 October 2023), 2023, available at https://www.lmfdb.org.
- [35] D. Loeffler and J. Weinstein, "On the computation of local components of a newform", *Math. Comp.* **81**:278 (2012), 1179–1200. MR Zbl
- [36] K. Mahler, p-adic numbers and their functions, 2nd ed., Cambridge Tracts in Mathematics 76, Cambridge Univ. Press, 1981. MR Zbl
- [37] G. Martin, "Dimensions of the spaces of cusp forms and newforms on $\Gamma_0(N)$ and $\Gamma_1(N)$ ", *J. Number Theory* **112**:2 (2005), 298–331. MR Zbl

- [38] K. Martin, "Refined dimensions of cusp forms, and equidistribution and bias of signs", J. Number Theory 188 (2018), 1–17. MR Zbl
- [39] K. Martin, "Root number bias for newforms", Proc. Amer. Math. Soc. 151:9 (2023), 3721–3736.
 MR Zbl
- [40] T. Miyake, Modular forms, Springer, 1989. MR Zbl
- [41] J. Neukirch, Algebraic number theory, Grundl. Math. Wissen. 322, Springer, 1999. MR Zbl
- [42] K. Okada, "Hecke eigenvalues for real quadratic fields", Experiment. Math. 11:3 (2002), 407–426. MR Zbl
- [43] M. R. Palm, Explicit GL(2) trace formulas and uniform, mixed Weyl laws, Ph.D. thesis, Universität Göttingen, 2012, available at http://dx.doi.org/10.53846/goediss-2555.
- [44] I. Petrow and M. P. Young, "The Weyl bound for Dirichlet *L*-functions of cube-free conductor", *Ann. of Math.* (2) **192**:2 (2020), 437–486. MR Zbl
- [45] I. Petrow and M. P. Young, "The fourth moment of Dirichlet *L*-functions along a coset and the Weyl bound", *Duke Math. J.* **172**:10 (2023), 1879–1960. MR Zbl
- [46] Q. Pi and Z. Qi, "Bias of root numbers for modular newforms of cubic level", *Proc. Amer. Math. Soc.* **149**:12 (2021), 5035–5047. MR Zbl
- [47] I. Piatetski-Shapiro, Complex representations of GL(2, K) for finite fields K, Contemporary Mathematics 16, Amer. Math. Soc., Providence, RI, 1983. MR Zbl
- [48] R. Schmidt, "Some remarks on local newforms for GL(2)", *J. Ramanujan Math. Soc.* 17:2 (2002), 115–147. MR Zbl
- [49] J.-P. Serre, A course in arithmetic, Graduate Texts in Mathematics 7, Springer, 1973. MR Zbl
- [50] J.-P. Serre, Local fields, Graduate Texts in Mathematics 67, Springer, 1979. MR Zbl
- [51] J.-P. Serre, "Répartition asymptotique des valeurs propres de l'opérateur de Hecke T_p ", J. Amer. Math. Soc. 10:1 (1997), 75–102. MR
- [52] A. J. Silberger, *Introduction to harmonic analysis on reductive p-adic groups*, Mathematical Notes **23**, Princeton Univ. Press, 1979. MR Zbl
- [53] A. J. Silberger, "Special representations of reductive p-adic groups are not integrable", Ann. of Math. (2) 111:3 (1980), 571–587. MR Zbl
- [54] J. T. Tate, "Fourier analysis in number fields, and Hecke's zeta-functions", pp. 305–347 in Algebraic number theory (Brighton, 1965), edited by J. W. S. Cassels and A. Fröhlich, Academic, London, 1967. MR Zbl
- [55] J. B. Tunnell, "On the local Langlands conjecture for GL(2)", Invent. Math. 46:2 (1978), 179–200. MR Zbl
- [56] J. Weinstein, "Hilbert modular forms with prescribed ramification", Int. Math. Res. Not. 2009:8 (2009), 1388–1420. MR Zbl

Received 10 Apr 2024. Revised 16 Mar 2025.

ANDREW KNIGHTLY:

andrew.knightly@maine.edu

Department of Mathematics and Statistics, University of Maine, Orono, ME, United States



ESSENTIAL NUMBER THEORY

msp.org/ent

EDITOR-IN-CHIEF

Lillian B. Pierce Duke University pierce@math.duke.edu

EDITORIAL BOARD

Adebisi Agboola UC Santa Barbara

agboola@math.ucsb.edu

Valentin Blomer Universität Bonn

ailto:blomer@math.uni-bonn.de

Frank Calegari University of Chicago fcale@math.uchicago.edu

Laura DeMarco Harvard University

demarco@math.harvard.edu

Ellen Eischen University of Oregon

eeischen@uoregon.edu

Kirsten Eisenträger Penn State University

kxe8@psu.edu

Amanda Folsom Amherst College

afolsom@amherst.edu

Edray Goins Pomona College edray.goins@pomona.edu

University of Turku

ksmato@utu.fi

Sophie Morel ENS de Lyon

Kaisa Matomäki

James Newton

Raman Parimala

sophie.morel@ens-lyon.fr

Oxford University

newton@maths.ox.ac.uk

Parimala Emory University parimala.raman@emory.edu

Jonathan Pila University of Oxford jonathan.pila@maths.ox.ac.uk

Johannan.pha@mans.ox.ac.uk

Peter Sarnak Princeton University/Institute for Advanced Study sarnak@math.princeton.edu

Richard Taylor Stanford University

rltaylor@stanford.edu

Anthony Várilly-Alvarado Rice University

av15@rice.edu

John Voight Dartmouth College

john.voight@dartmouth.edu

Melanie Matchett Wood Harvard University

mmwood@math.harvard.edu

Zhiwei Yun MIT

zyun@mit.edu

Tamar Ziegler Hebrew University

tamar.ziegler@mail.huji.ac.il

PRODUCTION

Silvio Levy (Scientific Editor)

production@msp.org

See inside back cover or msp.org/ent for submission instructions.

Essential Number Theory (ISSN 2834-4634 electronic, 2834-4626 printed) at Mathematical Sciences Publishers, 798 Evans Hall #3840, c/o University of California, Berkeley, CA 94720-3840 is published continuously online.

ENT peer review and production are managed by EditFlow® from MSP.



https://msp.org/

© 2025 Mathematical Sciences Publishers

ESSENTIAL NUMBER THEORY 2025 vol. 4 no. 2 Cohomology of Fuchsian groups and Fourier interpolation 217 MATHILDE GERBELLI-GAUTHIER and AKSHAY VENKATESH Harder's denominator problem for $SL_2(\mathbb{Z})$ and its applications 251 HOHTO BEKKI and RYOTARO SAKAMOTO The Bombieri–Pila determinant method 327 THOMAS F. BLOOM and JARED DUKER LICHTMAN Counting locally supercuspidal newforms 349 ANDREW KNIGHTLY